Chapter 3

The future of work in Europe: job quality and work organisation for a smart, sustainable and inclusive growth⁽¹⁾

1. BETTER JOBS AND WORK ORGANISATION YIELD A MORE PRODUCTIVE WORKFORCE

This chapter assesses the future EU labour market challenges and opportunities in terms of job quality and work organisation and their likely impact on labour market developments over the next 10 years. It presents recent developments in job quality and work organisation (and their interactions) and highlights their impact on productivity, labour market participation and social cohesion as indicated by recent research. It then explores how technological progress and innovation, globalisation, demographic change and the greening of the economy may affect the workforce's potential via their impact on job quality and work organisation. It ends by discussing how labour market policies can help prevent, cushion or correct adverse developments in job quality and work organisation associated with those structural changes, including issues such as polarisation and inequality, while reinforcing positive developments. The chapter builds on the analysis presented in the 2014 ESDE review(²).

Since the onset of the crisis, job creation has been high on the agenda of policy makers across the EU. As the signs of an economic recovery (albeit weak and unevenly spread across Member States)

are growing, attention is turning to other emerging challenges such as those associated to globalisation or technological progress. These may exacerbate some of the negative developments ensuing from the economic crisis. Such forces may render some jobs obsolete, increase the health and accident risks associated with certain types of jobs or increase the pressure to ensure employees' availability around the clock. They may also bring new opportunities. In this context, forward-looking policies need to address the impact of such forces on jobs, job quality, work organisation and human capital formation. Policy makers will need to monitor, prevent and correct adverse developments, while strengthening positive ones.

The chapter is structured as follows. Section 2 introduces the general and EU concepts of job quality. It also identifies different forms of work organisation across the EU(³). Section 3 presents patterns and trends in job quality across EU Member States and highlights the link between some dimensions of job quality and labour productivity and labour market participation. Section 4 identifies future challenges to job quality associated with globalisation, technological progress and innovation, demographic change and the greening of the economy. Challenges include rising job insecurity, increased polarisation, accelerating skill erosion, gender inequality and a stronger emphasis on knowledge and

creativity. Section 5 describes different types of work organisation, distinguishing those that offer greater autonomy to employees. It explores how work organisation can foster productivity and longer working lives and reduce both absences and health-related costs. It discusses how workplaces can stimulate creativity and foster exchanges between workers, prevent stress, help maintain good physical and mental health and accommodate older workers or those with disabilities or certain diseases. It identifies modern management strategies that can facilitate employees' empowerment and are key to facing future challenges. Section 6 concludes on how to strengthen productivity growth and labour market resilience via improved job quality and increased work organisation innovation, while ensuring that costs and benefits are distributed equitably.

2. JOB QUALITY AND WORK ORGANISATION: MULTI-DIMENSIONAL CONCEPTS

This section reviews the concepts of job quality and work organisation. It describes the EU concept of job quality, based on the EU Quality of Work system of indicators, as agreed within the Employment Committee (EMCO). In this system, indicators are grouped in four main dimensions: socioeconomic security; education and training; working conditions; and work-life and gender balance (Annex 1, Table A1.1). The section then focuses on the EU's four main different forms of work organisation that relate to employees' performance and labour market participation. These are:

^{(&}lt;sup>1</sup>) By Eric Meyermans, Teodora Tchipeva and Bartek Lessaer, with a contribution on work organisation by Agnès Parent-Thiron (Eurofound) and Milos Kankaras (Eurofound).

⁽²⁾ Employment and Social Developments in Europe 2013 Review. Chapter 1, European Commission (2014f).

^{(&}lt;sup>3</sup>) Note that while the analytical framework of this chapter makes a distinction between job quality and working conditions, due regard is given to the possible reinforcing interactions between the two.

Discretionary Learning forms; Lean Production forms; Tayloristic forms; and Traditional Simple forms. See Annex 2 for a brief discussion of the methodology used to classify different types of work organisation.

2.1. Job quality dimensions

Job quality is a complex and multidimensional concept that has been extensively analysed and debated by economists, sociologists, and psychologists. Several factors make its definition and measurement a challenge.

There are a variety of perspectives on work and jobs depending on each individual's work role, and the perspectives of workers and their employers may not necessarily always coincide. Nevertheless, from an employer's perspective there are several factors that should encourage employers to increase the quality of jobs. For example, there is a direct link between a higher level of skills and a firm's productivity, which may encourage employers to provide continuous training. Furthermore, a physically safe and healthy working environment reduces accidents and absences from work and improves productivity and output. Hence, increased job quality can result in better quality goods and services together with a positive impact on companies' income and welfare as a whole.

2.1.1. Job quality: 'subjective' and 'objective' concepts

It is commonplace in analytical research to distinguish between the subjective and objective concepts of job quality. The subjective approach assumes that job quality is the 'utility' a worker derives from the job. That utility depends on job features over which each worker has personal preferences. Each worker values one feature against another in a different way. Some academics argue that measures of wellbeing or job satisfaction can be used as subjective indicators of job quality (4). Such measures take the individual differences into account as it is workers who evaluate the positive and negative aspects of a job and rank them (5).

However, the use of job satisfaction as a one-dimensional measure of job quality

has limitations. For instance, it may be sensitive to each individual's aspirations and expectations. Indeed, workers with low aspirations or expectations often express high job satisfaction, even when— on the basis of measurable variables such as earnings — they are in low-quality jobs. Moreover, factors like one's cultural environment and traditions or personality (e.g. disposition to pessimism/optimism) can affect subjective job satisfaction. Therefore, subjective job satisfaction is prone to bias and can be misleading in measuring and monitoring job quality.

Objective approaches assume that job quality encompasses job features that meet workers' needs. Objective measures of job quality are derived from a given theory of human needs and measure how far jobs meet those needs(⁶). Thus, the objective concept of job quality is not assessed by a one-dimensional measure (e.g. job satisfaction) but by a set of indicators measuring various dimensions associated with the job(⁷).

Different disciplines tend to focus on different dimensions. Economists tend to focus on monetary aspects such as wage levels or working hours(⁸). Sociologists tend to focus more on such factors as occupational

- (?) Some confusion may arise regarding selfreported variables in surveys (e.g. in the EWCS), which sometimes are referred to as 'subjective'. It should be stressed that the variables included in the EWCS refer to 'objective' job features; the term 'subjective' is reserved for reports of feelings, perceptions, attitudes or values. See Eurofound (2012b).
- In the standard neo-classical model, for example, (8) work is disutility and wages are the sole motivation of workers. At market equilibrium the wage level fully reflects the job quality, and it equals the level of productivity and compensates for the disutility of work. In the framework of compensating wage differentials some displeasures that arise from work are explicitly taken into account in the utility function (e.g. injury and occupational disease commuting costs, working hours); they are fully compensated by a wage premium because (by assumption) workers trade off working conditions and benefits for pay (see e.g. Rosen, 1986). In other words, ceteris paribus, workers with similar qualifications who work under bad working conditions are paid more by employers to compensate for the unpleasantness of the job. In a perfectly competitive labour market with perfect information, as assumed in the framework, the wage level reflects job quality. Bustillo et al. (2012), part 5, provide an overview of the empirical literature testing the link between working conditions and differences in pay. By contrast, dual labour market theorists (e.g. Piore, 1971; Edwards, 1979) have contended that bad iob characteristics tend to cluster so that a job that is bad in one dimension tends to be bad in others.

status and the extent to which workers have autonomy and control over their jobs (e.g. Jencks et al., 1988; Goldthorpe and Hope, 1974; Prandy, 1990; Stewart et al., 1980). Psychologists often emphasise how intrinsically meaningful and challenging work is, and thus analyse a variety of psychological measures of job satisfaction such as workers' discretion and trust in their jobs (Guillen and Dahl, 2009; Kalleberg and Vaisey, 2005).

Even though different academic fields conceptualise and measure job quality in different ways, there is some convergence in terms of the work features that are seen to be crucial. Integrated insights from psychology, sociology, applied economics and other fields are enriched by considering the workers' point of view, notably through the development of surveys on job satisfaction and workers' well-being (e.g. Layard, 2005).

Therefore, objective approaches to job quality are based on a *selected* set of indicators depending on the researcher's objectives (see Annex 1 for examples of objective definitions of job quality). Some researchers tend to focus on the characteristics of the job (e.g. Eurofound, European Parliament); others include broader indicators of the economic and labour market environment as well as indicators relating to the personal characteristics of the worker (e.g. ILO 'decent work concept', with indicators on child labour, social protection; UNECE concept).

Most approaches either group the multitude of individual indicators into a system of indicators, or aggregate those indicators into a composite index. Both approaches have advantages and disadvantages. An aggregate index typically trades off the ease of presentation for strong assumptions on the weighting attributed to each indicator, i.e. assumptions about people's preferences for one job feature over another(⁹). Several examples of such aggregation and the use of composite indices are available (Annex 1).

2.1.2. A set of job quality indicators for policy-making at EU level

Job quality issues were first explicitly introduced into the European policy agenda at the Lisbon Council in March 2000, which

^{(&}lt;sup>4</sup>) For a discussion, see Eurofound (2012b).

⁽⁵⁾ Some questions in the semi-structured interviews of the NEUJOBS project reflect this focus on preferences by asking 'Which of the following features (attributes) of your job are more/less important to you?'.

⁽⁶⁾ E.g. Maslow's hierarchy of needs applied to the world of work leads to a number of key job characteristics. Similarly, Green (2006) adapts Sen's capability approach and develops the idea that a 'good job' is one that offers workers a high capability to do and be things that they value.

⁽⁹⁾ The pros and cons of composite indices against a system of indicators are discussed in more detail in Annex 1.

established the objective of 'more and better jobs for all'. In 2001, the Laeken European Council agreed to a comprehensive framework on job quality. The resulting concept of job quality included 10 dimensions, categorised into two themes: 1) characteristics of the job/worker and 2) the wider socioeconomic and labour market context (Annex 1). In 2013, the EU's Employment Committee (EMCO) Indicators Group agreed upon a four-dimensional concept of job quality, subdivided into 10 further sub-dimensions, each with several indicators (Annex 1, Table A1.1). The indicators are drawn predominantly from the EU Labour Force Survey (EU-LFS), the Statistics on Income and Living Conditions (EU-SILC) and Eurofound's latest European Working Conditions Survey (EWCS). The four dimensions are:

- 1. Socioeconomic security, including adequate earnings and job and career security;
- Education and training, including skills development through life-long learning and employability;
- Working conditions, including health and safety at work, work intensity, autonomy and working practices, as well as collective interest representation;
- 4. Work-life and gender balance.

Operationalising the multitude of indicators to facilitate monitoring, assessment and policy-making remains a challenging work in progress. Through factor analysis, their number has recently been compressed but the list still remains long⁽¹⁰⁾.

2.2. Work organisation can take different forms

In the ever-changing world of work, employees' well-being, performance and labour market participation depend on the organisation of work by firms. Based on the findings of the three most recent EWCS waves (2000, 2005 and 2010), four broad forms of work organisation can be identified. Table 1 describes the main characteristics of these forms of work organisation among private nonagricultural establishments employing 10 or more workers (see Annex 2 for the methodology used to underpin the classification). The 'Discretionary Learning form' (hereafter Learning) is the first and most prevalent type of organisation, covering some 36% of employees in private companies with 10 or more employees. They are characterised by the highest level of task autonomy (methods of work, speed of work), a very high cognitive dimension (problem solving, learning new things, complexity of tasks), a high level of selfassessment of guality and some autonomous teamwork. There is a low level of monotonous and repetitive tasks and a low incidence of various forms of work pace constraints. These characteristics are similar to models of learning organisations or the 'adhocracy' model of Mintzberg (1979), and coincide with many features of the Scandinavian sociotechnical model, notably its emphasis on self-autonomous teams.

The 'Lean Production form' (hereafter Lean) covers nearly 29% of employees. It is characterised by a strong presence of team work, including self-managed teams, the highest reported use of quality norms and self-assessment of quality, the highest level of task rotation and horizontal and norm-based constraints, a very high level of cognitive demands and higher levels of task autonomy. This type of organisation displays strong learning dynamics and relies on employees' abilities to solve problems themselves. Work is embedded in numerous quantitative and organisational pace constraints and requires the respect of strict quality standards, granting employees a rather 'controlled' autonomy in their work.

The 'Tayloristic form' covers about 20% of employees. This type of work organisation displays a high level of non-autonomous team work, the lowest level of task autonomy, limited cognitive demands at work, very high levels of use of pre-defined quality standards (and lower levels of selfassessed quality standards) and a very high level of pace constraints, especially those created by limitations in the speed of machines or production flow.

The 'Traditional or Simple form' of organisation covers nearly 16% of employees. It is characterised by the lowest incidence of work pace constraints and the use of pre-defined or self-assessed quality standards. Workers belonging to this organisational form have less work pace autonomy and generally face the least cognitively demanding tasks, with only a few instances of teamwork and work rotation. In such establishments, work organisation methods are not (strictly) codified and are largely informal, probably as a consequence of the lower complexity of the work tasks involved.

2.3. Work organisation impacts on job quality and performance

As can be seen in Table 1, nearly two thirds of employees in private establishments with 10 or more employees (excluding agriculture) work in forms of work organisation characterised by strong learning dynamics and high problem-solving activity: the Learning and the Lean production forms. These are often labelled together under the heading of High Performance Work Places — HPWS (Appelbaum and Batt, 1993).

Though similar, Learning and Lean organisations differ in a number of dimensions. Learning organisations place additional importance on the wholeness of tasks, a higher level of personal autonomy and initiative, less emphasis on strict adherence to standards and more open access to decision-making process. In contrast, Lean organisations are more hierarchical, and task autonomy and pace of work are more limited and controlled. Also, Learn organisations do not appear to compensate workers fully for their increased level of responsibility and the need to address ongoing problem-solving activities in an increasingly complex environment. This may result in problems relating to personal well-being, health or work-life balance similar to those experienced in Tayloristic organisations.

Employees working in Tayloristic and more Traditional or Simple forms of work organisation, which account for around a third of all employees, have much less task autonomy, rarely deal with cognitively demanding tasks and have fewer opportunities to learn new things. Furthermore, while workers in more Traditional and Simple forms of work organisation face fewer quality norms or work pace constraints, Tayloristic forms of organisation are marked by much stricter controls in both respects.

Finally, a meta-analysis of 92 studies (Combs et al., 2006) found evidence that HPWS enhance organisational performance. These organisations are better suited for more volatile and complex environments, including more competitive and globalised markets.

^{(&}lt;sup>10</sup>) In the table in Annex 1 these indicators are marked 'FACTOR indicating...(the particular aspect of work)'.

Tabl	e I: work organisation variab	les across the c	lasses (%) of em	Jioyees) — 201	0	
		Work organisation classes				
		Discretionary learning	Lean production	Tayloristic	Traditional or simple	
	Methods of work*	85.90%	64.20%	7.70%	33.70%	
Autonomy in work	Speed or rate of work*	88.80%	66.20%	13.80%	46.20%	
	Order of tasks	80.80%	62.20%	14.60%	35.70%	
	Learning new things*	83.40%	90.80%	37.60%	22.50%	
Cognitive dimen-	Problem solving activities*	98.00%	91.50%	58.10%	46.00%	
Sions of work	Complexity of tasks*	74.60%	86.00%	32.20%	12.70%	
Quality	Self-assessment*	83.20%	91.30%	63.40%	23.00%	
Quality	Quality norms*	77.80%	97.70%	94.50%	35.40%	
Monotony of tasks*		29.60%	60.60%	75.90%	52.40%	
Repetitiveness of tasks*		16.50%	38.20%	51.60%	24.00%	
Task rotation*		40.20%	76.30%	46.30%	31.20%	
	Automatic*	8.00%	43.20%	64.00%	13.40%	
	Norm-based*	41.80%	77.20%	73.00%	17.70%	
Work pace	Hierarchical*	28.50%	68.40%	65.90%	27.20%	
	Horizontal*	29.50%	86.40%	66.60%	27.00%	
	Direct demands from other people	62.80%	65.00%	53.10%	55.10%	
Toomwork*	With autonomy	32.46%	46.28%	16.78%	16.18%	
Teamwork	Without autonomy	24.94%	46.86%	43.72%	25.99%	
Accistance	From colleagues	70.49%	82.61%	65.54%	62.70%	
ASSISTANCE	From hierarchy	61.06%	62.29%	47.66%	46.35%	
Overall proportion of w	n of workers in the four forms ork organisation	36.00%	28.70%	19.50%	15.80%	

- - - -.

ij

Source: Eurofound based on EWCS (2010).

Note: Variables with an asterisk (*) have been used to identify the four main different organisation forms. Further variables are used to provide additional information.

THE EFFECTS OF JOB QUALITY ON PRODUCTIVITY, LABOUR MARKET PARTICIPATION AND SOCIAL COHESION

This section presents patterns and trends in job quality based on the EU job quality concept and selected(11) EU Quality of Work Indicators agreed by the EMCO Indicators group. The structure of this section follows the breakdown of the EMCO job quality indicators into its four dimensions (here subsections): 1) socioeconomic security, 2) education and training, 3) working conditions and 4) work-life and gender balance (EMCO Indicators table in Annex 1). For each subsection, the transmission mechanism between job quality and productivity, labour market participation or inequality is presented.

3.1. Socioeconomic security: synergy of interests

3.1.1. Earnings affect workers' motivation and effort

Earnings from work are an important dimension of job quality: they are the main source of income for workers, and affect many dimensions of workers' well-being, including better access to goods and services or better health.

An adequate level of pay helps avoid in-work poverty and social exclusion (12).

The literature suggests that the level and distribution of earnings can have a direct impact on productivity and output. A higher wage (above the free market level) increases the cost of job loss for workers and creates incentives to be productive and not to shirk (e.g. Akerlof and Yellen, 1986). Alternatively, the amount above the market level rate may be seen by the worker as a 'gift', inducing higher motivation, commitment and effort. For employers, a wage above the market level can reduce labour turnover and thus reduce the cost of recruitment and initial training, especially of highly qualified workers.

^{(&}lt;sup>11</sup>) The aim of the chapter is not to review all indicators in the EMCO list. Rather, it reviews a selected number to illustrate main trends and the links between job quality and outcomes such as productivity, labour market participation and existing inequalities among groups. Furthermore, the high levels of correlation between indicators within each sub-dimension make it unnecessary to provide a detailed analysis of all the indicators on the EMCO list. Additional information is presented in footnotes or in Annex 3.

⁽¹²⁾ More details on brochure on in-work poverty available at http://epp.eurostat.ec.europa.eu/ cache/ITY_OFFPUB/KS-RA-10-015/EN/KS-RA-10-015-EN.PDF

Workers derive job satisfaction not only from the level of their earnings but also from their earnings relative to those of other workers, i.e. the distribution of earnings (e.g. OECD 2014), though the effect may be ambiguous(13). On one hand, a wider wage dispersion may induce workers to make a stronger effort to get into the upper wage scale, increasing individual and overall productivity (e.g. Lazear and Rosen, 1981). However, it may undermine cooperation among workers, decreasing the overall productivity level (e.g. Akerlof and Yellen, 1990). Moreover, it may limit the ability to pay for education and training of those in the lower brackets and result in an under-investment in human capital with a negative impact on the individual's own productivity (e.g. Galor and Zeira, 1993) and potentially that of their co-workers (e.g. Lucas, 1988; Lloyd-Ellis, 2003). Finally, to the extent that workers perceive they are not receiving their fair share of the wealth they create, the call for redistribution via taxes may increase, with an effect on innovation and productivity growth (e.g. Alesina and Rodrik, 1994; Alesina and Perotti, 1994; Ostry et al., 2014; Piketty 2014).

Based on the EWCS 2010, Chart 1 shows that satisfaction with pay in 2010 was lowest in Hungary, Lithuania, Portugal and Latvia, and highest in Denmark, Luxembourg and the Netherlands(14).

Chart 2 shows that in-work poverty was highest in Poland, Luxembourg, Greece and Romania in 2013, while it was among the lowest in Finland, the Czech Republic, the Netherlands and Denmark. The in-work atrisk-of-poverty rate measures the share of persons who are at work and have an

- (13) The literature on the determinants of subjective well-being has focused on the relative importance of absolute and relative earnings, without however providing for a conclusive answer so far. Easterlin (1974), who sparked the debate, aroued that once basic needs have been met it is only the relative income that matters for increasing one's well-being. Recent studies challenged this proposition by arguing that the relationship between income and life satisfaction is log-linear (Deaton and Kahneman, 2010; Sacks et al., 2012: Stevenson and Wolfers, 2008 and 2013), or that there are declining marginal returns to income in terms of subjective well-being, from which follows that overall welfare is a function of both absolute income and its distribution. Most studies that have analysed the role of relative wage comparisons for well-being found negative effects (Clark and Oswald, 1996; Luttmer, 2005; Card et al., 2012) that have been typically interpreted as status effects: the higher the earnings of the reference group relative to one's personal earnings, the lower one's social status and well-being.
- (14) See also Annex 3. Charts A3.1 and A3.2 for real wages adjusted for productivity and mean monthly earnings.

equivalised disposable income below the risk-of-poverty threshold, which is set at 60% of the national median equivalised disposable income (after social transfers). Chart 3 shows the distribution of earnings, as measured by the Gini earnings index, in 2000 and 2010 for the Member States for which the data are



Chart 2: In-work poverty, 2007 and 2013 20 18 ♦ 2007 ■ 2013 16 14 12 10 % 8 6 4 2 0 CZ

Source: Eurostat SILC, 2012, table: ilc iw01.

Notes: The in-work at-risk-of-poverty rate measures the share of persons who are at work and have an equivalised disposable income below the risk-of-poverty threshold, which is set at 60% of the national median equivalised disposable income (after social transfers). For more details see http://epp.eurostat. ec.europa.eu/cache/ITY_OFFPUB/KS-RA-10-015/EN/KS-RA-10-015-EN.PDF; 2012 observation for IE.





Notes: Gini coefficient calculated on basis of earnings of employed persons, not income. The Gini coefficient is an indicator with a value between 0 and 1. Lower values indicate higher equality No observation available for HR and DE for 2000.

ij

available (¹⁵). On average, this indicator remained fairly constant over the period at around 0.3, but ranging from about 0.2 (e.g. Sweden, Finland) to more than 0.4 (e.g. Romania, Portugal) — with a higher value indicating higher inequality. Since the beginning of the decade, inequality has decreased substantially in the Baltic countries and France, while it has increased in Cyprus and Italy.

3.1.2. Job and career security effects on commitment, enhanced firm-specific skills and productivity

Job security strengthens workers' commitment and the opportunities to acquire firm-specific skills, which in turn may enhance individual and team performance, with a positive impact on productivity (e.g. Auer et al., 2005; Brown et al., 2011)(¹⁶). In contrast, involuntary part-time work or long spells of inactivity/ unemployment between temporary jobs) may erode human capital and lead to poor mental health and low life satisfaction (e.g. Green, 2011; Sverke et. al, 2006), negatively affecting personal performance and overall productivity. Moreover, involuntary part-time work or long spells of inactivity/unemployment between temporary jobs decrease the household work intensity and increase the risk of in-work poverty and social exclusion. Job security may, nevertheless, induce shirking in some circumstances if not counteracted by specific measures (e.g. Yellen, 1984; Shapiro and Stiglitz, 1984; IchoNo and Riphahn, 2005)(17).

A high proportion of workers in Spain, Greece, Portugal, Cyprus, Romania and Slovakia are on involuntary temporary contracts (Chart 4) (¹⁸). Moreover, the

(¹⁶) Using macro-data covering 13 European countries between 1992 and 2002, Auer et al. (2005) report a positive (though eventually decreasing) relationship between job tenure and productivity. Using micro-data from 2004, Brown et al. (2011) show strong employee commitment decreases the probability that labour productivity is below the sample mean by about 10 pps.

(17) Note that job security need not exclude internal job flexibility. For example, it is possible that short-time working arrangements adopted during an economic downtum can have a positive impact on long-run labour productivity to the extent that the free time is used for skill formation

(¹⁸) Note that this chart draws from different surveys covering data for 2013 and 2011.



Sources: Eurostat LFS, table Ifsa_etgar and ilc_lvhl32. Data for involuntary temporary employment is for 2013 (¹); data for transitions if for 2011. Transitions data for Denmark and Ireland is missing.

Note: 15 to 64 years age group; % of total temporary workers.

(1) Involuntary temporary work used in this subsection is based on the Eurostat concept. In particular, employees with temporary contracts are those who declare themselves as having a fixed-term employment contract (see below) or a job which will terminate if certain objective criteria are met, such as completion of an assignment or return of the employee who was temporarily replaced (for more details check the Eurostat metadata available at http://epp.eurostat.ec.europa.eu/cache/ITY_SDDS/EN/lfsa_esms.htm. Employees with fixed-term contracts: Following Eurostat, the concept of fixed-term contract is only applicable to employees, not to the self-employed. In some countries, contracts of this type are settled only in specific cases, e.g. for public-sector jobs, apprentices or other trainees within an enterprise. Given wide institutional discrepancies, the concepts of 'temporary employment' and 'work contract of limited duration' (or 'permanent employment' and 'work contract of unlimited duration) describe situations which, in different institutional contexts, may be considered similar. For the reference definitions, please consult the EU-LFS explanatory notes at http://epp.eurostat.ec.europa.eu/statistics_explained/index.php/EU_labour_force_survey_-_methodology#LFS_explanatory_notes

transition from temporary to permanent contracts is particularly difficult in Spain, Greece, Cyprus and Portugal. In contrast, Austria, Germany, the Netherlands and Estonia have low rates of involuntary temporary employment and high transition rates to permanent employment (¹⁹).

Temporary work needs not necessarily be a negative job feature. If, for example, the reason for temporary employment of young people is that they are in education or training (as in Germany, Austria and Denmark)⁽²⁰⁾ or on a probation period, then a temporary job can be seen as a stepping stone to more stable forms of employment. However, if upward transitions in pay level and/or contract type are impeded and the labour market is highly polarised, the prospects for career advancement and perceptions about the quality of their jobs will be poorer. This may reduce motivation, and thus

(20) In 2013, the share of employees aged 15–24 in temporary contracts due to education or training in all temporary employees aged 15–24 was 85% (though this figure is flagged as unreliable by Eurostat), 80% and 54% in Germany, Austria and Denmark. This percentage remained stable between 2007 and 2013. productivity and growth (see for instance OECD, 2014).

excel file

gif

Chart 5 shows the unfavourable changes observed in the majority of the Member States during the recent crisis (²¹). Involuntary temporary work increased while the transition to more stable employment contracts fell. The situation appears to have deteriorated further in the Southern countries (Greece, Spain, Cyprus) and Slovakia, followed by Bulgaria, the Czech Republic, Hungary and Latvia. Noticeable changes are also seen in Luxembourg and Italy(²²)(²³).

Chart 6 shows a positive change in at least one of the indicators for a limited number of Member States (²⁴). In Austria,

- (²³) These trends may reflect an increased tendency of firms to use temporary contracts to absorb more easily shocks in product (and hence also in labour) demand during the crisis, especially in countries where employment protection legislation is much stricter for permanent than for temporary contracts.
- (²⁴) Note that this chart draws from different surveys covering data for 2013 and 2011.

142

^{(&}lt;sup>15</sup>) Earnings distribution is not to be confused with distribution of income, wealth or opportunities. For a comprehensive study on the latter, see, for example, the GINI project available at http://www.gini-research.org/ articles/home

⁽¹⁹⁾ However, there is a high gender imbalance in transition rates to permanent contract in Estonia (see Annex 3, Chart A3.5).

^{(&}lt;sup>21</sup>) Note that this chart draws from different surveys covering data for 2013 and 2011.

⁽²²⁾ The share of employees aged 15–24 in temporary contracts due to education or training in all temporary employees aged 15–24 decreased substantially in Italy (from 54% to 40%) and Luxembourg (from 52% to 44%) between 2007 and 2013.

involuntary temporary work declined. In Finland and Portugal it increased slightly, but transitions improved. In Germany and Lithuania, involuntary temporary work declined and transitions to more stable employment contracts became easier. Annex 3 gives more detail about the evolution of involuntary temporary work and transitions by country (Annex 3, Charts A3.3–A3.8).

There are significant gender inequalities in the transition from temporary to permanent contracts. In many Member States,







men have higher transition rates to permanent contracts than women. Gender differences in transition rates stand out in Lithuania (30 pps), Estonia (15 pps) and Cyprus (18 pps) (Annex 3, Charts A3.5). Women show better transition rates than men in Romania and Latvia. Annex 3 also shows how transition rates evolved during the recent crisis by gender (Annex 3, Charts A3.5–A3.7). Involuntary temporary work is also more widespread among workers on temporary contracts who are aged 55-64 than among younger workers (aged 15-24), especially in Germany (where the gap is the highest at 60 pps), Luxembourg, Denmark and Ireland(25) (Annex 3, Chart A3.8).

Box 1: How much does job security (duration) contribute to earnings dispersion?

The extent to which individual (i.e. gender, age, educational level), job (i.e. occupation, job duration, employment contract type) and firm (i.e. economic activity, size of the enterprise, existence and type of pay agreement, ownership) characteristics affect the earnings distribution differs within Member States. (See Chart 7 for those for which data are available). On average, occupation is estimated to contribute about 25% to earnings dispersion, followed by education (12%), industry (10%), enterprise size (6%), job duration (6%), age (5%) and gender (3.5%), leaving some 30% of earnings dispersion unexplained by these factors. Job duration appears relatively strong in explaining earnings dispersion in Southern European countries, and also in Germany and Luxembourg. Whether a contract is permanent or of fixed duration contributes strongly in Germany, Poland and the Netherlands, while part-time versus full-time work is estimated to contribute strongly to earnings dispersion in Germany, Latvia, Hungary, the Netherlands, Belgium and Lithuania (WiiW, 2014).

(25) The low share of involuntary temporary young workers in Germany, Austria, Luxembourg and Denmark may be due to the fact that many young people on temporary contracts in these countries are in education or training (see footnotes 20 and 22).





3.2. Education and training may enhance employability and productivity

The literature (e.g. Lucas, 1988; Rebelo, 1991; Dearden et al., 2006; Christen et al., 2008) suggests that human capital formation is directly and positively linked to productivity and labour market participation. Investment in education and training leads to individual increasing returns and generates positive spill-over effects increasing the productivity of co-workers⁽²⁶⁾. Strengthening human capital and its formation may be crucial to strengthen European firms' comparative advantage on international markets in the face of increased global competition and the knowledge economy, as developed in section 4. However, investing in human capital formation through education alone is not enough. Appropriate skill-development and skill-anticipation policies and working conditions (i.e. ensuring good skills matching and the best use of the accumulated human capital) are crucial.

There is a wide variation between Member States in terms of their efforts to strengthen skill development. Denmark, Sweden and Finland perform the best across all the selected indicators (See participation in life-long learning (Chart 8), on-thejob training (Chart 9) and new learning opportunities on the job (Eurofound, EWCS 2010, question 49f)). The lowest participation rates on lifelong learning are found in Bulgaria, Romania, Croatia and Greece. Spain and Italy perform poorly in terms of on-the-job training. These countries also show the poorest outcomes in other indicators of skills development(²⁷). Bulgaria, Romania, Cyprus, and Greece also rank the lowest of all EU Member States of the OECD in the latest PISA test (2012)(²⁸). Note that less effective training systems and an inappropriate skill mix due to weak training and skill-anticipation policies can lead to lower productivity and output and result in persistent labour market structural problems (fragmentation, polarisation).

The recent crisis has affected participation in life-long learning in around one third of the Member States, but in different ways (Chart 8). Sweden, France, Luxembourg and Portugal saw an increase, while the United Kingdom and Slovenia saw the highest declines (29). Employers may tend to increase training during a recession because training costs, including opportunity costs (lost productivity is less problematic when demand is slack), are lower (e.g. Caponi et al., 2010; Felstead et al., 2011). In addition, difficult conditions may encourage employers to compete on quality or to diversify their products, both of which require increased training efforts (e.g. Felstead et al., 2011). In contrast, a crisis can make employers reluctant to provide training if this is seen as a financial strain with an uncertain return on

- (28) The Member States performing best on the PISA test in 2012 are the Netherlands, Finland, Belgium, Germany and two new Member States (Estonia and Poland). More information about the PISA results is available at http://www.oecd.org/pisa/ keyfindings/pisa-2012-results-overview.pdf
- (29) Based on data from the European Social Survey of 19 countries over the period 2004–10, Dieckhoff (2010) found that the odds of training in 2010 were 20% lower than in 2004, even after controlling for a range of employee and workplace characteristics. However, there were country differences: there was no significant change in the volume of training in any of the Nordic countries, there was an increase in two Continental countries, and there was a decrease in the UK and Ireland and in some of the Eastern European countries.

⁽²⁶⁾ Endogenous growth models illustrate how human capital accumulation increases the growth rate (Lucas, 1988; Rebelo, 1991). Christen et al. (2008) show that differences in job performance between male and female physicians were fully accounted for by differences in their communication skills. Dearden et al. (2006), using a dynamic perspective on skills, show that training which enhances skills is also associated with higher productivity.

⁽²⁷⁾ Percentage of early school leavers (highest shares are in Spain (23.5%), Malta (21%), Portugal (19%), Romania and Italy (17%), Bulgaria (13%); percentage of population with at least medium computer skills (lowest shares are in Romania (21%), Bulgaria (29%), Greece (41%), and Italy (44%). Data source: Eurostat, tables [edat_lfse_14], [edat_lfse_08] and under the link http://epp.eurostat.ec.europa.eu/ tgm/table.do?tab=table&init=1&plugin=1&l anguage=en&pcode=tsdsc460. The data on early school leavers refers to 2013, while data on level of computer skills is from 2012, the latest available at the time of drafting.



Notes: Lifelong learning (LLL) measures participation rate in education and training (last 4 weeks). The Chart shows the relative difference in the life-long learning participation rates between those with high education and those with medium, respectively low, education. It reflects the situation of the population (aged 25–64) engaged in formal or non-formal education and training. 'Low' stands for preprimary, primary and lower secondary education corresponding to levels 0–2 (ISCED 1997); 'medium' stands for upper secondary non-tertiary education corresponding to levels 3–4; and 'high' corresponds to levels 5–6. *No data for 'low' education for 2013.

their investment (e.g. Dieckhoff, 2013; Felstead et al., 2011; Majumdar, 2007).

The low-skilled, who are already disadvantaged in terms of obtaining a job, also receive less life-long learning, see Chart 10. The difference in participation rates between highly and lowly educated people is the highest in Poland, the Czech Republic, Greece, Cyprus and Italy. It is the lowest in Denmark, Sweden, Finland and the Netherlands.

3.3. Good working conditions can attract and develop human capital and improve performance and output

Good working conditions create the environment to attract and develop human capital and improve the performance of workers. A physically safe and healthy working environment leads to fewer accidents and absences from work and, hence, to lower costs (European Commission, 2014; OECD, 2014; Cottini and Lucifora, 2011; Lewis and Malecha, 2011). Furthermore, work-related stress or negative social relations in the workplace may lead to employees working below their full potential, higher distraction levels or neglect of responsibilities, and may affect career-related decisions (Lewis and Malecha, 2011; Mather and Lighthall, 2012). A working environment too focussed on competition may also generate unethical behaviour (Shleifer, 2004; Schwieren and Weichselbaumer, 2010; Gill et al., 2013; Charness et al., 2013)(³⁰).

The EMCO framework distinguishes four sub-dimensions of working conditions and organisation: health and safety at work; work intensity; work autonomy; and collective interest representation(³¹).

- (30) Work-related psychological disorders and mental health problems were behind 42% of all early retirements of white collar workers in Austria in 2009 and the main reason for long-term sick leaves in the Netherlands (55 days on average) in 2010 (European Commission 2014 - Social Agenda 02/2014, p. 9). High psychological job demands, long working hours and poor physical environment are detrimental to the mental and physical health of workers (e.g. increasing obesity) and can influence the health status of the worker's family (Morrissey et al., 2011; Cottini and Lucifora, 2011). Lewis and Malecha (2011) find that negative social relations in the workplace have detrimental effects on the productivity of nurses. Mather and Lighthall (2012), reviewing the literature on mental stress and reward processing, find that overly stressed employees are more likely to be distracted and may neglect to adjust their working habits after negative feedback from their hierarchy, Halko et al. (2014). Shleifer (2004), Schwieren and Weichselbaumer (2010), and Gill et al. (2013) suggest that competitive pressures at the workplace notably compensation-related, can lead to greater risk-taking by men and lower risk-taking by women (shyness to compete for promotion and under-representation in leading positions), and can increase cheating, sabotage, corruption, excessive executive pay and corporate earnings manipulations with no or a negative effect on productivity.
- (³¹) For the 'working conditions' dimension the EMCO set of indicators relies mostly on the EWCS questions. One should note that while they relate to objective outcomes, the indicators reflect people's feelings and perceptions about their working environment. However, this adds valuable information to the comprehensive picture about the general labour market conditions.

3.3.1. Reducing health and safety risks may increase overall productivity

Health and safety at work can have a direct impact on employers' costs and employees' productivity, absenteeism and job satisfaction. While the incidence of work accidents has declined in recent years, significant differences across different groups of workers can be observed. Chart 11 shows the relative accident rate of those with medium (alternatively high) education to those with low education. It can be seen that the lower the education level, the higher the accident rate. More generally, those with lower levels of education are more often in jobs that present greater risks in terms of health and safety conditions at work(32).

Note that important structural changes will likely bring along new products and production processes with potentially unknown health and safety risks which may need to be borne in mind, as discussed in section 4.

3.3.2. Combining work autonomy with work intensity can increase productivity

Work intensity (³³) and work autonomy are two important characteristics of work organisation that can affect workers' performance through their impact on the level of motivation, stress and physical and mental health. They can also impact the labour market participation decisions of particular groups such as older workers, second earners with children and/or people with disabilities. By reinforcing positive interactions between work intensity and work autonomy, an organisation can achieve greater effort from its employees, thus increasing productivity and output.

^{(&}lt;sup>52</sup>) The Chart refers only to accidents rate, while in many jobs work-related health and safety risks are much broader, including respiratory diseases, skin conditions, musculoskeletal disorders, etc.

Note that in this subsection 'work intensity' (33) is used in line with the sociological literature in the sense of a characteristic of work organisation, rather than in the most narrow sense used by Eurostat (the indicator persons living in households with low work intensity is defined as the number of persons living in a household having a work intensity below a threshold set at 0.20). The work intensity of a household is the ratio of the total number of months that all workingage household members have worked during the income reference year and the total number of months the same household members theoretically could have worked in the same period. More details at http://epp. eurostat.ec.europa.eu/statistics explained/ index.php/Glossary:Persons living in households_with_low_work_intensity.



gif

Source: Eurostat ESAW 2009, table [hsw_ac1]. 'Low' stands for pre-primary, primary and lower secondary education corresponding to levels 0–2 (ISCED 1997); 'medium' stands for upper secondary non-tertiary education corresponding to levels 3–4; and 'high' for levels 5–6.



In general, work intensity does not need to have a negative connotation. Arguments emphasising the negative effects of work intensity on employees' wellbeing focus on "constrained" work intensity, where employees have little choice about the effort they put into their work. Higher work intensity may be a result of organisational policies, such as management strategies, supervisory pressures or machine pacing, but may also reflect individuals' choice (e.g. Gallie and Zhou, 2013). Some degree of work intensity is an inherent part of creative effort, providing a challenge that enables people to develop their skills (Gallie and Zhou, 2013).

Empirical research indicates that the combination of high work intensity and low job autonomy increases work stress and can severely impact employees' physical and mental health. Excessive workloads and unclear or conflicting demands on the job-holder, combined with the lack of role clarity, lack of involvement in decision-making, lack of influence over the job design, poorly managed organisational change and job insecurity, lead to psychosocial risks and physical and mental ill health, in particular depression, burnout and cardiovascular diseases, and therefore lower productivity and output (Karasek and Theorell, 1990; Theorell and Karasek, 1996; Marmot, 2004; Theorell, 2007; European Commission, 2014d; OECD, 2014).

As Chart 12 shows, according to the Eurofound EWCS (2010), in some Member States (Bulgaria, Poland, Latvia and Lithuania) people appear not to experience stress and do not work at high speed nor to tight deadlines. In contrast, in Sweden, Germany, Austria, Greece and Cyprus people work at very high speed, to tight deadlines and under stress. However, the level of self-responsibility is also much higher in the Nordic countries. Note, though, that measuring exposure to stress across the EU is not straightforward, since workers' perception of stress may be affected by cultural differences, their understanding of the notion of stress or their propensity towards admitting to stress.

Chart 13 links the three dimensions of working conditions: work intensity, work autonomy and the level of job stress. The Chart shows that there are two groups of countries that are characterised by a low level of stress: one where job control and work intensity are low (e.g. Bulgaria and Lithuania); and one where the 'demands' of the job are high but are compensated by a high level of self-responsibility (e.g. the Netherlands and Denmark). High levels of stress are experienced in Germany, Cyprus and Austria, where the 'demands' are among the highest but levels of self-responsibility are relatively low (³⁴). Unsurprisingly, there are no countries with high autonomy and low 'demands'.

³⁴) Sweden represents an exception: even though it is in the yellow circle, stress is perceived to be high in Sweden regardless of the high level of job autonomy. However, if one looks at the separate indicators behind the composite factor of self-responsibility, one can see that the control over the speed of own work (EWCS question 50c) is remarkably low, the third lowest in the Union. This may convey the impression of time pressure and explain the registered high levels of stress in the country.



excel file

Box 2: Stress, happiness and productivity

In 2014, stress was the second most-reported work-related health problem in the EU. A 2013 European opinion poll conducted by EU-OSHA(³⁵) found that more than half of all workers considered work-related stress to be common in their workplace. The most common causes of work-related stress are job reorganisation or job insecurity (72%), hours worked or workload (66%), being subject to unacceptable behaviour such as bullying or harassment (59%), lack of support from colleagues or superiors (57%), lack of clarity on roles or responsibilities (52%), and limited possibility of managing one's work patterns (46%) (European Commission, 2014d). In the Enterprise Survey on New and Emerging Risks (2010)(³⁶), around 8 in 10 European managers expressed concern about work-related stress in their workplaces, though less than 30% admitted having implemented policies to deal with its risks. Between 50% and 60% of all lost working days are related to stress and psychosocial risks.

The question of whether happiness makes people more productive occupies economists, behavioural scientists and policy makers. The well-being of employees concerns many company managers. For example, "At Google, we know that health, family and wellbeing are an important aspect of Googlers' lives. We have also noticed that employees who are happy demonstrate increased motivation ...[We] ...work to ensure that Google is ...an emotionally healthy place to work" (Lara Harding, People Programs Manager, Google). Several studies show the link between positive mood and productivity (Oswald et al., 2014)(³⁷), between wellbeing and motivation and higher capacity to solve anagrams (Erez and Isen, 2002), and between job satisfaction and value added per hours worked in manufacturing (Boeckerman and Ilmakunnas, 2012).

- (³⁵) See reports in figures at https://osha.europa.eu/en/safety-health-in-figures.
- (³⁶) Results and publications available at https://osha.europa.eu/en/esener-enterprise-survey.
- (³⁷) Oswald et al. (2014) set up three short (five-minute) GMAT-style maths experiments on more than 700 individuals whose mood was measured and then manipulated with video clips, snacks and drinks. The measurements took into account negative real-life events in the previous five years (e.g. bereavement and family illness). The study concluded that those made happier had productivity gains of 12%, while individuals who suffered a major real-life shock in the preceding five years showed lower productivity.

3.3.3. Job autonomy can boost productivity

Job control or autonomy(³⁸) is a core factor in determining the guality of work. Several studies report that workers who are free to make choices in the workplace and are accountable for their decisions are happier, more committed, put more effort into their work, and are therefore more productive and show a lower tendency to quit their job (Chirkov et al., 2011 for a review; Mahdi et al., 2012; Gellatly and Irving, 2001; Langfred and Moye, 2004). This is especially the case when the work is complex or requires more creativity, though in a very routine job, autonomy can still increase satisfaction and reduce turnover (DeCarlo and Agarwal, 1999; Finn, 2001; Liu et al., 2005; Nguyen et al., 2003; Thompson and Prottas, 2005). Job autonomy has also been seen as an important factor in moderating the impact of work intensity (Liu et al., 2005).

Chart 14, Chart 15 and Chart 16, based on the Eurofound EWCS (2010), show that job autonomy is the highest in Sweden, Denmark, Finland and the Netherlands. It is the lowest in Cyprus, Greece, Portugal and Bulgaria. Germany and Austria score among the lowest on the perceptions of the level of self-responsibility (Chart 16)(39). Gallie and Zhou (2013), using European Social Survey data⁽⁴⁰⁾, report similar country patterns that are stable over time. The authors explain this stability over time with the fact that job autonomy is embedded in wider institutional structures. In some countries, job autonomy and control have been embedded for many years at company as well as national level institutions.

- (³⁹) Germany scores low in terms of control over the speed of own work (EWCS question SOc), order of tasks (SOa), employee consultation on targets (S1c), ability to apply own ideas (S1i) and employee involvement in improving work organisation (S1d). Austria scores low in terms of control over speed of work, ability to apply own ideas and involvement in improvements of work processes.
- (40) There are three items in the ESS that provide a measure of job control: how much 'the management at your work allows you (a) to decide how your own daily work is organised; (b) to influence policy decisions about the activities of the organisation; and (c) to choose or change your pace of work'. The items then cover not only immediate control over the work task (task discretion), but also people's perceptions of wider influence over organisational decisions.

^{(&}lt;sup>38</sup>) Job autonomy can take different forms depending on the country context and the organisational culture. Organisations may let employees set their own schedules or choose how and where to do their work.









your work; q510 — can influence decisions that are important for your work.

file excel

ij

Note: No observation available for HR.

Social dialogue 3.3.4.

By promoting win-win solutions for employers and workers, social dialogue(⁴¹) plays an important role in the improvement of working conditions. Throughout Europe employers' and workers' representatives combine their expertise on work-related matters to promote job quality(42).

Workers' and employers' representatives are uniquely well-placed to identify skill needs and promote lifelong learning. Social partners play a key role in European Sector Skills Councils, which are designed to anticipate the need for skills in specific sectors more effectively and achieve a better match between skills and labour market needs(43). European social partners have concluded a number of skills-related autonomous agreements, which national social partners implement in accordance with procedures and practices specific to management and labour in the Member States (44).

- (43) See http://ec.europa.eu/social/main jsp?catId=784
- (44) Examples include a European licence for drivers on interoperable services (railway sector), training standards and European certificates (hairdressing), or core competences for process operators and firstline supervisors (chemical sector).

Employers and workers have a joint interest in promoting safe and healthy workplaces. EU cross-industry social dialogue led to agreements on stress and violence at work, while EU sectoral social dialogue led to sectors-specific agreements or campaigns(⁴⁵). With the support of the European Agency for Health and Safety at Work, social partners at European and national level cooperate to develop 'Online Interactive Risk Assessment' (OiRA) tools(46).

Europe has a rich tradition of social dialogue on working time, contractual arrangements and the reconciliation of work and family life. Framework agreements cover a large number of areas from parental leave(47) and working time, to equal treatment between parttime workers and full-time workers and between fixed-term contract workers and those on open-ended contracts(48).

A number of EU Directives establish minimum requirements regarding information and consultation of workers at

- Established at cross-industry level, giving all (47) employees an individual non-transferable right to parental leave was first signed by European social partners in 1995, revised in 2009
- (48) Each of these cross-industry agreements has been made legally binding through Council Directives. The same applies to a number of sectoral agreements on working time of mobile workers, including sea farers, mobile civil aviation staff and mobile workers assigned to interoperable cross border rail services. A recent agreement between social partners of the inland waterways sector has been forwarded to the Council for implementation by directive.

⁽⁴¹⁾ Social dialogue refers to discussions. consultations, negotiations and joint actions involving organisations representing the two sides of industry (employers and workers).

^{(&}lt;sup>42</sup>) This section cannot exhaustively cover all social partners' activities at company, sectoral national and European level Rather, it focuses on a number of key initiatives at European level. Interested readers will find additional information in the 'Industrial Relations in Europe' series published by the European Commission (e.g. European Commission 2010b and 2013d), and in publications by the European Foundation for the Improvement of Living and Working Conditions (e.g. Eurofound 2014a).

⁽⁴⁵⁾ For instance in the hospital sector http://ec.europa.eu/social/main.jsp?catId=521 &langId=en&agreementId=5136

⁽⁴⁶⁾ These tools can beln micro and small organisations to put in place a step-by-step risk assessment process - starting with the identification and evaluation of workplace risks, through to the decision-making and implementation of preventative actions, to monitoring and reporting.

company level (49). A recent fitness check of these directives (50) found that information and consultation of workers at company level can contribute to solving problems at work, engage workers in changes in work organisation and work conditions, appease conflicts, promote trust and partnership, increase job satisfaction and commitment, reduce the rate at which workers leave the company, and improve the physical health and well-being of workers. It was also found that information and consultation has a positive impact on staff performance and on the company's competitiveness and reputation.

Finally, setting wages is one of the key functions of industrial relations systems in the EU. Despite a tendency for the company level and individual bargaining to gain importance, multi-employer bargaining remains important in many European countries. Beyond the main level of bargaining, it is important to consider coordination of different processes, both vertically (between different levels) and horizontally (between units at a given level, e.g. between companies).

While the promotion of dialogue between management and labour is an objective of the European Union (Article 151 TFEU), there is no single model of social dialogue in the EU. Across Europe, there exists a large diversity of national industrial relations systems, which the Union has to take into account when promoting social dialogue at its level (Article 152 TFEU). Industrial relations should be considered as complex systems whose institutions interlock, which cannot be measured along a single dimension or in a single statistic. There are different qualities, each with different effects on the regulation of the economy and the labour market. In whichever form, social dialogue makes an important contribution to job quality, both directly as a key dimension of a 'good job' and through its positive impact on working conditions.





3.4. Work-life and gender balance to strengthen participation, efficiency and equity

3.4.1. Work-life balance

Insufficient opportunities (⁵¹) for combining work with other private and social responsibilities may lead to higher inactivity rates among certain groups of the population (e.g. older people, persons with family responsibilities), with potential consequences in terms of social exclusion and greater dependence on social protection systems.

The inactivity rate due to family responsibilities is the highest in Malta, Cyprus, Spain and Ireland (Chart 17), followed by Luxembourg and the United Kingdom. It has decreased over time in Spain, Cyprus, and Greece though. In the United Kingdom, supply of childcare facilities is around the EU average (Chart 18) but childcare costs are high(⁵²). In contrast, Denmark, Sweden, France, Estonia, the Netherlands and Portugal have some of the lowest inactivity rates due to family responsibilities (Chart 17), as well as more readily available childcare facilities (Chart 18) and at an affordable cost.

(50) The results of the fitness check were published on 26 July 2013 in a Commission Staff Working Document available at http://ec.europa.eu/social/main.jsp?langld= en&catld=707&newsld=1942&furtherNe ws=yes

(⁵¹) For example, availability of easily accessible and affordable childcare facilities, voluntary part-time work patterns, parental leave, adaptations for older workers or workers with disabilities, etc.

⁽⁴⁹⁾ In particular, Directives 98/59/EC on collective redundancies, 2001/23/EC on transfers of undertakings and 2002/14/EC on a general framework relating to information and consultation of workers.

⁽⁵²⁾ The 2013 Northern Ireland Childcare Cost survey (Dennison, 2013) shows that a full-time childcare place (50 hours) costs GBP 213 per week in Britain. In Northern Ireland, childcare costs are also high, and increased to GBP 158 per week in 2013. Moreover, 35% of the nurseries in Ireland charge the same price or even higher for a part-time place than for a full-time one.



3.4.2. Gender balance

available for DE and HR in 2002

Gender stereotypes can lead to lower labour market participation of women, fewer job opportunities for women and lower pay which, in turn, may lead to a higher risk of social exclusion. The gender pay gap is an important indicator of persistent discrimination in the labour market. The unadjusted gender pay gap stood at 16.4% in 2012 in the EU as a whole. For example, WiiW (2014), using EU-SES data, estimates (53) that in 2010 the adjusted gender pay gap ranged from around 5% in Romania and Bulgaria to more than 15% in Estonia (Annex 3, Chart A3.11). The median adjusted gender pay gap decreased from 13.2% in 2002 to 10.4% in 2010. The adjusted gender wage gap has declined over time in all except four Member States (Lithuania, Poland, Portugal and Slovakia).

Chart 19 shows the extent to which gender pay gaps contribute to overall wage inequality. On average, gender differences contribute about 3% to overall inequality, though there are important crosscountry differences ranging from more than 6% in Finland to less than 1% in Bulgaria and Romania. The contribution of gender differences to inequality fell between 2002 and 2010 in all countries except Lithuania, Poland and Portugal. The declines were particularly large in Cyprus (from initially around 9% in 2002 to only 3% in 2010), the Czech Republic (from

(53) Correcting for differences in age, education, contract type, occupation, enterprise type (private, public), firm size, industry and country. initially 8% to 4% in 2010) and Italy (from 5% in 2002 to 2% in 2010).

3.5. Summary of findings

The literature review suggests that higher productivity can be attained through adequate levels of earnings; higher job security; higher education and life-long training, including on-thejob training; good working conditions - a safe and healthy working environment, an appropriate balance between work intensity and job autonomy and greater employee participation and empowerment, including social dialogue; and better work-life and gender balance. These can strengthen human capital formation, including firmspecific human capital, and increase motivation, commitment and effort. They can reduce accidents, absenteeism and stress, induce creative effort, foster cooperation and generate positive externalities on co-workers. They may also contribute to fostering higher labour market participation and longer working lives, particularly of certain population groups (e.g. older workers, those with family responsibilities or disabilities), reducing dependency on social security systems and ensuring greater social cohesion.

EMCO indicators show that job quality varies across EU Member States and may have deteriorated during the crisis in several dimensions.

 In-work poverty increased in most countries during the crisis. In-work poverty is high in Romania (due to low earnings), Italy, Spain and Greece (due to a high proportion of people in low work-intensity households).

- The crisis increased the share of involuntary temporary work and impeded the transition rates to permanent contracts in many Member States. There is a high share of people on involuntarily temporary contracts in Italy, Spain, Greece, Portugal, Cyprus, Romania and Slovakia. Transitions from temporary to permanent contracts are difficult in Italy, Spain, Greece, Portugal and Cyprus, and in most Member States are more impeded for women than for men. Austria, Germany, the Netherlands and Estonia have the lowest rates of involuntary temporary work and high transitions rates to permanent employment.
- The lowest participation rates in life-long learning are found in Bulgaria, Romania, Croatia and Greece, while Denmark, Sweden and Finland have the highest participation rates. Spain and Italy perform rather low on on-the-job training.
- Low work intensity and low autonomy lead to low levels of perceived stress in Bulgaria and Lithuania; high work intensity and job autonomy generate average stress in Netherlands and Denmark; high intensity with low autonomy lead to high levels of stress in Germany, Austria and Cyprus.
- The unadjusted gender pay gap stood at 16.4% in 2012 in the EU as a whole. Inactivity rates due to family responsibilities are the highest in Malta, Cyprus, Spain, Ireland and UK, and the lowest in Denmark and Sweden. They decreased in most Member States during the crisis due to increased strain on family budgets.

4. STRUCTURAL CHANGES CAN IMPACT ON JOB QUALITY AND PRODUCTIVITY GROWTH

Job quality can have a direct impact on labour productivity and labour market participation and both are crucial to the success of the European social market economy. This section identifies future challenges to job quality and labour market outcomes (labour market participation, productivity) brought about by a range of structural changes such as further technological progress and innovation, further globalisation, demographic change and the general greening of the economy. The section assesses to what extent labour market polices can reinforce positive developments and prevent or correct adverse outcomes associated with those changes and which are to a large extent conditioned by labour market institutions (e.g. social dialogue mechanisms, wage bargaining systems, minimum wages schemes, employment protection legislation, unemployment insurance, active labour market policies and life-long learning) and the business cycle (⁵⁴).

This section starts by assessing the extent to which further innovation in information and communications technologies (ICT) and key enabling technologies (KETs)(55) may affect job quality. It focus on the job quality potential of an industrial renaissance, the role of small and medium enterprises (SMEs) and the risk that skill and talent biased technological progress may involve an unequal distribution of the costs and benefits between low and medium-skilled workers and high-skilled workers. In other words, technological progress has strong potential to improve productivity but may have a polarising effect in terms of job quality, impeding further technological progress and productivity growth and generating inequalities.

Next, the section looks at globalisation (associated with changes in international trade, foreign direct investment and labour mobility) and its potential to increase productivity and hence earnings. Again, costs may be incurred primarily by the most vulnerable workers, such as the low-skilled and employees on temporary contracts. They may experience stronger job insecurity (e.g. due to offshoring or relocation) and lower wages (e.g. to compete with countries with an abundant supply of low-skilled workers). Such adverse outcomes may, in turn, have negative feedback on productivity and labour market participation if they reduce workers' commitment, motivation, abilities and upward job mobility.

The section then focuses on the opportunities and challenges for job quality brought about by an ageing population and high youth unemployment. These developments pose some important labour market policy challenges related to active ageing, gender equality, workprivate life balance and discrimination, which may have a negative impact on employment and productivity if they hinder the optimal allocation of resources.

Finally, the section explores the policy challenges and opportunities related to job quality in the transition to a greener economy. The shift to a green and resource-efficient economy is above all an opportunity to support sustainable and high-quality employment, while contributing to the recovery from the recent economic crisis. However, better targeting and coordination of labour market measures and tools are essential in order to create the necessary conditions to bridge skill gaps and overcome labour shortages, manage restructuring, anticipate change and emerging health and safety risks (especially for low-skilled manual workers), and ensure gender balance. These may have an important impact on workers' performance and participation in the production of new green goods and services.

4.1. The two sides of knowledge and technology-intensive growth

This subsection focuses on challenges posed by technological progress on job quality. It starts by assessing the increasing importance of knowledge and creativity in the future labour market and the risks associated with the automation of tasks, such as jobs losses and labour market polarisation. It investigates the extent to which an industrial renaissance associated with the potential for further innovations in ICT and KETs has the ability to generate more and better jobs. It looks at the role of SMEs and the challenges they face in improving job quality in the context of technology innovation. It then discusses the role of labour market policies in tempering labour market polarisation driven by technological progress.

4.1.1. Technology change and innovation will change the job landscape of the future and can render jobs obsolete

Technological progress is a key defining factor in how goods and services are produced and delivered to consumers. The fact that production processes are changing is by no means new, but the speed of that change may be. It can take decades for a new invention to be applied, but when it is applied, changes accelerate. The typewriter was invented in the 1860s but was not introduced into the office until the early 20th century, when it joined a wave of mechanisation, with Dictaphones, calculators, mimeo machines, address machines, and the predecessor of the computer — the keypunch (Frey and Osborne, 2013, after Beniger, 1986; Cortada, 2000). There are many signs that the cumulative effect of advancements in information sharing, computing power, machine learning, machine vision and data mining will soon accelerate the changes in terms of the types of jobs that are needed, how rewarding these jobs are and the requisite organisational arrangements.

In the not-so-distant past the switchboard operator became obsolete due to direct number dialling, the copy typist gave way to personal word processing, the bank teller was replaced by cash machines, the travel agent fell prey to online booking systems and many car assembly line workers were replaced by industrial robots. Deindustrialisation and relocation to low-cost countries have been shaping the economic landscape and labour markets of the high-income countries over the past forty or so years. A long-term decline of heavy industries such as mining or steel production has been observed, while specialised hightech industries have been holding ground even if employing fewer workers per output (automotive manufacturing being one of many examples).

However, current changes are expected to have a stronger and polarising impact on labour markets (e.g. Acemoglu and Autor, 2010; Eurofound, 2013). Currently, technology is changing the face of education through online lectures classes

⁽⁵⁴⁾ A macroeconomic downturn may reduce job quality through: lower job security, lower skill formation, stronger health and safety risks, more involuntary temporary/part-time labour contracts, distorted work-private life and gender balances (e.g. Eurofound, 2012a; RWI, 2014; Tahlin, 2013; Dieckhoff, 2014; Johnson, 2012; McGinnity and Russel, 2013; Ravn and Sterk, 2013; Gallie, 2014).

⁽⁵⁵⁾ Key enabling technologies (KETs) enable the development of new goods and services and the restructuring of industrial processes needed to modernise EU industry and make the transition to a knowledge-based and low-carbon resource-efficient economy. They play an important role in the R&D, innovation and cluster strategies of many industries. More particularly, KETs cover micro-/nano-electronics, nanotechnology, photonics, advanced materials, industrial biotechnologies. See European Commission (2012b) and HLGKET (2010).

and learning resources that are available globally and often at a fraction of the cost. Digital applications have shown the ability to compete with and potentially undermine various traditional service providers such as taxis or hotels (e.g. ridesharing app 'Uber' or 'AirBnb' flat rental and sharing). Computerisation, typically confined to manual and cognitive routine tasks, is now spreading to activities that were commonly defined as non-routine (e.g. Autor and Dorn, 2013; Goos et al., 2009). Tasks regarded as non-routine only a decade ago have since been computerised at a rapid pace (Autor, et al., 2003; Markoff, 2011; Frey and Osbourne, 2013). Recent examples of how the boundary between routine and non-routine tasks and between automatable and nonautomatable routines will be pushed further by technology include handwriting recognition, machine translation and the use of language analysis to identify general concepts in documents (56).

Authors speculate about the scale of the challenge ahead if new technologies mature and spread beyond prototypical and experimental applications: selfdriving vehicles, health diagnostics, automated call centres and robot-assisted remote surgery are some examples. The impact may spread to related sectors. For example, self-driving vehicles can reduce drivers' jobs and, if safer, reduce business opportunities in the insurance sector.

In some sectors, there are already palpable signs of rising automation in work spaces such as container ports, logistics warehouses or even hospitals (e.g. robots pulling trolleys with meals, medicines and blood samples in hospitals (Bloss, 2011) or climbing wind turbines much faster than a human and inspecting the blades 100 metres above ground (Robotics-VO, 2013)).

While intellectual and knowledge work (e.g. computer programming) is flourishing and craftsmanship-based manual trades remain in high demand, many middle-class occupations, typical of the industrialised societies of the latter half of the 20th century, are being eroded. Programmable machines are expected to take over many routine and less routine tasks, many of which are performed by unskilled and semiskilled industrial and clerical service workers who typically occupy the middle layers of employment. Some studies strike an alarmist tone and argue that the process has only just begun. Frey and Osborne (2013) predict that 47% of current jobs in advanced economies like the United States are at risk of being automated over the next 20 years.

Further technological change is therefore expected to have a strong and polarising effect, affecting jobs and skill levels in a different manner (see below). In this context, managing the transition into a new labour market where many jobs succumb to automation must become a key priority for policymakers.

4.1.2. Occupations resilient to automation: the importance of knowledge and creativity (human capital) in view of technology change

The non-routine jobs that are likely to resist automation in the foreseeable future are located at either the lower or higher end of the wage and skill spectrum. At the lower end, there are services such as hospitality, care, beauty, cleaning, customer service, construction, decorating and installation. These may be subjected to some vocational training and licensing in particular legal settings but require soft skills such as empathy, improvisation and complex decision making. Further, they feature complex manual tasks which in turn rely on specific skills and experience. These jobs are not suited to outsourcing since they have to be performed on-site.

Despite their undisputed social utility, such non-routine, manual, low- to medium-skilled jobs often offer modest remuneration with precarious job arrangements and physically demanding working conditions. Likely reasons for this are the abundant labour supply, the possibility of using underpaid migrant workers and, in some cases, the threat to relocate some part of these tasks to lowwage countries (Standing, 2011). In this context, there is clearly a need to step up efforts to improve the working conditions in these jobs and to ensure the application of existing worker protection laws.

At the high end of non-routine and nonautomatable jobs are those consisting of complex cognitive tasks and a high level of professional competence, usually combined with a long and versatile formal education (e.g. computer programmers, creative industries, engineers, managers, investment bankers, lawyers, doctors, teachers and scientists). Europe has great stakes in developing the knowledge-based economy, investing in high-end skills and assuring optimum job conditions for knowledge workers. Compared with low-skilled workers, knowledge workers already enjoy a more privileged position on the labour market, with more favourable working conditions and a higher pay. Yet, the knowledge sector is where the highest potential for productivity growth is likely to lie. Hence, a focus on more efficient working arrangements will be key to securing Europe's position as a hotspot of high productivity.

4.1.3. Technology change can lead to a possible industrial renaissance in the EU

In the recent past, increasing job losses and the rise in job uncertainty have affected job quality, particularly in industry. For example, the employment share of the industry sector in the EU as a whole dropped from 22.1% in 2000 to 17.7% in 2013. At the same time, jobs in industry typically offer a high wage level (compared with the national average wage): average gross wages in industry were 10.6% above the national average gross wage in the EU. The drop in industry shares and high wages are a combined effect of: a) strong productivity growth in industry; b) the opening of world markets and changing business models, whereby manufacturers outsource certain tasks (such as logistics, marketing or legal advice); and c) a shortage of skilled human capital in engineering and science which may have been aggravated by the recent crisis that stifled access to funds for innovation (e.g. European Commission, 2013).

A variety of policy measures have been implemented to temper the adverse socioeconomic impact of delocalisation and offshoring (e.g. Eurofound s.a.)⁽⁵⁷⁾. These initiatives have primarily been used to accommodate the ongoing job shift from industrial activities to other

⁽⁵⁶⁾ For instance, Symantec's Clearwell system proved to be capable of analysing (conceptual contents, not just words) and sorting more than 570 000 documents in two days.

⁽⁵⁷⁾ At http://www.eurofound.europa.eu/areas/ industrialrelations/dictionary/definitions/ restructuring.htm http://www.eurofound. europa.eu/areas/industrialrelations/ dictionary/definitions/restructuring.htm

activities notably in the services sector (though not necessarily associated with higher job quality).

An important policy challenge will be to exploit the future job growth potential of emerging innovations in ICT and KETs, such as bio-based products, smart vehicles, sustainable construction and smart grids. Where future developments are characterised by a shift from mass-produced goods and services to more customised high-quality goods, there is strong potential for the resource-poor, skills-rich EU to create high quality and value added jobs.

SMEs will have an important role to play in this industrial renaissance since they are a major source of job creation and innovation. Workers' performance is largely determined by the scope with which educational systems are complemented by in-work training (see Chapter 2). Therefore, job training in SMEs will be important to ensure their workers' productivity and their international competitiveness. In this context, SMEs face very specific challenges that may reduce their efforts to reinforce their workers' educational attainment. Indeed, compared to larger firms, SMEs have fewer internal human and financial resources for skill development both at managerial and lower personnel level. Therefore, improved regulation of credit markets and lending conditions for SMEs is crucial to ensuring sufficient access to credit for skill formation. Where financial markets may fail to provide such finance, public funding should be considered.

4.1.4. Technology change can produce unbalanced outcomes in the population: stronger labour market polarisation

As discussed, there is a risk that the skill and talent-biased industrial renaissance brought about by strong technological changes will sharpen the ongoing labour market polarisation. It is expected that further digitalisation of economic activity, in combination with globalisation (see below), will increase the demand for highly skilled workers, increasing their job security and earnings, while the opposite may be observed for low- to medium-skilled groups and the long-term unemployed. The coming technology change is expected to benefit the strongest talents disproportionately, i.e. the 'superstars' that create services such as Facebook, for which there is a very strong demand (e.g. Brynjolfsson and McAfee, 2014).



Source: DG EMPL calculations based on Eurofound (2013) available at http://www.eurofound.europa.eu/ emcc/ejm/summary.htm

Notes: LTI: low-tech industry, LKIS: low knowledge intensive services, HTI: high-tech industry, KIS: knowledge intensive services. No data for BG, MT, PL or RO.

Such developments will reinforce ongoing labour market polarisation in the private sector in the EU economy (Chart 20). The left pane of Chart 20 shows changes in the earnings quintiles in low-tech industries and basic knowledge services, while the right pane of Chart 20 shows changes in the earnings quintiles in the high-tech and knowledge intensive services. Blue bars show the pre-crisis period (i.e. 1995–2007), while the red bars show the period since the onset of the crisis (i.e. 2008–2010).

In the run-up to the crisis, the lowest guintile within the low-tech and basic knowledge services showed the strongest increase, while the highest quintile within the high-tech and knowledge intensive services showed the strongest increase. In both sectors, the other quintiles showed more modest increases. During the crisis, the same pattern can be observed but in the context of employment reduction: the lowest quintile within the low-tech and basic knowledge services showed the weakest decrease, while the highest quintile within the hightech and knowledge intensive services showed an increase.

Uncertainties about projecting future developments in employment and earnings distribution remain. For example, some analysts, e.g. Gordon (2014), claim that today we are facing the first phase of a secular stagnation as future innovation will not carry the same productivity growth potential as past innovations related to the use of power generation, chemistry, etc., and that the observed changes in employment distribution generated by information technology will be short-lived. This view is in sharp contrast with, for example, Brynjolfsson and McAfee (2014) who argue that the ongoing 'digital revolution' (characterised by exponential growth in computing power, digital information and supply of relatively cheap devices which leads to new business opportunities) carries an even stronger potential for sustainable innovations and growth than the past 'industrial revolution' — though its benefits will not automatically be distributed in an equitable way.

excel file

gif

At the same time, others, e.g. Autor (2014), emphasise that employment polarisation does not automatically lead to wage polarisation since the latter is also determined by 1) degree of complementarity (e.g. performances of workers may improve significantly to the extent they can be complemented with the computing power of machines), 2) the price- and income-elasticity of demand for services (e.g. low price elasticity for intensive manual work allows for stronger wage increases for these service providers) and 3) elasticity of labour supply (e.g. it usually takes more time to form highly-skilled workers than to train intensive manual workers).

This uncertainty requires permanent monitoring and assessment of developments in the field of technological progress.

4.1.5. The role of adequate labour market policies

In this context, the potential for technology change to improve job quality will require proactive labour market policies developed in synergy with other policies (⁵⁸) to support the reallocation of labour towards these new activities in a secure way and in a way that benefits all employees, especially the lowskilled. For these workers, adequate earnings could be provided in the short to medium run by measures that have a direct impact on wages (such as the minimum wage paid by the employer), hiring subsidies (paid by the taxpayer to employer) or by social transfers or fiscal benefits (paid by the taxpayer to the employee).

However, since differences in gross earnings are largely driven by differences in productivity(⁵⁹), closing the earnings gap through nominal unit labour cost increases⁽⁶⁰⁾ may be unsustainable for companies. In other words, to keep workers with excessive labour costs (relative to productivity) may be financially unsustainable to companies, especially in the face of increased international competition. Therefore, it is crucial immediately to reinforce the incentives and opportunities for skills formation and life-long learning directed at the low skilled and both inside and outside the job (European Commission, 2010a). These productivity-enhancing measures should complement the wage/income and other targeted measures towards workers at the lowest end of the earnings distribution (e.g. access to support services, such as child and elderly care).

Anticipating future changes in jobs and associated skill needs will remain a challenge, requiring a stronger collaboration between stakeholders (including employers, employees, education providers and skills forecasters) and better support for job mobility including through better information flows on job availability and the portability of social security benefits (health, pensions).

4.2. Globalisation creates opportunities but also challenges for job quality and productivity

This subsection looks at the potential impact of further globalisation brought about by the removal of barriers to free and fair trade, foreign direct investment (FDI) and migration. These are expected to create upward and downward impacts on job quality which have a direct impact on labour market participation and productivity, as the following analysis shows.

4.2.1. International trade may enhance productivity and job quality

Further opening to world markets strengthens countries' ability to exploit their comparative advantages, thereby reinforcing their overall productivity growth. For example, it is estimated that a 1% increase in the openness of the economy generates an increase of 0.6 % in labour productivity the following year, based on an analysis of EU trade flows between 1996 and 2005 (e.g. European Commission, 2007c). These increases in productivity create the potential to raise real wages, which is an important determinant of job quality. In turn, these increases in earnings may strengthen workers' commitment, with further positive impacts on productivity.

Production patterns will change as globalisation, in combination with technological progress, will allow (large) firms to specialise in core activities and delegate much of their non-core activities to global suppliers so as to reduce production costs. For the resource-poor, skill-rich EU this may imply a shift from traditional manufacturing (e.g. agro-food, steel, textiles) to more knowledge- and technology-intensive activities (e.g. hightech business services, haute couture and design, as well as industrial activities such as computing, biotechnology and nanotechnology)(⁶¹). These developments will strengthen workers' opportunities to move to jobs of higher quality and value added (e.g. in terms of earnings or autonomy).

(⁶¹) For example, from 1970–2003, the textile workforce dropped by 60% in the G7 countries (Huwart and Verdier, 2013). However, not all workers (especially the low-skilled) will have the opportunity to benefit from the opportunities created by globalisation (in combination with technological progress). Moreover, increased international competition from firms located in countries with lower iob quality standards and low wages may also result in increased job insecurity (e.g. due to offshoring, restructuring), poorer worker conditions (e.g. in terms of maintenance of hygiene, occupational health and safety norms) and cuts in wage and non-wage labour costs (e.g. severance pay, individual and collective dismissal procedures), especially for workers performing routine tasks in the production of tradable goods and services. Globalisation may then have a persistent adverse impact on job quality in these types of activities.

Nevertheless, several policy instruments can be used to strengthen upward job mobility, including job-searching assistance, skill formation and portability of social security benefits⁽⁶²⁾. Job-searching assistance is a relatively effective, lowcost tool for smoothing the reallocation of labour. However, as the transition to new knowledge- and technology-intensive activities poses new challenges, awareness of job opportunities and skills requirements by workers, employers and employment services can be low.

Hence, European and National platforms that facilitate the exchange of information between all stakeholders should be strengthened to improve the effectiveness of job-searching structures. Another policy is to strengthen the expertise and capacity of employment services to be more proactive and to increase their offer of re-training programmes and other relevant services.

In addition to modernising education and training systems to meet the emerging demand for new skills, equal access to skills formation should be ensured to avoid any further polarisation. Despite a strong political commitment to life-long learning, only half of all European workers underwent training in 2010 (Eurofound, 2010). The figures are particularly low among women, older workers, lower-skilled workers, workers in small companies and workers on temporary contracts.

⁽⁵⁸⁾ Such as investments in innovation, improvements in the functioning of the Internal Market and opening up international markets, mobilising public resources and unlocking private funds, equipping labour force for industrial transformations. See, for example, 'Industrial revolution brings industry back to Europe' at http://ec.europa. eu/enterprise/initiatives/mission-growth/ index_en.htm#h2-4

⁽⁵⁹⁾ At least if competition and information flows are not distorted too much. Notable exceptions are, for example, in 'winnertakes-it-all' games where it is relative (not absolute) productivity which determines earnings, as is the case, for example, for Olympic athletes or employees in the financial sector.

⁽⁶⁰⁾ I.e. nominal compensation per employee adjusted for productivity, whereby gross wages are an important part of compensation per employee.

⁽⁶²⁾ Apart from guidelines for Multinational Enterprises that establish responsible business conduct wherever they operate, as is outlined in, for example, OECD (2011a).

Improving the cross-border portability of social security benefits and pensions, together with better information about rights and assistance and their enforcement, can further reduce institutional barriers to labour mobility and increase the opportunities to exploit job quality to the fullest extent across the EU.

Finally, particular attention will need to be placed on the low- to medium-skilled workers who are in a disadvantageous position in their ability to upgrade their skills to meet the requirements of the new knowledge- and technology-intensive activities and are employed in jobs subject to international competition from countries with lower job quality standards. In such cases, just as with technology, a combination of targeted measures in terms of adequate earnings, support services, targeted skill-formation programmes and appropriate health and safety standards is necessary.

At the same time, a level playing field with trading partners could be assured via, for example, the inclusion in Free Trade Agreements of provisions covering minimum working conditions and the enforcement of national labour laws, with monitoring and enforcement of labour standards, in line with existing good practices. The ILO (2013a) reports a substantial growth in the number of trade agreements featuring labourrelated measures since the mid-1990s as a result of a growing awareness of social and employment effects of trade liberalisation(63). In this context, implementing health and safety at work legislation in the EU but also more globally may be important (Box 3).

Box 3: Promoting Health and safety at work

The EU⁽⁶⁴⁾ has a strong interest in health and safety at work and develops, implements and monitors EU legislation to improve occupational health and safety in all activity sectors. EU legislation seeks to reduce the risk element associated with particular jobs (e.g. magnetic fields), therefore protecting the health of those workers⁽⁶⁵⁾. Part of the EU awareness-raising and legal process is also to promote workers' rights to make proposals to improve their health and safety and to appeal to competent authorities and stop their work in the event of serious danger. The European Commission Strategic Framework on Health and Safety at Work 2014–20⁽⁶⁶⁾ identifies the following key challenges:

- to improve the implementation of existing health and safety rules, in particular by enhancing the capacity of micro and small enterprises to put in place effective and efficient risk prevention strategies;
- to improve the prevention of work-related diseases by tackling new and emerging risks without neglecting existing risks;
- to take account of the ageing of the EU's workforce.

Actions to address these challenges include:

- Consolidating national health and safety strategies through, for example, policy coordination and mutual learning;
- Practical support (technical assistance and practical tools, such as the Online Interactive Risk Assessment tool (OiRA) that assesses sector-specific risks) to micro and small enterprises to help them comply with health and safety rules;
- Evaluating and improving the enforcement ability of national labour inspectorates;
- · Eliminating unnecessary administrative burdens associated with existing legislation;
- Addressing the ageing of the European workforce and improving prevention of work-related diseases associated with new risks such as nanomaterials, green technology and biotechnologies;
- Improving data collection and developing monitoring tools;
- Reinforcing coordination with international organisations (e.g. ILO, WHO and OECD) and partners to contribute to improving working conditions and reducing work accidents and occupational diseases worldwide.

^{(&}lt;sup>64</sup>) Supported by Committees of national experts such as the Advisory Committee on Safety and Health at Work (ACSH), the Scientific Committee on Occupational Exposure Limits (SCOEL) or the Senior Labour Inspectors Committee (SLIC).

^{(&}lt;sup>65</sup>) For more details see Council Directive 89/391/EEC of 12 June 1989 on the introduction of measures to encourage improvements in the safety and health of workers at work, available at http://eur-lex. europa.eu/legal-content/EN/ALL/?uri=CELEX:31989L0391. See also IRE (2014 — Chapter 7).

⁽⁶⁶⁾ For more details, see http://ec.europa.eu/social/main.jsp?langId=en&catId=89&newsId=2053&fur therNews=yes

⁽⁶³⁾ In total, there were 58 agreements with labour provisions in June 2013 — almost a quarter of the total 248 trade agreements currently in force.

4.2.2. Labour mobility and free movement of services within the EU may affect job quality

Strengthening labour mobility and free movement of services (such as posted workers) within the single market can have an important impact on job quality and productivity.

Labour mobility can have a positive impact on job quality as it can reduce the risk of unemployment and increase employment opportunities in more productive activities, thus yielding higher earnings for the workers involved. This is especially true for workers in sectors that are vulnerable to ongoing structural changes (e.g. energy-intensive industries). Nevertheless, labour mobility within the EU is still only taking place on a small scale despite the considerable opportunities offered by the EU single market(⁶⁷). This is the combined outcome of factors such as language barriers or family constraints, as well as skill mismatches (such as in the case of coal miners — ILO and OECD, 2012). In such cases, workers tend to remain in their countries despite poor prospects for high quality jobs.

The free movement of services may also have a positive impact on job quality. The free movement of services is one of the founding principles of the European Union (Article 56 TFEU). The principle of the freedom to provide services enables a business providing services in one Member State to offer services on a temporary basis in another Member State, without having to be established. The exercise of this right entails that companies, when providing services in another Member State, may need to post employees to work temporarily in an EU country other than the one where they are habitually employed. Posting workers(68) allows

- (⁶⁷) At the end of 2012, 14 million EU citizens resided in another Member State, i.e. 2.8% of the total EU population, up from 1.6% at the end of 2004, but lower than the share of non-EU nationals (4%) (European Commission, 2013b). On average the employment rate of mobile EU citizens was 67.7% in 2012; mobile EU citizens not in employment represent only a limited share (European Commission, 2013b).
- (⁶⁸) A 'posted worker' is a worker who, for a limited period, carries out their work in the territory of a Member State other than the Member State in which they normally work. Posted work relates to the free movement of services and is legally distinct from individual migration, which relates to the free movement of labour. For a summary on EU legislation on posting of workers, see, for instance, http://europa.eu/legislation_ summaries/employment_and_social_policy/ employment_rights_and_work_organisation/ c10508_en.htm

Box 4: Job quality of posted workers

The 1996 Posting of Workers Directive⁽¹⁾ requires Member States to ensure that posted workers are subject to the host country's laws, regulations or administrative provisions, and generally applicable collective agreements in the construction sector as regards a core of employment conditions, such as: applicable minimum wages, maximum work and minimum rest periods, minimum paid annual leave, health, safety and hygiene at work, employment terms for pregnant women and young people, rules prohibiting child labour, and equality of treatment between men and women. However, the Commission's close monitoring of the implementation of the 1996 Directive found that the rules laid down by the Directive were not always correctly applied in practice by Member States.

This led the Commission in 2012 to table a proposal for an Enforcement Directive, aimed at providing clearer rules on posted workers and practical safeguards. A new Enforcement Directive⁽²⁾ on the posting of workers based on the Commission's proposal was subsequently adopted by the co-legislator in May 2014. Some of the measures in the new Directive are: a clearer definition of posting to inhibit the growth of letter-box companies; a list of national control measures that the Member States may impose on posting companies; the possibility for Member States to require the designation of a contact person within the posting company to liaise with enforcement authorities; the option of applying an obligation to declare the identity of the company, the number of workers, their period of posting and the nature of services, and to keep basic documents such as employment contracts and payslips available at the workplace. Moreover, the new Directive includes the provision that penalties imposed on service providers in one Member State can be enforced and recovered in another. In addition, it also introduces a limited subcontracting liability in the construction sector. This means that posted workers in this sector can hold the contractor in a direct subcontractor relationship liable for any outstanding net remuneration corresponding to the minimum rates of pay, in addition to or in place of the employer.

- (1) Directive 96/71/EC of 16 December 1996, available at http://eur-lex.europa.eu/LexUriServ/ LexUriServ.do?uri=0J:L:1997:018:0001:0006:EN:PDF
- (2) Directive 2014/67/EU of 15 May 2014, available at http://eur-lex.europa.eu/legal-content/EN/ TXT/PDF/?uri=CELEX:32014L0067&rid=1

companies to exploit their competitive advantages across borders and to meet temporary shortfalls in labour supply (such as in construction and transport)(⁶⁹), while it may offer workers an opportunity to increase their job quality. At the same time, it may benefit European consumers by increasing competition in the single market(⁷⁰).

- Comparable estimates of the number (⁶⁹) of posted workers across sectors occupations and countries are not readily available but some studies provide ad-hoc estimates. For example, Idea and ECORYS (2011), European Commission (2011) and European Commission (2014) (June 2014 supplement to EU Employment and Social Situation, available at http:// ec.europa.eu/social/BlobServlet?docld=1194 5&langId=en) estimate that there are about 1 million posted in the EU, employed in the construction, transport, telecommunications, entertainment, repairs, maintenance and servicing sectors, but also in specialised, highskilled activities such as in the IT sector.
- (⁷⁰) Nevertheless, the impact of current regulations on competition is not unambiguous. For example, Mustilli and Pelkmans (2013) identify the imposition of, for example, a minimum wage for posted workers as a barrier to freedom of services in that it pre-empts Eastern European EU workers from exploiting their lower wages as a competitive advantage in the internal market.

Nevertheless, the job quality of posted workers is a policy concern. Posted workers may suffer abuses such as not being appropriately or fully paid. In addition, one study found that only a very small minority of foreign workers were unionised compared to native workers(71). In turn, abuses may affect the job quality of residential workers by placing downward pressure on their wages and working conditions, with a potential negative impact on their motivation and productivity. All in all, the empirical evidence on the impact of posted work on the job quality of resident workers is scant and not clear-cut, partly due to lack of adequate data.

To counteract such adverse outcomes, important EU measures have been put in place to ensure that posted workers are not deprived of the protection of basic employment rights in the host country and that enterprises face a level playing field of competition (Box 4). Nevertheless, the

^{(&}lt;sup>71</sup>) See, for example, Hansen and Andersen (2008) for the case of Eastern European workers working in the Danish construction sector.

monitoring of the employment conditions of posted workers — where relevant in cooperation with the social partners in the 'posting' as well as 'hosting' countries needs to be intensified, while maintaining a balance between the protection of workers' job quality and the cost of administrative requirements imposed on service providers operating across borders.

4.3. Demographic change calls for an innovative approach to job quality

This subsection looks at the workplace challenges faced by older workers, and briefly discusses how structural changes are expected to affect their job quality and how policies can address present and future challenges and improve job quality of older workers. The section then looks at young workers, who have seen their unemployment rate soar to historical levels in recent times, the challenges they face and the policies needed to improve young workers' employment and avoid human capital erosion.

An ageing population and changing family structures (including a rising number of one-parent families) are important future demographic developments that pose important challenges to EU job quality, with a direct impact on labour market participation, productivity and growth. Job quality (e.g. straining working conditions) and specific characteristics of tax and benefit systems (including pensions) have an important impact on older workers' decisions regarding labour market participation and retirement (e.g. European Commission, 2011a; Lindström, 2006). In addition, the crisis has shown that young workers' job quality can be especially vulnerable, potentially reducing future opportunities for employment in high-guality jobs and thus productivity and growth. In order to preserve the European social model, a set of policies is needed to help older people stay active longer, retire later and become more productive, while ensuring that young workers find and keep a suitable job and use and reinforce their human capital.

4.3.1. More flexible work arrangements and skillupdating for older workers while addressing age discrimination

About six out of ten EU citizens perceive that workplaces are not adapted to the needs of people aged 54 and over, although there are large differences across Member States: from about 80% in Hungary to below 40% in Sweden (e.g. Eurobarometer, 2012). Work psychosocial(⁷²) and physical strain is a strong push factor to early retirement for older workers (e.g. Bonsdorff et al., 2010; Park, 2010; Pollack, 2012). Such strains are often rooted in a loss of control over working conditions and a (perceived) lack of recognition of their performance (e.g. Siegrist et al., 2007; Oorschot and Jensen, 2009; Siegret and Wahrendorf, 2011).

Older workers have the lowest probability to transit to unemployment (if compared with the other age groups), but their probability to transit from unemployment to employment is also the lowest and their probability to transit to inactivity is the highest (e.g. RWI 2014). In addition to skewed financial incentives (e.g. expected pension income that exceeds contributions), poor career prospects may contribute to such an outcome. Poor career prospects for older workers often reflect a lack of recognition of their experience and expertise, which in turn discourages the search for a better or more adequate job. This calls for measures that strengthen the recognition of older workers' informally acquired qualifications, in combination with an enhancement of their job search intensity (in close collaboration with public employment services).

Furthermore, skills, especially ICT skills, are an important driver of job opportunities for older workers. For instance, Biagi et al. (2011) estimate that being skilled and using a PC at work reduced the probability of exiting employment by 12 percentage points in Italy in the early 2000s. This example illustrates that barriers to learning and training for older workers should be lowered.

Age discrimination in the workplace is still prevalent. For example, in 2011 one in five people surveyed experienced or witnessed age discrimination in the workplace or when looking for work (Special Eurobarometer, 2012). Strong differences exist between Member States, from almost 40% in Hungary compared to about 15% in Ireland. Nevertheless, employees aged 54 and over are thought to be more experienced and more reliable than younger employees (i.e., respectively 87% and 67% 'more likely'). Age discrimination and stereotyping may push older workers to early retirement (e.g. Gringart et al., 2011). They may be rooted in the perception that older workers are more reluctant to accept organisational change or new types of work (e.g. Taylor and Walker, 2003). Institutional reforms can address these forms of discrimination. Note though that 'age discrimination' laws that counteract these trends by reinforcing the employment protection of older workers may in fact reduce their hiring opportunities by increasing firing costs (e.g. Heywood and Siebert, 2009).

Measures to strengthen older workers' control over their working conditions could include the promotion of technologies that create more flexible and safer and healthier working conditions such as flexible working time and teleworking. The provision of eldercare facilities for partners may also ensure a better balance between family and working lives. Barriers to learning and training for older workers should be lowered. Special programmes (including training subsidies) focused on updating the skills of older workers, especially the low-skilled, may play an important role. Finally, ensuring an appropriate balance between efforts spent and earnings may improve their motivation, career prospects and recognition.

4.3.2. Investing in young workers' job quality

Occupational together with geographical mobility will be key to improving job quality in the future. In an ever-changing economy, workers will have to become receptive to more frequent job change if they want to improve their job quality. Young people have a stronger potential in this regard. They are on average more willing and able to move geographically since they often face fewer family commitments and are more likely to speak foreign languages and therefore adapt more quickly to new settings.

However, young people often lack the initial experience in professional life to kick-start their career along a path of high-quality jobs. Well-targeted labour market policies that invest in young people and improve the job quality of present and future cohorts of young workers are crucial. Such policies include modern apprenticeship systems, skill development that matches better the (short- and long-term) needs of the labour market, and guidance. From this perspective, it is imperative to ensure that all young people, whether registered with

^{(&}lt;sup>72</sup>) Such as working in a post that does not correspond with the level of qualification.

employment services or not, receive an offer of employment, an apprenticeship, a traineeship or the chance to continue their education or training within four months of becoming unemployed or leaving formal education — as has been stipulated in the Youth Guarantee(⁷³).

In support of this, there is a need to strengthen the capacity of the public employment services, reform education and training systems, and strengthen partnerships to reach out to inactive young people who are not registered with the employment services. In addition, the first experiences should offer quality learning content and satisfactory working conditions — as will be promoted by the European Alliance for Apprenticeships(74). Furthermore, the European social partners' Framework of Actions on Youth Employment is addressing several of the challenges related to bringing more youth into employment. Also, as the single market becomes more open and integrated, there will be an increasing need for a regular, low-cost and real-time flow of information on jobs across the EU, for young (as well as all) workers, as is envisaged under the European Jobs Network (EURES)(75).

Finally, persistent scarring effects affecting the current cohort of young workers will require attention to improve their job quality in coming years.

4.3.3. Tackling persistent gender discrimination

Women's job quality continues to be adversely affected by persistent forms of discrimination. In addition to lower participation and employment rates and shorter careers, Section 3 shows significant gender differences in earnings in the EU: on average women earn 16% less than men per hour of work. They also participate less in decision-making: women account for an average of 18% of the members of the board of directors in the largest publiclylisted companies (far from the 40% target for 2020) and 3% of CEOs (⁷⁶).

(76) See for instance http://ec.europa.eu/justice/ gender-equality/files/documents/140303_ factsheet_progress_en.pdf



As labour market participation, employment and retirement age are positively linked to education levels, an increasing share of women receiving a higher education is expected to result in better labour market outcomes. Nevertheless, along with appropriate legislation and social policies (e.g. European Commission, 2014c), addressing discrimination in general also calls for labour market policies that focus on the further strengthening of occupational and geographical mobility at the European level, through strengthening job search facilities, improving the portability of social security rights (such as pensions, medical care, unemployment benefits, etc.), and the recognition of skills and education certificates. Increased labour mobility decreases the bargaining power of employers and as a consequence also their scope to discriminate (77).

4.4. The jobs potential of the green economy

The greening of the economy can be a source of employment growth, as by increasing the efficiency of production processes, adopting innovative solutions to save resources and reduce pollution, developing new business models, or offering more sustainable products and services, companies can expand their markets and create new jobs, while transforming existing ones. In a knowledge economy, higher resource productivity can augment employment and allow wage increases without reducing the profit rate on the reduced capital stock (⁷⁸). It is estimated that reducing the total material requirement of the EU economy by 24% could boost GDP by up to 3.3%, while creating 2.8 million jobs (⁷⁹).

There has been considerable job creation in the environmental goods and services sector (EGSS) even during the economic crisis. Employment in the EU increased from 3 to 4.2 million between 2002 and 2011, including by 20% during the recession years (Eurostat). This trend is expected to continue as the EGSS sector supports the overall greening of the economy. Take the example of recycling. Many everyday goods are made out of materials that can be recycled. Recycling has introduced new production processes to treat used materials and to make new products out of old ones. This can generate new jobs of different levels of skills. ILO and OECD (2012) in turn provide a review of studies pointing to a significant job-creation potential in renewable energy sectors and associated with energy-efficient buildings.

Workers expect that further greening will have a positive impact on job quality, especially on their health (Gaušas et al., 2012) (Chart 21). Nevertheless, in a successful transition towards a greener economy, several downward risks for job quality in all its dimensions may have to be considered, as highlighted below.

⁽⁷³⁾ For more details, see Council Recommendation of 22 April 2013 on establishing a Youth Guarantee, available at http://eur-lex.europa.eu/legal-content/EN/ ALL/;ELX_SESSIONID=INQSTntdhQbGNI1z 7P6hZ0YHvy8dS2IKN2wkn8lffx29RxTFL mTLI-60128961?uri=CELEX:32013H0426(01)

⁽⁷⁴⁾ More details available at http://ec.europa. eu/education/policy/vocational-policy/ alliance_en.htm

^{(&}lt;sup>75</sup>) For more details, see https://ec.europa.eu/ eures/page/homepage?lang=en

^(**) In perfect competition and perfect information, wage differences reflect differences in productivity and job quality, and lower job quality should give rise to a positive wage premium and discrimination cannot persist (e.g. Becker, 1957; Rosen, 1986). However, once perfect labour mobility does not hold and employers have an inclination to discriminate or stigmatise, then lower wages may be paid to the victims of discrimination (e.g. Black, 1995). The stronger the barriers to all forms of job mobility, the more likely low job quality will be associated with low pay, as employers can use their bargaining position.

⁽⁷⁸⁾ http://www.unido.org/fileadmin/user_media_ upgrade/Media_center/2013/GREENBOOK.pdf

⁷⁹) http://ec.europa.eu/environment/enveco/studies_ modelling/pdf/report_macroeconomic.pdf

4.4.1. Skills and training needs in the green economy

The introduction of new products and processes associated with greening (e.g. improving resource efficiency, recycling waste or preserving biodiversity) are due to entail changes in skills requirements and occupational profiles. Traditional skills remain important but new tasks are required, with increased demand for a skilled workforce in growing ecoindustries, up-skilling of workers across all sectors, and re-skilling of workers in sectors vulnerable to restructuring. Workers may not be fully prepared for such tasks, or they may pose new safety and health risks (see below). For example, electricians are not trained to work at extreme heights and construction workers may not know how to deal with new material or electrical hazards.

Opportunities to move to green jobs of better quality will depend largely on workers' ability to upgrade their skills. This, in turn, will require enough flexibility in educational and training schemes to keep pace with green products and process innovations.

Education and training systems (vocational training, life-long learning programmes, on-the-job training) can be effective tools for coping with the demand for new skills and preventing skill bottlenecks. Targeted bridging programmes which put low-skilled workers on a sustainable long-term career path (e.g. pre-vocational training schemes providing basic skills to enter technical training) could temper emerging inequalities in job quality. E-learning throughout the career supported by instruments such as online libraries and interactive tools also constitute interesting options (e.g. Cedefop, 2010; EU-OSHA, 2013). Anticipating future skill needs and supporting the dissemination of new training opportunities, as outlined in the New Skills for New Jobs agenda and the European Quality Framework for anticipation of change and restructuring(⁸⁰), in close cooperation with public employment services (European Commission, 2010a and European Commission, 2014e), are another policy priority.

4.4.2. Anticipating change, securing transitions and considering new health and safety risks

As stated, the greening of the economy can bring along many occupational risks. Some new tasks (e.g. accessing the exterior peak of windmills (AEE, 2012)) or products (e.g. the use of microorganisms in the production of biofuels (Driscoll et al., 2005)), the use of nanomaterials) may involve risks in terms of workers' health and safety (e.g. falls or respiratory illnesses). These uncertain risks are often without monetary compensation: for example, intensive manual workers in waste management face strong health and safety risks and often receive low pay (e.g. Antonsson, 2014; EU-OSHA, 2013).

Monitoring the impact of new technologies on job quality may pose challenges, in particular to SMEs, which may not have the necessary resources to make adequate assessments of new processes and products vis-à-vis larger firms. These have better access to financial resources and technologies, better access to information, internal human resources and access to skills programmes.

These developments raise several important challenges (which go beyond labour market policies), including: filling the gaps in our knowledge to make more reliable risk assessments; promoting technologies that reduce health and safety hazards of intensive manual work such as in waste management and recycling (including collection, transport, and disposal and processing); promoting product designs that cover the whole life cycle of products, including their recycling at the end of their use; and integrating in the production process an independent assessment of the health and safety risks associated with the introduction of new green products or processes (e.g. EU-OHSA, 2013).

Awareness-raising activities informing workers of their employment rights and obligations and upgrading their skills to include the new "greener" forms of materials and production methods can substantially improve working conditions and decrease safety and health hazards. Promoting social dialogue at industry and sector levels will be key in this respect (e.g. European Commission, 2014e). Further strengthening international cooperation and health and safety more generally, via for instance, the Green Growth Knowledge Forum (⁸¹), can also provide innovative solutions.

4.4.3. Addressing gender stereotyping

Women are more often employed in occupations that are seen as less closely related to the greening of the economy (e.g. health and social work, education and retail), while men are more likely to be employed in research, engineering, manufacturing and construction of energy- and resource-saving technologies (82), requiring STEM skills. Such activities are also often characterised by a lack of managerial positions held by women. While the greening of the economy is likely to affect all sectors (for instance via the integration of environmental considerations in education and training, or adding skill sets in retail to advise customers on the environmental performance), there is a risk that it may be perceived as primarily creating more and better jobs for men.

4.5. Strengthening job quality to foster future productivity growth in the face of significant structural changes

Ongoing structural changes are expected to have a significant impact on job quality and workers' performance. They can bring along a host of opportunities for job creation and improving job quality. This may happen through: widening the opportunities to exploit countries' comparative advantages through new production processes, new products and new markets; mitigating physical or psychosocial barriers to labour market participation, notably of more vulnerable workers (e.g. older and disabled workers); and generating greater (occupational and geographical) labour mobility and thus a larger choice of jobs and the opportunity to perform tasks that best fit workers' abilities and preferences. This may reinforce overall productivity growth and earnings potential.

⁽⁸⁰⁾ See, for example, http://ec.europa.eu/social/ main.jsp?langld=en&catId=89&newsId=201 2&furtherNews=yes

^{(&}lt;sup>e1</sup>) See the Green Growth Knowledge Forum launched by the Green Growth Institute, the OECD, the United Nations Environment Programme and the World Bank, see http://www.greengrowthknowledge.org

^{(&}lt;sup>82</sup>) For example, Blanco and Rodrigues (2011) report that 78% of the workforce in wind energy is male.

However, technological change, globalisation, demographic ageing and the greening of the economy can have significant negative implications for job quality, including: rendering jobs obsolete (just below 50% according to some authors), skill erosion, stronger job insecurity, longer or uncertain (e.g. zero hour jobs) working hours, lower wages, new and unknown health and safety risks, and polarisation (i.e. non-equitable distribution of the gains in job quality with low to middle skills losing out and larger gender imbalances). These may in turn have adverse feedback on productivity and labour market participation.

Therefore, to realise the full potential of the ongoing structural changes, allow workers to benefit from the opportunities generated and correct any adverse challenges will require relevant labour market reforms. These should allow workers to transit to jobs of better quality in a flexible but secure way, increasing workers' receptivity to innovations and changes in work organisation. Given the polarisation effects of skill-biased technological progress in combination with globalisation, ageing and greening, well-targeted policies will need to ensure that costs and benefits are more equitably distributed.

These policies include (⁸³): implementing active labour market policies such as better profiling, job searching assistance and connection between employment services; improving access to life-long learning and on-the-job training; strengthening labour laws and social security provisions (including portability of benefits); eliminating gender and age stereotyping, discrimination and stigmatisation and reducing the informal economy; strengthening the capacity to anticipate and assess risks to job quality structural changes and strengthening health and safety at work legislation; promoting effective social dialogue at all levels and with non-EU partners and increasing employees' empowerment to identify improvements to job quality. These will strengthen labour allocation efficiency with a positive impact on productivity and labour market participation.

5. MODERNISING WORK ORGANISATION TO FOSTER PRODUCTIVITY GROWTH

This section looks at the distribution of different types of work organisations across sectors, occupations and Member States, and its evolution in recent years. It describes differences in job guality associated with different forms of work organisation in the EU. It then explores how work organisation(84) can be shaped to increase productivity and labour market participation under the continuous pressure of ongoing structural changes (technological progress, globalisation, demographic change and the greening of the economy). It looks at how stimulating creativity and fostering exchanges between workers can prevent stress and help maintain good physical and mental health, while at the same time improving productivity and innovation capacity. It sees how special arrangements can be implemented to accommodate older workers, workers with disabilities or certain diseases, and workers with family responsibilities.

The section then discusses future challenges with respect to workplace learning. It ends by examining how expanding global values chains will affect work organisation, focusing on risks related to the global restructuring of value chains, the virtual collaboration across time zones and the absence of multi-layered social dialogue.

5.1. Work organisations differ across sectors, occupations and Member States

Analysis of EWCS 2010 data shows that work organisation varies across economic sectors and occupational categories, more than by company size or its age and gender composition. Chart 22 shows large differences in work organisation across sectors in 2010 (85). The Learning form is more prevalent in the financial intermediation and public utility sectors and the Lean form is less common in the wholesale and retail, transport and communication and hospitality sectors. Chart 23 shows large differences in work organisation across occupations in 2010. Learning forms are especially characteristic of the work of professionals, technicians and senior managers, but also of 31% of craft workers, 20% of plant and machine operators and 18% of elementary occupations. The Lean form is more frequent for senior managers (41%) and skilled blue-collar workers (38%). A high proportion of blue-collar workers are employed in Tayloristic forms of work organisation. Service and sale workers, clerks and unskilled workers mainly work in Traditional or Simple work organisations.

⁽⁸³⁾ Apart from other structural measures that are not directly related to labour markets, such as fragmentation of credit markets and access of SMEs, strengthening of single market and trans-European networks.

⁽⁸⁴⁾ In this chapter, work organisation refers to processes and relationships, including worker-worker as well as workermanagement interactions and workplace learning.

⁽⁸⁵⁾ See section 2.2 for more details on different forms of work organisation.

Work organisation varies across Member States. Chart 24 shows that in the Netherlands, Denmark, Sweden and Malta, half or more of the workers in private companies with 10 or more employees are employed in Learning organisations. In contrast, Bulgaria, Romania, Greece, Ireland, the United Kingdom and the Czech Republic show a very low share of workers in this type of organisation. More than a third of workers work in Lean organisations in Finland, the United Kingdom, Ireland, Estonia, Romania, Malta and Austria. In



Source: Eurofound estimates based on EWCS 2010 data.



contrast, Lean organisations are least common in the Netherlands and Greece. Tayloristic organisations account for at least 30% of employees in Hungary and Greece, and less than 10% of workers in Denmark, Finland and Malta. Simple organisations are most common in Bulgaria, Greece and Cyprus, where they involve more than a quarter of workers, and are least common in Sweden, Estonia, Austria and Hungary.

5.2. The interaction between work organisation and job quality: the importance of Learning and Lean Organisations

This subsection describes the interaction between job quality and work organisation as two important drivers of productivity growth (see Table 2).

Learning and Lean organisations are associated with relatively high job security (86) compared to Tayloristic organisations, which show the lowest levels of job security both in terms of higher chances of losing the current job and in terms of higher anticipated difficulties in finding another similar job. Learning organisations are associated with a higher level of employability compared to Lean organisations, probably due to the fact that training is less firm-specific and more general. The number of employees in Learning and Lean organisations with good career prospects is almost two times larger than those in Tayloristic and Simple organisations.



(86) Longest seniority is also reported in Learning organisations.

Table 2: Job quality and work organisation: interactions							
	Discretionary Lean Learning production				Simple		
	Earnings	I am well paid for the work I do	50.6%	44.1%	31.9%	36.0%	
	Job/career security						
	Employment contract	Permanent	88.4%	85.2%	79.7%	77.9%	
1 Socio-oconomic	Employment contract	Fixed term or TAW	8.3%	11.0%	16.2%	15.1%	
security	Career prospects	Strong career prospects	37.6%	38.7%	19.9%	22.3%	
Security	- Job security	I might lose my job in the next 6 months	14.1%	19.8%	26.0%	21.4%	
		It's easy for me to find					
	- Transitions	an other job with a	33.0%	29.1%	24.4%	31.7%	
		similar salary					
2. Education	On the job training	On the job training	40.4%	48.6%	31.4%	23.9%	
	Health and safety						
	Destance exclusion designed	Repetitive hand or arm movements	44.8%	62.3%	74.3%	55.3%	
	Posture related risks	Tiring or painful positions	23.6%	39.7%	48.8%	28.9%	
		Noise	16.9%	34.5%	43.0%	16.7%	
	 Ambient risks 	High temperature	9.0%	23.4%	26.6%	11.1%	
	Chemical risks	Breathing in smokes, dust	16.9%	33.1%	33.2%	14.6%	
	Stress	Direct reporting	25.5%	33.9%	30.4%	22.0%	
	Work intensity	High speed work all or	20.7%	36.4%	45.0%	21.3%	
3. Working			265%	45.6%	397%	21.2%	
conditions		A say in choice of	20.5 /0	0, 0.64	55.270	21.2 /0	
		working partners	23.8%	25.5%	8.6%	8.2%	
	Work authonomy	Able to apply your own					
		ideas at work	57.8%	45.9%	16.1%	24.4%	
		You are involved in					
		improving the work					
		organization or work	48.9%	44.5%	19.0%	16.8%	
	Employee concultation	processes of your					
	Employee consultation	department / organisation					
		You can influence deci-					
		sions that are important	40.6%	33.0%	10.9%	11.3%	
		for your work					
	Work-life balance						
	Asocial working hours	Night work	5.9%	11.3%	18.6%	12.0%	
	-	Shift work	13.7%	28.0%	40.4%	23.0%	
		Not fixed starting and finishing times	33.5%	30.5%	23.5%	25.0%	
4. Work-life balance	Elexible work hours	Easy to take time off to					
	LEXIDLE WOLK HOULS	during working hours to	72.8%	635%	489%	525%	
		take care of personal or	. 2.0 /0	22.2 /0		52.5 /0	
		family matters					
	Discrimination	Nationality	0.8%	2.2%	2.8%	0.9%	
		Gender	1.0%	1.2%	2.5%	1.4%	

Source: Eurofound estimates based on EWCS 2010 data.

The long-term investment in employees of Learning organisations is supported by compensation systems based on the overall performance of the company (26% of workers versus 22% in Lean and 12% and 10% in Tayloristic and Simple forms) and profit-sharing schemes (6.4% of workers in Learning organisations). Payments for bad or dangerous working conditions are highest in Lean organisations (around 13% of workers). Piece rate and productivity payments are most frequent for employees in Lean and Tayloristic organisations (around 19% of workers).

Learning and Lean organisations both report relatively high levels

of training (49% in Lean and 40% in Learning organisations) (⁸⁷). Nevertheless, employees in Learning and Lean organisations also report more frequently that the skills demands put on them are too high.

(⁸⁷) They also report most that the training has helped them to improve the way they work.

162

gif

Employees in Tayloristic but also in Lean organisations report the highest exposure to physical risk factors (environmental, posturerelated risks, chemical risks, ambient risks, dangerous substances). Employees in Tayloristic organisations also report the highest levels of exposure to psychosocial risks factors (violence, fear, discrimination, stress, emotional demands, poor leadership (88)). Interestingly, about 90% of employees report being 'very well' or 'well' informed about the health and safety risks associated with their work with only small differences between organisations.

Work intensity is highest in Tayloristic and Lean organisations and lowest in Learning organisations. Workers in Learning and Lean organisations report the highest level of autonomy in terms of choosing partners and in terms of applying their own ideas. Learning organisations are more likely to offer more sustainable jobs in that workers are able and willing to keep and successfully manage their jobs until the age of 60.

There are few differences in exposure to long working hours across organisations. Workers in Learning and Lean organisations most often report having to work in their free time (around 11%), but they also report the highest level of employee-led short-term working time flexibility (56% of workers in Learning and 40% in Lean organisations). **Workers in Learning organisations report the highest level of worklife balance and satisfaction with working conditions** (85% and 90% respectively).

The data show a decrease in the number of workers undergoing employerpaid training in Learning organisations, and a slight increase in employer-paid training among Lean, Tayloristic and Simple organisations compared to 2000 (Annex 4, Table A4.9). Nevertheless, in 2010 workers in both Learning and Lean organisations were making greater use of self-paid training than in 2000. **Workers in Simple and Tayloristic work organisations were less likely to have any form of training in 2010 than in 2000**.

Table 3: Organisational forms across EWCS waves (2000-10)

	E	Total			
	2000	2005	2010	Total	
Learning	39.1%a	40.1%a	36.8%b	38.6%	
Lean	25.7%a	27.2 %b	28.6%c	27.2%	
Tayloristic	18.6%a	18.8%a	18.3%a	18.5%	
Simple	16.6%a	13.9%b	16.3%a	15.8%	

Source: Eurofound estimates based on EWCS 2010 dataset.

Note: Subscripts denote whether difference between values across columns (from different EWCS waves) are statistically significant at the 0.05 level. In particular, values having the same subscripts do not differ significantly while values with different subscripts are significantly different.

5.3. Declining Learning organisations and the move towards Leaner forms

Table 3 shows that the proportion of employees involved in Learning organisations has been decreasing between 2005 and 2010 (down from 40.1% in 2005 to 36.8% in 2010). At the same time, and probably as a consequence of the decline of the number of Learning organisations, the proportion of employees in Lean production forms of work organisation has been increasing significantly first between 2000 and 2005, and then also between 2005 and 2010. Tayloristic organisations remain stable over time: 1 in 5 organisations in Europe are structured in Tayloristic forms of work organisation. The proportion of Simple organisations has been decreasing between 2000 and 2005, then increasing back to 2000 levels. Such trend developments carry downward risks in terms of job quality and human capital resilience — as discussed in the previous subsection.

5.3.1. Different trends across Member States between 2000 and 2010

Perhaps surprisingly, the overall proportion of workers in Learning organisations appears to have decreased significantly between 2005 and 2010 (from 40.1% to 36.8%), replaced by an increasing number of Lean organisations, while the number of Tayloristic organisations was stable (1 in 5 EU organisations). However, this general trend does not apply to each individual Member State (Annex 4, Tables A4.1–A4.4). Countries can be grouped in four groups according to the developments in work organisation observed between 2000 and 2010.

In the first group (Annex 4, Table A4.1), Learning organisations increased either constantly between 2000 and 2010 or since 2005. In Latvia, Portugal and Malta this type of organisation increased between 12% and almost 20% over the 10-year period. Somewhat smaller increases are seen in Romania, Lithuania and Poland. In the Netherlands, Denmark, Cyprus and Estonia an initial decrease in the number of Learning organisations between 2000 and 2005 was followed by an increase in the following five years, bringing most of these countries back to the levels of 2000. In the case of the Netherlands and Denmark, these are among the highest in Europe: 60% of employees in private companies with 10 and more employees work in Learning organisations.

gif

In the second group (Annex 4, Table A4.2), Learning organisations are decreasing while Lean organisations are increasing, and these two trends are likely to be related. This development is most prominent in Germany, Luxembourg, Belgium and Austria. In Germany, for example, the difference in the proportion of Learning and Lean organisations was 31.5 percentage points in 2000, dropping to 15.8 percentage points in 2010. A somewhat smaller drop in the proportion of workers employed in Learning organisations occurred in Slovenia, Italy and Finland. A more complex trend is present in Sweden and Ireland: they saw a steep increase in Learning organisations from 2000 to 2005, then followed by a decrease in the later five years. Yet, Sweden is still the EU country with the highest proportion of employees working in Learning organisations - two out of three private organisations with more than 10 employees.

In the third group (Annex 4, Table A4.3), the general decrease in the number of

^{(&}lt;sup>88</sup>) Supportive leadership is most frequently reported by employees in Learning and Lean organisations, contrasting lightly the rather negative picture of exposure to physical and psychosocial risks in both Tayloristic organisations.

Learning organisations is mostly coupled with an increase in Tayloristic organisations. In France and the Czech Republic Learning organisations decreased, replaced by an increasing proportion of Tayloristic and Simple organisations. On the other hand, Learning and Simple forms of work organisation are replaced by Lean and Tayloristic in Hungary and Bulgaria. In Greece, Simple types of organisations are replaced by Tayloristic.

Finally, in Slovakia, Spain and the United Kingdom there were no substantial changes in the proportion of the four types of organisations between 2000 and 2010 (Annex 4, Table A4.4).

5.3.2. Different trends across economic sectors between 2000 and 2010

Trends are also different across sectors (Annex 4, Table A4.5). In the public utilities, financial intermediation, transport and communication and hospitality sectors there was a marked increase in the number of Learning organisations and a decrease of Lean organisations between 2000 and 2005, followed by the exact opposite trend over the next five years. For example, from 2000 to 2005, the share of Learning organisations in public utilities increased by more than 5 pps and the share of Lean organisations decreased by almost 10 pps. In the following five years, the share of Learning organisations decreased by over 7 pps, compensated by a 5 pps increase in the proportion of Lean organisations. The retail industry changed work organisation from Learning to mostly Lean and Tayloristic organisations, with the share of the latter increasing by almost 5 pps. In the construction sector there was a shift towards Lean and Tayloristic work organisations, especially in the first five years, though this trend appears to have halted now. In mining and manufacturing, a slight shift towards Lean organisations forms can be seen.

5.3.3. Different trends across occupations between 2000 and 2010

Trend developments in work organisation across occupations are also different (Annex 4, Table A4.6). Among high-skilled clerical workers such as legislators, managers, senior officials and professionals, the relatively high share of Learning organisations decreased substantially between 2000 and 2005, with some reverse trend in the case of professionals between 2005 and 2010. In contrast, Lean organisations have become more prominent. From 2000 to 2005, an increasing share of clerks and service and sale workers worked in Learning organisations, but this trend was reversed in 2005 and the share of Lean and Tayloristic work organisations increased. This is not the case for technicians and associate professionals, for whom nothing much changed over the period, except perhaps for some decrease in the proportion of Tayloristic organisations. By 2010, a higher share of low-skilled manual workers, those working in plants, assemblers, machine operators and those in elementary occupations were working in Simple organisations than in 2000. High-skilled manual workers, such as craft and related trade workers are primarily in Lean organisations (38% in 2010). The share has consistently increased since 2000.

5.3.4. A decrease in Learning organisations in smaller establishments

The biggest decrease (3 pps) in the share of workers in of Learning organisations between 2000 and 2010 occurred in smaller establishments, which switched to Lean organisations and to a certain extent also to Tayloristic organisations (Annex 4, Table A4.7). The strongest increase in the share of Lean organisations (6 pps) occurs in the biggest companies, though in this case the increase in the number of Lean organisations is due to a shift from Simple (down by 5 pps) rather than from Learning organisations.

5.3.5. Trends across different levels of seniority

In 2010, new workers (one year or less in a company) were less likely to find employment in Learning organisations and more likely to find employment in Lean organisations, and to a smaller extent Tayloristic organisations, compared to 2005 and 2000 (Annex 4, Table A4.8), although Learning organisations still represent the highest share. In contrast, the shares across the four different types of organisations did not change over the period for workers with two or more years of experience.

5.4. Complementing technological innovation with workplace innovation

Section 4 indicated how the ongoing technology change can create new opportunities for jobs and growth. The interaction between knowledge, innovation and education is seen to be a key driver of productivity growth in a knowledge-based economy(⁸⁹). However, for the knowledge-based potential to materialise, the knowledge triangle has to be complemented by forms of work organisation that use workers' human capital to their fullest (e.g. Totterdill, 2014).

Section 4 also indicated how structural changes can pose challenges since the nature of knowledge work differs markedly from routine work. In modern knowledge-based tasks, existing working arrangements that were functional in the manufacturing industry or clerical organisation such as vertical decision structures, Tayloristic division of tasks, repetition of work items, low level of autonomy, strict time management and high levels of intrusive control, may no longer result in higher productivity. Success in modern knowledge-based tasks is likely to rely more on the possibility of choosing to do what one is best at, a lack of interruptions and strong personal motivation.

Future developments in ICT and KETs are likely to affect work organisation internally (e.g. generating virtual workerworker interactions) and externally (e.g. greater outsourcing of tasks), while the production of new KETs-based products and services may pose new occupational hazards (e.g. through the use of microorganisms), all with a potential impact on productivity and labour market participation (e.g. EU-OHSA, 2013).

Finally, the impact of changes in work organisation on earnings distribution will also be affected by firms' human resource policies. Indeed, if firms encourage training they could promote their workers at the bottom end of the occupational or skill structure up to higher levels, rather than hiring new already-trained workers (e.g. Aghion et al., 1999).

^{(&}lt;sup>99</sup>) I.e. the so-called 'knowledge triangle' (see also http://ec.europa.eu/ education/policy/higher-education/ knowledge-innovation-triangle_en.htm).

Box 5: Work organisation and earnings distribution

Work organisation, including worker-worker and worker-employer interactions as well as workplace learning, is also an important driver of enterprises' productivity and distribution of factor income (e.g. Aghion et al., 1999). As further technological progress strengthens communication and information flows, a less hierarchical (more organic) structure of work organisation will likely emerge. This may give rise to fewer specialised tasks supervised by middle management, and to more multitasking within teams. Such team work may then give rise to knowledge and learning externalities that give an extra boost to productivity from which all team players may benefit via higher earnings (if profits are not extracted by 'team leaders').

However, as transaction costs decrease facilitating outsourcing, a stronger skillsegregation between enterprises may emerge (e.g., low-skilled employment in McDonald's Inc. versus high-skilled employment in Google Inc.), leading to stronger earnings parity among workers within enterprises but to stronger earnings dispersion between workers of different enterprises.

5.4.1. More autonomy and responsibility for workers may strengthen the EU's innovation capacity but also increase polarisation

Technology change will provide opportunities to strengthen firms' innovation capacity. Future developments in ICT (e.g. an expansion of cloud computing) will increase the potential for virtual workplaces with workers who are physically located in different places (including their own home) interacting in real time. Such developments may give workers more autonomy and responsibility and allow for better reconciliation of work and family life. As such, these changes in work organisation may strengthen the opportunities to make full use of existing knowledge, with the potential to generate new knowledge with new products and processes, or new applications of existing knowledge. Such developments may also strengthen labour market participation, notably of older workers and workers with disabilities or family responsibilities, and may become the primary driver of productivity growth for the resource-poor, skills-rich EU.

Appelbaum et al. (2011) estimate that a positive workplace environment and practices that develop employees' knowledge and ability to create value may increase productivity by 15% to 30% (taking account of specific characteristics of industries and occupations). Therefore, it will be important actively to engage employees in identifying and developing solutions, while allowing them to participate in the implementation of work innovations so that they become more receptive to change (e.g. Totterdill, 2014; Dhondt and Totterdill, 2014).

In this context, an important policy would be to facilitate the creation of EU-wide platforms that allow employees and employers to exchange experiences in developing and implementing solutions related to production and work organisation. The specific characteristics of such platforms will vary between production entities and may take place at European or national level. They can promote the exchange of experiences, help identify best practices, monitor their implementation, assess their impact on productivity and identify social implications.

Continuous change in work organisation may discourage individuals from staying in employment, especially older workers and workers with disabilities notably if low-skilled, and may adversely affect the commitment of the other workers. Moreover, greater flexibility may lead to further polarisation in the labour market with core workers remaining/being employed under attractive contractual arrangements (albeit with increased work intensity (90), and with other workers (such as temporary contract workers, hired self-employed or other forms of flexible contracts) acting as a buffer to accommodate the increased flexibility.

As stated, innovation may render tasks obsolete and skills obsolescence may accelerate to the extent that access to learning opportunities is not equitably distributed, thus reinforcing ongoing polarisation. Technology may also make task outsourcing easier, reducing job security, especially for low-skilled workers. In addition, virtual workplaces are expected to lead to more fragmented task organisation, which may have an adverse impact

on the team spirit of the workforce. The impact of this on productivity is unclear.

While future technological developments will create important opportunities to improve the EU's innovation capacity, realising and benefiting from this potential calls for workplace innovations that depend on the consensual effort of employees and employers. In this context, future workplace change should foster workers' engagement, promote social dialogue helping to align employers' and employees' objectives and motivation, address new challenges in office and workflow design such as information overload and distractions, and give workers more responsibilities and autonomy. As knowledge and autonomy become more important, more attention should be paid to the challenges faced by Learning organisation (as described in Section 5.3).

5.5. Fostering workers' engagement

It is often said that 'an organisation's greatest asset is its people'. But this is only likely to be true if and when they are committed to their job. Studies on the current shape of modern workplaces, such as Gallup's 'Q12' survey(⁹¹) (which, it should be noted, covers only United States workers), suggest that as little as one third of workers show high engagement at work and a further third of workers are 'not engaged', while another third are 'actively disengaged'.

Gallup's employee engagement index is based on worker responses to 12 policy-relevant workplace elements with proven linkages to performance outcomes, including: productivity; customer service; quality; retention; safety; and profit (Gallup 2013). Workplaces where workers score low in that survey suffer from lower productivity, are

what you do best every day? 4

- Does your supervisor, or someone at work, seem to care about you as a person? Is there someone at work who encourages your development? 6.
- At work, do your opinions seem to count? Does the purpose of your organisation 8
- make you feel your job is important? Are your colleagues committed to doing quality work?
- 10. Do you have a best friend at work?
- 11. In the last six months, has someone at work talked to you about your progress
- 12. In the last year, have you had opportunities at work to learn and grow?

⁽⁹⁰⁾ Which is not necessarily related to a decrease of job quality, as discussed in section 3.

⁽⁹¹⁾ The questions are: Do you know what is expected of you at work? 1.

^{2.} Do you have the materials and equipment

vou need to do vour work right? 3. At work, do you have the opportunity to do

In the last seven days, have you received recognition or praise for doing good work? 5.

less likely to create new jobs, are more likely to be reducing their workforce and are more likely to see employees leave.

The 'not engaged' are passive and less productive; they are, in Gallup's words, 'sleepwalking through their workday, putting time but not energy or passion in their work'. Actively disengaged employees 'are not just unhappy at work; they are busy acting out their unhappiness. Every day, actively disengaged workers undermine what their engaged co-workers accomplish' (Gallup 2013) and are a liability to the company. They spread frustration and demotivate colleagues. Gallup argues that management typically responds to low engagement with extrinsic motivation, e.g. by offering fringe benefits. However, these cannot address the fundamental needs of workers such as: sense of purpose; perceived relevance of their work; opportunity to use one's skills and learn new skills.

Similarly, engagement tends to diminish with educational attainment, with a 6 percentage point difference found between those with less than a high school diploma (34% feel engaged at work) and college graduates (28% engaged). This may reflect graduates having higher expectations that are harder to meet following their investment in education.

Workers' engagement seems to be sensitive to the organisation's size, with Gallup's research suggesting that there is something unique and beneficial about working in a small, tightlyknit environment.

Their research suggests that workers of all generations are most engaged when they have the opportunity to do what they do best every day. While those born between 1981 and 2000 are particularly prone to job-hopping compared to previous generations, this characteristic is clearly dependent on engagement levels. Nearly half of those who actively disengage want to change jobs, while only 17% of engaged ones do.

Findings by experimental psychologists (Pink, 2010) offer surprising insights into the mechanisms of motivation. Laboratory experiments highlight the limitations of external rewards (such as gifts or money) as incentives for creative problem-solving(92). However, studies also show that such extrinsic motivators work when it comes to routine tasks. In other words it appears that it is the worker's intrinsic motivation, curiosity and emotional engagement that drive performance when it comes to solving problems and carrying out non-routine tasks. The consequences for future work organisation are potentially very significant. If success in the future economy relies on innovation and solving complex problems, then employers will need to foster genuine personal interest in the work of their employees. Annex 5 provides examples that illustrate the positive link between the mental state of knowledge workers and productivity.

5.6. Management strategies for organisational efficiency: supervision and control versus common values

In the traditional bureaucratic industrial model, management has typically focused on designing and supervising work processes to minimise the (intellectual) effort and skill necessary for workers to carry out their work. Taylorism summarises this managerial ethos as the focus on constructing work procedures constrained to the point where workers can only do the correct thing in an economic way (McIntyre, 1984; Jackall, 1988). It features vertical division of labour, hierarchy, and formalised and standardised work processes (Mintzberg, 1983; Wright, 1996). Traditional management theorises that work can be divided between those who work and those who: plan; organise; coordinate; and control work.

However, management methods have evolved since changing patterns of work organisation require other forms of managerial intervention. Many modern professional organisations operate in conditions where it would anyway be difficult or even counterproductive to organise and control behaviour. Management in modern organisations turn to targeting behaviour indirectly, through norms and values (e.g. Etzioni, 1964). This is accomplished through managerial practices such as normative control: 'the attempt to elicit and direct the required efforts of members by controlling the underlying experience, thoughts, and feelings that guide their actions' (Kunda, 1992). Employees then accept and adopt as their own a corporate culture: the norms of behaviour preferred by the corporate organisation.

5.7. Office and workflow design for optimum efficiency

Efficiency in a typical modern office is prone to the risk of distraction, information overload and lack of control over one's personal space. Companies may overlook these risks as they strive to encourage team work through faster work pace via heavy IT use, multitasking and office design, as well as greater control over employees.

5.7.1. The strain of multitasking in intellectual work

Many contemporary employers require staff to engage in multitasking. Yet studies have demonstrated that this may be counterproductive. Clifford Nass, who carried out seminal studies on how people interact with communication technology, concluded that modern life is overloaded with information and that this is not conducive to remaining focused and analytical thinking (Ophir et al., 2009). His studies have shown that people who frequently engaged in multitasking actually score worse in performing parallel tasks.

Since the brain has very specialised modules for different tasks, like language processing and spatial recognition, it stands to reason that it is much harder to perform two similar tasks simultaneously. Driving and talking do not use the same bits of brain but answering an e-mail while talking on the phone does — creating information bottlenecks. Studies by Gloria Mark, professor of informatics at the University of California, have found that when people are continually distracted from one task. they work faster but produce (Mark et al. 2008) less. New computer and media 'advances' can thus be seen as placing new demands on cognitive processing and particularly on attention allocation. Experiments have demonstrated

^{(&}lt;sup>92</sup>) A classic psychological experiment from 1969 by Edward Deci (echoing pioneering experiments on rhesus macaques by Harry F. Harrow from 1949) showed that extrinsic motivators (gifts) are counter-productive in puzzle-solving tasks. The gifts distract the subject from the task and undermine the intrinsic motivation and the pleasure of performing the task itself.

that students solving a maths puzzle took 40% longer — and suffered more stress — when they were made to multitask (Ophir et al., 2009).

Such a working environment saturated with media and the resulting information and task overload is a recent phenomenon. Multitasking is still perceived as a target that is often actively encouraged in the corporate environment, but there is a case for serious analysis into how progressive employers could shape communication policies in order to minimise the psychological burden and productivity losses stemming from multitasking.

5.7.2. The pace of work and efficient time use in knowledge occupations

Ergonomics of work reflect the cyclical balance of intense effort and focus with recovery and rest. Sensible time management should take a long time perspective. Like a long-distance runner, cognitive workers tend to pace themselves in order to achieve optimum results. Productivity should then be assessed not over a day or week but over years or even a worker's entire productive life. What may appear to be high productivity, from the employer's point of view, can mask hidden costs. If a worker achieves high output in the short term but, as a result, suffers burnout or illness and exits early from the labour market, many of the costs are ultimately borne by society at large through the welfare system.

Knowledge work that requires intense mental focus has been found (Hobson and Pace-Schott., 2002) to follow a particular cycle of 90 minutes with corresponding performance benefiting from short breaks. In cognitive activities such as assembly-line production, the break or rest does not follow the same pattern. Adding variety, changing the subject of work, off-time, freedom from meetings and rapid reaction to external demands, being given time to reflect and think are all factors that can help achieve a balanced working day. The need for the body and mind to recover is clearly recognised in legislation covering occupations such as pilots and truck drivers, since the consequences of human error due to overwork in such areas are obvious. A Directive also sets for all EU workers minimum standards in terms of rest periods and limits to working time (93). Numerous studies have linked excessive working hours with health risks, including mental illnesses. Common mental disorders, such as depression, are an important public health concern (Mathers, 2006; Eaton, 2008). According to projections by the World Health Organisation, depressive disorders will be the leading cause of disease burden in high-income countries by 2030 (Mathers, (2006). In addition to human misery, mental disorders often result in substantial work impairment and lost work days (Eaton, 2008; Adler, 2006; Wang, 2004; Demyttenaere et al., 2004).

As mentioned above, the unrestrained and ever-increasing use of information technology can be seen as a mixed blessing. Solutions to avoid productivity-killing interruptions could include reducing the number of alerts to a manageable level and creating periods of a digital down-time, devoted to deep thinking and concentration.

Finally, productivity assessment needs to be seen in relation to the type of job. Jackson (2012) argues that seeking higher productivity in a conventional way may be counterproductive in occupations that rely on allocating one's time to the service recipient. For example, chasing productivity growth in caring professions, social work, medicine and education according to the manufacturing paradigm leads to degradation of the service provided.

Finally, looking at successful companies at the forefront of workplace innovation suggests that taking a holistic approach to office design can be an important driver of productivity growth, as in the example, albeit somewhat exceptional, described in Box 7 in the annex.

5.8. Addressing future challenges in the Learning organisation

Section 5.3 suggests that Lean organisations are increasing mostly at the expense of Learning organisations, and that these two forms of work organisation are becoming increasingly divergent. The shift to Lean forms of work organisation may risk eroding the performance and job quality of European workers. Indeed, such a move is happening when technology and globalisation emphasise the importance of knowledge and the speed at which knowledge and skills may become obsolete. Learning rather than Lean organisations appear better placed to exploit the opportunities brought about by structural changes.

As a consequence, there is a need for firms to engage in organisational learning and for workers to engage in acquiring new competencies to strengthen the EU's comparative advantages in world markets. In this context, policies should develop the framework conditions to increase the number of Learning organisations and support the change process. Box 6 provides some considerations on how to revert the shift from Learning to Lean forms of work organisation.

A coherent and holistic policy approach, integrating policy objectives from various policy domains such as employment, social policy and enterprises' competitiveness policies may be necessary. In addition, these processes would have to be implemented across different levels — EU, national, local, individual companies — and will involve a number of actors - various governmental bodies, social partners, management, workers of private and public companies acting in Europe. The number of actors and fields of actions will require an organised effort to create a comprehensive and consistent framework of policy recommendations and initiatives at the EU and national levels. These would then be used for guiding and supporting the process of change of local workplaces by ensuring coherence of actions between different actors and different levels to find the optimal form of work organisation for each (locally specific) circumstances.

^{(&}lt;sup>93</sup>) Directive 2003/88/EC of the European Parliament and of the Council of 4 November 2003 concerning certain aspects of the organisation of working time.

Box 6: Promoting Learning forms of work organisation

Labour market policies aimed at reducing, halting and reversing the decline in Learning organisations should:

- Promote mutual learning and exchange of good practices in the design of programmes: e.g. Denmark, the Netherlands and Sweden have been developing national initiatives and research programmes to support innovations in organisations.
- Provide staff training and development, with emphasis on learning a broader skill set that enables workers to engage with a wider range of problems, to be more able to respond to unforeseen events and to support processes of workplace innovation.
- Involve social partners (when social partners are involved in work organisation) in the initiation and streamlining of the process of organisational change, thereby adding to its legitimacy and increasing acceptance.
- Provide innovative policy instruments that help to initiate, streamline and guide the process of change and the introduction
 of new, more innovative work practices. Aside from various forms of direct or indirect financial support, this could include
 consultancy helpdesks or information databases with (locally relevant) good and bad examples. These would be especially
 relevant to SMEs that may lack the resources for such activities compared to bigger companies.
- Emphasise the synergies between workers' well-being and companies' performance, which may increase workers' involvement and intrinsic motivation, improving their learning and problem-solving abilities and benefiting their physical and psychosocial state.
- Assist individual workers in developing their abilities throughout their working life, via, inter alia, the provision of necessary information and facilities, certain types of training or (subsidised) access to various forms of education and life-long learning, and encouraging workers to take a more active approach to the development of their skills and abilities.

5.9. Further globalisation brings changes to work organisation with job quality implications

5.9.1. Global restructuring of value chains

Globalisation and the expansion of global value chains is expected to have a deep impact on work organisation, giving rise to a stronger division of tasks (including conception, design, production, advertising and marketing) spread across the world (Newhouse, 2007; Dedrick et al., 2008). For workers, this means increasing the need for specialisation in specific tasks at the local level and the acquisition of skills (e.g. foreign languages and ICT skills) related to global collaboration. As global value chains expand, workers have the opportunity to specialise in those activities in which they have a comparative advantage while gaining more international experience and interacting in multicultural environments. Further specialisation and participation in networks may lead to increased overall productivity which in turn may increase job quality, including earnings and learning ability (e.g. Grossman and Rossi-Hansberg, 2006).

As global value chains expand and European enterprises want to remain at the cutting-edge of innovation, employees and their representatives may get more involved in participative and empowering forms of work to use their knowledge and experience to the fullest extent. Nevertheless, further opening to international markets creates stronger opportunities to off-shore activities and may increase pressures to deregulate, which can weaken the bargaining power of employees (as employers can use, for example, the threat of offshoring) (⁹⁴).

5.9.2. The risk of further polarisation

Such changes in work organisation associated with the expansion of global value chains will also pose risks to workers, adding to polarisation and inequality among workers just as seen with technology. The restructuring of global value chains may place stronger emphasis on unit labour costs competition. This may lead to either lower wages or job losses due to firm relocation to exploit differences in unit labour costs, notably in areas with fewer job alternatives. While this may be (partly) off-set by taking up new activities, the risk exists that the patterns of specialisation built up in the past will no longer meet the requirements of the new tasks. This may be of especial concern in the case of older workers and workers with limited learning capabilities.

Furthermore, in anticipation of a further restructuring of the global value chain,

local employers may be inclined to hire temporary contract workers to act as a buffer against unexpected developments further down or up the chain (e.g. Lehndorff and Voss-Dahm, 2005). Consequently, while workers in core activities may gain favourable working conditions, workers in non-core activities may see their job insecurity increase. This may in turn affect adversely the motivation and effort of workers who are most affected and perpetuate their unfavourable position.

In other words, future developments in global value chains may imply job losses or lower job quality (lower wages, job insecurity), affecting primarily the 'weakest' workers including the lowskilled or those on temporary contracts (e.g. OECD, 2006).

In addition, the resilience of a global chain is largely determined by the resilience of all of its components. In that sense, job security may be adversely affected by events beyond the control of local management and employees, such as geopolitical tensions or natural disasters.

5.9.3. Working across time zones

Expanding global value chains will also intensify real-time collaboration across different time zones (e.g. Stanoevska-Slabeva, 2009). Alongside the gains in productivity and earnings mentioned,

^{(&}lt;sup>94</sup>) See, for instance, ILO at http://www.ilo.org/ global/research/topics/labour-standardsand-socially-inclusive-globalisation/lang--en/ index.htm

such international workplaces pose work organisational challenges. Local working time will have to be aligned with working time in other time zones (i.e. 24-hour reachability), resulting in more flexible and longer working times to participate in digital teamwork spread across different time zones. They can also increase stress from differences in work cultures, mediated communication and language barriers and the fragmentation of work organisation, which may generate faltering team dynamics and erode trust between workers, potentially reducing workers' motivation and effort. Nevertheless, at the same time, they can also reduce longer work hours or shift work, as workers in another time zone can take over the task

5.10. Conclusion: stronger employee empowerment matters for productivity growth

The restructuring of global supply chains combined with technology may benefit the resource-poor, skills-rich European Union as its skills structure may have a comparative advantage in world markets. Ongoing structural changes may bring changes to work organisation that can improve job opportunities, through greater mobility and skill matching. These can in turn improve job quality (e.g. greater autonomy, responsibility and flexibility in the workplace, more flexible working arrangements, which may entice/maintain older workers, workers with disabilities and those with family responsibilities in the workplace; higher earnings).

However, changes in organisation due to technology and globalisation can render skills, tasks and jobs obsolete at a high speed (through automation and relocation) and reduce job quality (more flexitime and longer working hours to fit diverse time zones). They will also require specific skills to act in international environments (e.g. languages and ICT). In addition, the gains and losses may be unequally distributed between employees and employers (as it changes the bargaining position) and between different groups of workers (low- versus high-skilled workers) resulting in further polarisation and inequity.

Unless such challenges are addressed, changes in work organisation due to technology and globalisation may carry a severe and persistent socioeconomic cost for individual workers and for society as a whole (e.g. lower production capacity, dependence on social assistance). Such adverse outcomes can be counteracted by adequate labour market policies and improvements in work organisation that benefit both employees and employers and facilitate labour reallocation in a flexible but secure way.

Active labour market policies, life-long learning (including investing in the skills relevant to knowledge occupations and new tasks more generally) and modern labour laws, complemented by an increased forecasting capacity to anticipate, 'locate' the challenges and adapt to change are important. Improving the link between education and the needs of enterprises that operate in different time zones, through linguistic education and enhanced cultural awareness, may prove useful. The links between labour market policies and other policies will need to be strengthened, including in areas such as the trans-European and international networks for communication and collaboration, and international cooperation on security (including internet transactions).

Given the ambiguous impact of expanding global value chains on industrial relations, promoting productivity and inclusive growth may require the promotion of a global social dialogue and through it the negotiation of topics that are of direct interest for employees' working conditions, such as training, health and safety and restructuring (⁹⁵). This will contribute to ensuring a greater acceptance of changes and that due attention is paid to the most vulnerable workers (the low-skilled, older workers and workers with family responsibilities).

Under the ongoing structural changes, strengthening the EU's productivity growth and labour market resilience calls for work organisations that make full use of workers' knowledge potential and that increase the quality of their jobs. In this context, work organisations, and notably managerial structures, should be reformed to promote higher well-being and engagement of workers. Greater focus should be placed on intrinsic motivation of workers that feeds on the ability of using one's skills on the job, sense of purpose, autonomy in managing one's time and control over the substance and methods of work tasks.

More participative and empowering forms of work organisation should be developed to strengthen employees' involvement in innovation implementation (and therefore understanding and acceptance of tasks changes) and enable workers (especially the low-skilled) to gain the abilities that enhance their employability through life-long learning (e.g. Totterdill, 2014). Loyalty and incentives to acquire firm-specific skills should not be adversely affected as workers will have to show more flexibility within and between enterprises. Otherwise, workplace innovations may have a negative impact on productivity, labour market participation and job quality. Workplace innovations should also avoid perpetuating or sharpening the existing gender segregation in the workplace (96).

Learning organisations have the potential to foster intrinsic motivation, support workers' involvement and skill use/ development, and therefore improve companies' performance. Worryingly, recent years have seen a reduction in the number of Learning organisations and a move towards Lean organisations. A coherent and comprehensive policy response to support changes in work organisation towards more effective and beneficial forms of work organisation would be in the mutual interest of EU companies and their workers.

6. Conclusions

Job quality and work organisation are high on the EU policy agenda

Since the Lisbon Growth and Jobs Strategy launched in 2000, the European Employment Strategy's overarching objectives have encompassed not only full employment, but also the promotion of quality and productivity at work. In 2001, the Laeken European Summit agreed to a comprehensive framework on job quality, and appropriate quality indicators were included in the 2002 Employment Guidelines. With the Europe 2020 Strategy, launched in 2010, it also became a priority to

⁽⁹⁵⁾ See, for instance the case of GDF Suez launching an international social dialogue in 2011 at http://www.eurofound.europa.eu/ eiro/2011/01/articles/eu1101011i.htm

^(%) See, for instance, http://www.genderportal.eu/ and http://www.eurofound.europa.eu/areas/ industrialrelations/dictionary/definitions/ horizontalsegregation.htm

support workplace innovation aimed at improving staff motivation and working conditions with a view to enhancing the EU's innovation capability, labour productivity and organisational performance. In 2013, the Employment Committee Indicators Group agreed upon a four-dimensional concept of job quality reflecting the complexity of the concept of job quality (1. socioeconomic security, 2. education and training, 3. working conditions, and 4. work-life and gender balance).

The level of earnings, job security, the level of education and access to life-long training, a safe and healthy workplace, an appropriate balance between work intensity and job autonomy, employee participation and empowerment and an adequate balance between work and private and social responsibilities, are all job quality dimensions that can foster commitment, motivation and higher effort and reduce absenteeism with a direct impact on labour productivity and labour market resilience.

Some Member States, such as Italy, Spain, Greece, Cyprus or Portugal have a higher share of involuntary temporary contracts and lower transition rates to permanent employment compared to Austria, Germany or the Netherlands. Denmark, Sweden and Finland have high participation rates in life-long learning of more than 50% or 60%, while Greece, Spain, Italy, Romania and Bulgaria have participation rates that are half or less than half of the Nordic ones. High work intensity and low autonomy leads to high levels of stress in Germany and Austria, for example. Inactivity rates due to family responsibilities are higher in Ireland and the United Kingdom where the availability of child care facilities is low and/or costs are high.

In addition, strong differences in job quality across population groups persist, especially across skills level, gender and age. Such heterogeneity in job quality may not only have an adverse impact on social cohesion, but it may also have a negative feedback on the overall performance of the labour force. For example, persistent gender stereotyping in certain types of work continues to prevent an optimal labour allocation while at the same time reducing job and earnings opportunities of a significant part of the labour force.

The crisis has seen the deterioration of some dimensions of job quality and in work organisation

The crisis may have led to the deterioration of some of the job quality dimensions in several or most EU Member States. For example, participation in life-long learning went down in recent years in about one third of the Member States. In recent years, there has been a downward trend from Learning to Lean forms of work organisation. Learning work organisations represent the newer type of work organisation that have the potential to foster intrinsic motivation, support job quality including workers' involvement and skill development and use, and therefore improve companies' performance.

... while ongoing structural changes bring along opportunities for job creation and productivity growth ...

Further innovations in ICT and KETs broaden the scope for job creation in industrial activities which are often associated with jobs of high quality and value added and therefore earnings. Technology change allows for more flexible working arrangements and has the potential to mitigate some physical or psychosocial barriers which reduce the labour market participation of certain groups such as older workers, workers with disabilities and those with family responsibilities and entice them to remain in the workplace. Technology is also likely to change the job landscape of the future by putting a premium on creative and knowledge occupations and allowing for greater autonomy, responsibility and flexibility in the workplace.

Globalisation also has the potential to create new quality jobs reinforcing overall productivity growth and earnings potential. Expanding global value chains can allow further task specialisation and higher mobility, giving workers a larger choice of jobs and the opportunity to perform those tasks that best fit their abilities and preferences. The restructuring of global chains combined with technology may benefit the resource-poor, skills-rich European Union, as its skills structure may have a comparative advantage in world markets. The greening of the economy through recycling and reusing, together with the call for energy efficiency and biotechnology, is generating new production processes, new products and new markets. This has the potential to generate new jobs at all levels of skills. As such, structural changes can generate jobs, increase motivation and effort and therefore productivity growth.

... but also pose important challenges such as polarisation ...

Technology change, globalisation, demographic ageing and the greening of economy can have significant negative implications. Technology change may render an important share of tasks and jobs obsolete at a high speed. Globalisation requires specific skills to act in international environments (e.g. languages and ICT) which some workers lack. It may also lead to task relocation, notably of low-skilled routine tasks (or lower wages as a result of the threat of relocation). Green jobs may bring along new and unknown health and safety risks. The combination of technology change and globalisation emphasises the importance of knowledge and creativity and the need to adjust quickly to new and complex tasks, skills that some groups of workers lack. Therefore, low to middle skills may see stronger job insecurity or a worsening of their job quality: longer working hours and higher occupational risks but lower wages.

Therefore, in the absence of policy action, the gains in job quality from ongoing structural changes may be distributed in a non-equitable way, generating polarisation and in turn adverse feedback on productivity and labour market participation.

...calling for adequate policy responses to improve job quality and ensure a more equal distribution of the benefit potential associated with structural changes...

The analysis suggests that in addition to correcting the current unfavourable developments, policy makers will have to gear up to the opportunities and face up to the challenges posed by ongoing structural changes in technology, international trade and foreign direct investment, demographic change and the greening of the economy. To reap the full potential of ongoing structural changes, priorities for labour market policies include:

- strengthen the tools to anticipate and assess risks to job quality from ongoing structural changes (via stronger partnerships between governments, social partners and academic researchers with a special focus on SMEs);
- promote health and safety in the workplace in general and notably in relation to new technologies and products (through legislation, awareness-raising activities and monitoring);
- remove institutional barriers to labour mobility (e.g. by strengthening cross-border portability of social security benefits);
- combat gender and age stereotyping, discrimination and stigmatisation (via among others, legislation and awareness-raising activities and an adequate provision of enabling and support services);
- green mainstream education policies, training and skill formation (e.g. by promoting STEM careers (⁹⁷) for women and to increase the number of women in the green economy);
- reduce the informal sector;
- increase participation in life-long learning and on-the-job training,

potentially considering stronger and dedicated public support to SMEs;

- improve job profiling, job search assistance and the connection between employment services, together with removing fiscal incentives that hinder further labour market participation;
- target the most vulnerable (e.g. by focusing on the low-skilled trapped in poor working conditions);
- promote social dialogue at all relevant levels (company, sector, national and EU).

...to promote work organisation innovation that supports the knowledge-based economy of the future

For the resource-poor, skills-rich European Union, the strengthening of its innovation capacity will be crucial in order to be able to exploit its comparative advantages in world markets to the fullest extent. The analysis underlines the need to:

- promote employee empowerment (e.g. employees creating their own team structure, employees involved in the identification of problems and solutions in production);
- promote the exchange of experiences in work organisation innovation to help identify best practices;
- monitor the implementation and support the assessment of the impact

of the changes in work organisation on productivity and social cohesion;

- strengthen employee's **capacity to learn including through education and life-long learning** (e.g. meeting the needs of knowledge-intensive work process with rapid technical change);
- strengthen social skills for digital workplaces spread around the world (e.g. languages and cultural awareness);
- develop benchmarks with a view to promoting the full exploitation of the complementarity of educational systems and employee in-work training to the fullest extent, especially in SMEs;
- promote social dialogue adapted to expanding global value chains (e.g. involving counterparts in other countries to discuss minimum standards and conditions);
- target the most vulnerable workers

 (e.g. strengthening skill formation of
 workers with limited learning capacity).

Finally, it is important to recognise that the impact of job quality and work organisation on productivity and social cohesion is conditioned by worker, firm and country specific conditions. Therefore, designing and implementing measures to correct adverse developments and to promote positive developments will be a complex task taking account of country, sector and firm specificities.

^{(&}lt;sup>97</sup>) STEM: Science, Technology, Engineering and Mathematics.

ANNEX 1: DEFINITIONS OF JOB QUALITY

EMCO indicators

	Table A1.1: EMCO indicators for job quality				
Dimension	Sub-dimension	Indicators and source	Source		
		Mean monthly earnings in PPS, companies with 10 employees or more	SES 2010		
	1.1 Adequate	In-work at-risk-of-poverty rate	SILC		
1. Socio-economic	earnings	Transitions by pay level - Fraction of individuals with at least the same pay level as in the previous year	SILC		
security		Am well paid for the work I do	EWCS 2010, Q77		
		Involuntary temporary employment	LFS		
	1.2 Job and career	Labour transition - employment security	SILC		
	security	Labour transition temporary to permanent	SILC		
		Job offers good prospects for career advancement	EWCS Q77c		
	2.1 Skills development	CVT-hours per participating person	CVTS 2005		
		CVT participation	CVTS 2005		
		Main paid job involves learning new things	EWCS Q49f.		
		Tasks do require different skills	EWCS Q54.		
		On-the-job training over last 12 months	EWCS Q61c.		
2. Education		Present skills correspond well with my duties	EWCS Q60.		
and		Participation LLL, employed	LFS		
training		Participation LLL, unemployed	LFS		
-		Early leavers from education and training (% of population)	LFS		
	2.2 Employability	Percentage of the population aged 25–64 having completed at least upper secondary education	LFS		
		E-skills of adults - Computer skills. Persons at least medium computer skills	Questionnaire on I		

Dimension	Sub-dimension	Indicators and source	Source	
		Serious accidents at work per 100 000 persons	55.004	
		in employment	ESAW	
		FACTOR indicating non-exposure to unhealty	Questions	
		environment	23a - 23 i EWCS	
			Questions	
	3.1 Health and safety at	FACTOR indicating healthy physical conditions	24a - 24e EWCS	
	work	Well infomed on health and safety risks	Q30 EWCS	
		Think that health or safety is NOT at risk because of		
		vour work	Q66 EWCS	
		Work does NOT affect health	Q67 EWCS	
		FACTOR indicating non-exposure to harassment,	Questions 70 and	
		humiliation etc.	71 EWCS	
		No work when sick over last 12 months/not sick	074a EWCS	
		NOT working at very high speed	045a EWCS	
	3.2 Work intensity	NOT working to tight deadlines	045h FWCS	
		Enough time to get the job done	051a EWCS	
		No experiencina of stress in your work	051n EWCS	
		Workpace NOT dependent on automatic speed of a	Q32.1. E1103	
3. Working conditions		machine or movement of a product	Q46d EWCS	
		Workpace NOT dependent on the direct control of		
		vour boss	Q46e EWCS	
		"Occasionally/pover" interrupt a task in order to take		
	3.3 Autonomy	on an unforescent task	Q47 EWCS	
		EACTOR indicating colf-responsibility	Questions 49 51 EN/CS	
		"Toom members deside by themselves on	Questions 45-51 EWC5	
		the division of tacks"	Q57a EWCS	
		"Toom more decide by the machine the timestable		
		Team members decide by themselves the timetable	Q57c EWCS	
		of the work"		
		Union density	ICTWSS database	
		Collective pay agreement, share any	SES 2010	
		"Have raised work-related problems with an	Q62b EWCS	
	3.4 Collective Interest	employee representative over last 12 months"		
	Representation	"Employee is acting as an employee representative"	Q63 EWCS	
		"Management holds meetings in which you can	Q64 EWCS	
		express your views about what is happening in		
		the organisation"		
		Inactivity due to family or personal responsibilities	LFS	
		Part-time work due to family or personal	LFS	
		responsibilities		
		Lacking formal care for small children:	SILC	
		% of children <3 years not formally cared for		
		Employment impact of parenthood - men	LFS	
		Employment impact of parenthood - women	LFS	
	4.1 Work-life balance	Certain possibilities to adapt working time	Q39 EWCS	
4. Work-life and gender		Taking hour or two off to take care of personal or	043 EWCS	
balance		family matters is NOT (too) difficult ?		
		FACTOR indicating no long working hours	Questions 32–36 EWCS	
		Working hours fit with family or social commitments	041 FWCS	
		outside work very well or well		
		"Less often/never" worked in free time in order to		
		meet work demands		
		Gender pay gap	SES 2010	
	4.2 Gender balance	Gender employment gap	LFS	
		"Immediate boss a woman"	Q59 EWCS	

Laeken Indicators of Job Quality

The Laeken indicators of job quality include 10 dimensions, categorised into two themes: characteristics of the job/ worker (e.g. skills, working conditions, reconciliation between working and non-working life, health and safety at work, job satisfaction) and the wider socioeconomic and labour market context (e.g. employment rates, growth in aggregate labour demand)(⁹⁸).

The Laeken indicators constitute the biggest attempt at that time to construct an EU system of job quality indicators. Nevertheless, there have been some critiques. For example, both the European Commission (2008) and the European Parliament (2009) recognise that this set of indicators covers economy-wide areas not directly related to job quality while lacking very relevant indicators such as wages, work intensity and some more qualitative aspects of human capital formation⁽⁹⁹⁾.

Another issue is the inclusion of gaps (gender and age gaps). The European Parliament (2009) considers that in order to reflect differences in job quality for specific groups, the way to do this is to compute the variables of job quality for each of the subgroups and then compare the overall results between them(¹⁰⁰).

- (⁹⁸) The EU defined several specific indicators for evaluating each dimension, except in the case of social dialogue where no agreement was reached. The 10 dimensions of job quality are: intrinsic job quality; skills, life-long learning and career development: gender equality; health and safety at work; flexibility and security; work organisation and the work-life balance; inclusion and access to the labour market; social dialogue and worker involvement: diversity and non-discrimination: overall work performance. All available sources at EU level were used (e.g. LFS, ECHS, etc.). For more details, see http://ec.europa.eu/social/ BlobServlet?docId=2134&langId=en and European Commission (2008)
- (99) According to the EP, a good job quality index should not include any information that does not relate directly to the well-being of workers because it tends to skew the results. The European Parliament refers to several such dimensions in the Laeken Indicators such as access to the labour market, overall performance and productivity and variables measuring the quantity of jobs. While important because it gives the general context, according to the EP this type of information can form part of another index on the socioeconomic context, for example.
- (100) In fact, this problem stems from the fact that the indicators are measured only at the aggregate level, and to deal with distributional aspects some indicators are measured as gaps. This problem is overcome in the EWCS, which will be reviewed next, which allows to compute the various dimensions separately for men and women (alternatively allows for breakdowns by age, occupation).

The Laeken indicators represent a system of indicators with no aggregation between the different dimensions. While this does not require any pre-judgement on the relative importance of the different attributes, each observer may use their own subjective system of weighing, emphasising the features they consider most important.

Several other organisations, such as Eurofound, OECD, ILO and UNECE, have also made efforts to assess and quantify the quality of work as reviewed in the following paragraphs.

Eurofound: Quality of Work and Employment

The European Foundation for the Improvement of Living and Working Conditions, Eurofound, has been working on the measurement of the concept since 1991 in the European Working Conditions Surveys (EWCS). The questionnaire covers all major areas of job quality identified in the social sciences literature.

The survey is carried out every five years (1991, 1995, 2000, 2005, 2010). The scope of the questionnaire as well as the country coverage has widened substantially since the first edition(¹⁰¹).

The Eurofound's concept of work and employment quality (see Eurofound, 2002) has four main dimensions: career and employment security, health and well-being, skills development, reconciliation of working and non-working life.

Historically, the EWCS has not come up with an index of job quality, but rather with a 'system' of indicators on job quality. In a study based on the 5th EWCS (¹⁰²), Eurofound presented, however, four composite indices of job quality: an Earnings index, a Working Time Index, a Career Prospects Index, and an Intrinsic Job Quality Index(¹⁰³)(¹⁰⁴). To illustrate the complexity, the index of intrinsic job quality, for example, is composed of a whole set of indicators measuring skills and discretion; good social environment; good physical environment; and work intensity(¹⁰⁵).

An advantage of the EWCS is that it is well documented and harmonised. The same questionnaire is used in all countries, which allows cross-country comparisons. However, because of important changes in the questionnaire, comparisons over time are possible only for a core of key questions which were retained unchanged since 1991.

One issue with the EWCS is its periodicity: it is conducted every five years. Also worth mentioning is the sample size, which does not allow for too many levels of breakdowns. Nevertheless, gender mainstreaming has been an important concern for recent reviews of the questionnaire, and the most recent addition allows for breakdowns by age, gender and occupation.

The EWCS has also been used as a basis for development of other job quality indices/systems of indicators by other organisations, for example, the EMCO indicators list or the European Trade Union Institute Job quality index (see below).

(105) Table 1 in Eurofound (2012b), p. 20, gives a brief description of the content of each index and survey questions on which it is based.

^{(&}lt;sup>101</sup>) The latest, 5th EWCS is available at: http://www.eurofound.europa.eu/surveys/ ewcs/2010/. For more details about the different extensions by waves, see Eurofound (2010), p. 141.

⁽¹⁰²⁾ Eurofound (2012b)

⁽¹⁰³⁾ Regarding the methodology of composing the indices, based on statistical correlations similar items were identified and normalised, and then grouped in a summative index. When multiple indices are aggregated together they were accorded equal weights, except where it was found that the indices had considerably different associations with subjective well-being. The weighting assumptions are accompanied by a sensitivity analysis. More methodological details are available in Chapter 2 of Eurofound (2012b).

Eurofound discusses the pros and cons of producing a single job quality index. This might be justified from a rather pure theoretical perspective, whereby it is assumed to be a utility associated with each job, i.e. the index is seen as measuring that utility. One feature that makes a single index very appealing is its tractability, ease of presentation, and ease of crosscountry comparisons. However, this argument is firstly not very persuasive since job quality, as discussed above, is a multi-faceted concept Secondly, it risks being interpreted differently by different users. For example, economists will tend to think about wages, social scientists about non-wage aspects, etc. Last but not least, to compute such an index would require ver strong assumptions about how individuals trade off job quality features against each other. The choice of four indices presented by Eurofound is something of a middle solution: they are smaller in number and allow country rankings in a meaningful way; yet, they sufficiently well portray the different aspects of job quality without mixing them up.

OECD Job quality indicator

A recent ongoing project, 'Defining, measuring and assessing job quality and its links to labour market performance and well-being' within the OECD, co-funded by the European Union, started in September 2013 and will run for two years. It starts from the insights provided by the EU flagship initiative *New Skills and Jobs in Europe*, the OECD *Re-assessed Jobs Strategy* and the OECD *Better Life Initiative*(¹⁰⁶).

The new OECD framework for measuring and assessing job quality considers three dimensions of job quality that are both important for worker well-being and relevant for policy, and together allow for a comprehensive assessment of job quality.

- Earnings quality refers to the extent to which employment contributes to the material living standards of workers and their families. While the average level of earnings provides a key benchmark for assessing the degree to which having a job ensures good living conditions, the way earnings are distributed across the workforce also matters for well-being. Therefore, the OECD measures earnings quality by a synthetic index that accounts for both the level of earnings and their distribution across the workforce.
- Labour market security captures those aspects of economic security that are related to the risk of job loss and its consequences for workers and their families. For OECD countries, labour market insecurity is defined in terms of the risk of becoming unemployed and its expected cost. The latter depends both on the expected duration of unemployment and the degree of public unemployment insurance. Labour market security is therefore defined in terms of the risk of unemployment, which encompasses both the risk of becoming unemployed and the expected duration of

unemployment, and unemployment insurance, which takes into account both benefit coverage among the unemployed and benefit generosity.

Quality of the working environment captures non-economic aspects of job quality and includes factors that relate to the nature and content of work performed, working-time arrangements and workplace relationships. Jobs that are characterised by a high level of job demands such as time pressure or physical health risk factors, combined with insufficient job resources to accomplish job duties, such as work autonomy and good workplace relationships, constitute a major health risk factor for workers. Therefore, the OECD measures the quality of the working environment by incidence of job strain, which is a combination of high job demands and few job resources.

While the three dimensions of job quality are key elements of the new framework, their actual measurement is flexible and can be adapted according to the purpose for which they are being used, data availability and different choices for weighting together the different sub-components. In order to ensure that indicators of job quality are conceptually sound and relevant for policy, the framework provides three guiding principles. These are to: i) focus on outcomes experienced by workers as opposed to drivers of job quality; ii) emphasise the objective features of job quality; and iii) derive indicators from data on individuals to allow going beyond average tendencies.

Chart A1.1, Chart A1.2 and Chart A1.3 report the values of the three job quality measures (earnings quality, labour market insecurity and job strain) for each country in the dataset.

Chart A1.4 plots the cross-country averages of different measures of job quality for different worker characteristics. For more details, see OECD 2014.

Overall, job quality outcomes vary substantially across OECD countries across each of the three dimensions:

- Denmark, Finland, Germany, Luxembourg, the Netherlands, New Zealand, Norway, Sweden and Switzerland are among the best performers. These countries do relatively well along at least two of the three main dimensions of job quality, without any outcomes in the bottom-10 of the ranking across OECD countries.
- Australia, Austria, Belgium, Canada, the Czech Republic, France, Ireland, Israel, Italy, Japan, Korea, Mexico, Slovenia, the United Kingdom and the United States display average performance. Over the three main dimensions of job quality, these countries display no more than one outcome in either the top-10 or the bottom-10 of the ranking across OECD countries, except for Ireland and Korea where the picture is more mixed.



Note: Moderate inequality aversion; see OECD 2014.

(1) Earnings Quality is measured as the Harmonic Mean of the earnings distribution in each country. Like other types of 'general means', the harmonic mean can be expressed as a function of the simple arithmetic mean and of a measure of earnings inequality. As such, it lends itself to being an encompassing measure of earnings quality, since it captures both the average of earnings and their distribution. See Section 2.1 in Chapter 3 of Employment Outlook 2014 for a detailed discussion. gif excel file

^{(&}lt;sup>105</sup>) The project is structured into seven work packages: 1. Job quality: what does it mean and how can it be measured?
2. Measuring work-related economic security and its determinants. 3. Measuring quality of working life and its determinants.
4. Reassessing labour market performance when accounting for job quality.
5. Maintenance of a permanent database on job quality. 6. The role of policies and institutions for job quality and employment performance. 7. Job quality in emerging economies.

 Estonia, Greece, Hungary, Poland, Portugal, the Slovak Republic, Spain and Turkey do relatively badly in two or all of the three main dimensions of job quality. In addition, none of these countries perform very well along at least one of these dimensions.

Looking at job quality outcomes across socio-economic groups provides new insights into labour market inequalities by shedding further light on the nature and depth of the disadvantages faced by some population groups.

- Youth and the unskilled face the worst outcomes with respect to job quality. By contrast, high-skilled workers perform well in all dimensions. For women, the picture is mixed. While men tend to enjoy higher earnings, women tend to enjoy a better quality working environment. The degree of labour market security is similar between men and women.
- Temporary work is strongly associated with poor job quality in all three dimensions. Part-time work, on the



Source: OECD, Employment Outlook 2014.

Note: Labour market insecurity: unemployment risk times one minus unemployment insurance which may be interpreted as the expected earnings loss associated with unemployment as a share of previous earnings.

excel file

Labour market insecurity is defined as uninsured labour market risk. More specifically, it is calculated as the ratio of the probability of becoming unemployed over the probability of finding employment, times one minus the effective rate of risk-absorption through the tax and benefits system. The latter can be viewed as the rate at which the tax and benefits system is able to 'replace' workers' earnings when they lose their job. See Section 2.2 in Chapter 3 of Employment Outlook 2014 for details.



Source: OECD, Employment Outlook 2014.

Job strain is produced by the interaction of high job demands and limited job resources. Job demands require sustained physical, cognitive and emotional effort. Resources include work autonomy, appropriate feedback, opportunities to learn and support from colleagues and managers. In the OECD Employment Outlook 2014, job strain is characterised by a set of combinations of job demands and resources that are most likely to have detrimental effects on workers' health (see Section 2.3 for exact definition of such combinations). other hand, is associated with weaker outcomes in terms of earnings and labour market security, however, the risk of job strain tends to be lower among workers on part-time contracts compared to the full-time workers.

ILO Decent Work Agenda

The ILO Declaration on Social Justice for a Fair Globalization, adopted in 2008, endorses the Agenda for Decent Work, which includes four equally important strategic objectives: creating jobs, guaranteeing rights at work, extending social protection and promoting social dialogue, with gender equality as a crosscutting objective.

The same year, the ILO adopted a comprehensive framework of Decent Work Indicators to monitor progress. The framework contains no country rankings and no composite index, and covers all four dimensions of Decent Work. The information is derived from various sources: household and establishment surveys, administrative records, qualitative legal framework information, among others.

The framework is based on both *statistical indicators* and qualitative information on the rights at work and the legal framework(¹⁰⁷) to take cognisance of the contextual environment in which the progress occurs. Progress of countries is recorded in the Decent Work Country Profiles. The ILO Manual on Decent Work Indicators: concepts and definitions was launched in 2012 (¹⁰⁸).

- (107) The statistical indicators cover the broader economic and social context as well as 10 thematic areas (employment opportunities, adequate earnings, working time, combining work and family life, child and forced labour, stability and security of work, equal opportunities, safe work environment, social security, social dialogue). The legal framework indicators are divided into 21 groups, some of which are labour administration, minimum wage, unemployment insurance, leave (paid annual leave, maternity and parental leave), child and forced labour, termination of employment, employment injury benefits. pension, incapacity due to sickness/ invalidity, freedom of association, collective bargaining, tripartite consultation. More information is available at http://www.ilo. org/integration/themes/mdw/lang--en/index. htm, which gives access also to the specific Decent Work Factsheets and Country Profiles as well as the Manual on Decent Work Indicators, see next footnote
- (¹⁰⁸) The link to the manual is: http://www.ilo. org/wcmsp5/groups/public/----dgreports/---integration/documents/publication/ wcms_229374.pdf. It presents a description of the statistical indicators and legal framework indicators related to the 10 substantive elements of decent work.

file

excel





excel file gif

Earnings quality and labour market insecurity data show an average for 2005-10, while job strain refers to 2010. (1)

The ILO provides support through integrated Decent Work Country Programmes developed in coordination with its constituencies. These programmes define the priorities and the targets within national development frameworks and aim to tackle major Decent Work deficits towards each of the strategic objectives. The country profiles provide an input for the Country Programmes and help spell out the targets.

There emerged synergies between the EU and the ILO's job quality strategies. Implemented by the ILO with funding from the European Union, the project 'Monitoring and Assessing Progress on Decent Work (MAP)' (2009 to 2013) involves joint work with government agencies, Statistical Offices, workers' and employers' organisations and research institutions to strengthen national capacity, particularly of developing and transition countries, to self-monitor and self-assess progress towards decent work. The project further facilitates the identification of decent work indicators that are relevant at the national level, supports data collection, and uses the collected data for an integrated policy analysis of decent work.

Source: European Union Survey on Income and Living Conditions (EU-SILC), European Working Conditions Survey (Eurofound, 2010), OECD Employment Database Note: Country coverage: Austria, Belgium, Czech Republic, Denmark, Estonia, Finland, France, Germany, Greece, Hungary, Iceland, Ireland, Italy, Luxembourg, Netherlands

Norway, Poland, Portugal, Slovenia, Slovak Republic, Spain, Sweden, Turkey and United Kingdom (24 countries, 23 countries excluding Iceland in Panel C).

The ILO Key Indicators of the Labour Market (KILM)

Published every two years since 1999, the KILM is a collection of 20 key indicators of the labour market, ranging from employment and variables relating to employment (status, sector, hours, etc.) to education, wages and compensation costs, labour productivity and working poverty. These indicators are relatively broad, capturing the economic and labour market situation in a country, but provide less insight into the quality of employment/jobs.

UNECE Task Force on measuring quality of employment

Since 2000, UNECE, Eurostat, the OECD and ILO organise joint seminars on quality of employment to share information between international experts and to develop a quality of employment framework. The new framework does not seek to reconcile the existing frameworks used in the different policy contexts: the ILO's Decent Work Indicators Measurement Framework, the EU Quality of Work Indicators, and the Eurofound's guality of work and employment framework. Rather, it aims to provide a 'toolbox' of indicators to be used for international and national initiatives to study quality of employment.

In 2007, under the auspices of the Conference of European Statisticians, a Task Force⁽¹⁰⁹⁾ was set up to develop a concept for statistical measurement of quality of employment unifying the elements in the existing approaches.

⁽¹⁰⁹⁾ The Task Force was composed of representatives from national statistical offices of Canada, France, Finland, Hungary, Israel, Italy, Poland, ESTAT, Eurofound, ILO, UNECE and the NGO Women in Informal Employment (WIEGO).

The 2007 Task Force created an initial framework for measuring quality of employment with seven dimensions and over 50 indicators (¹¹⁰). The framework was implemented by nine countries by the end of the Task Force's term, leading to nine pilot country reports (¹¹¹).

The ILO decent work concept and the UNECE Task Force-proposed set of indicators are designed to capture aspects of labour markets in both developing and developed countries, and thus they put more emphasis on labour rights (including no child and forced labour) and social protection aspects in their definitions than the European Commission's and Eurofound frameworks.

In 2012, with a time frame of 2012–15, an Expert Group on Measuring the Quality of Employment was established within the framework of the Conference of European Statisticians, with the main objective to revise the conceptual structure and the set of indicators of the quality of employment.

European Trade Union Institute's (ETUI) Job Quality index(¹¹²)

The ETUI started work on this issue in 2008. The job quality index (JQI) comprises six dimensions based on 16 indicators, which in turn are drawn from individual variables taken from different sources(¹¹³). The six dimensions are: wages, non-standard forms of employment, working conditions, working time and work-life balance, access to training and career advancement, and collective interest representation and participation.

- (¹¹⁰) 1) safety and ethics of employment (safety at work, child and forced labour, fair treatment in employment); 2) income and benefits from employment, including also non-wage pecuniary benefits; 3) working hours and balancing work and non-working life (working hours, working time arrangements); 4) security of employment and social protection; 5) social dialogue; 6) skills development and training; 7) workplace relationships and work motivation. For more information see UNECE (2009).
- (¹¹¹) Canada, Mexico, Finland, France, Germany, Israel, Italy, Republic of Moldova, Ukraine. See UNECE (2010).
- (¹¹²) ETUI is the research arm of the European Trade Union Confederation (ETUC), the most important representative body of workers at EU level and a major player as a social partner in the EU policy area of work and employment.
- (¹¹³) The index is based on five sources: LFS, SILC, AMECO, EWCS and ICTWSS (the latter stands for Database on Institutional Characteristics of Trade Unions, Wage Setting, State Intervention and Social Pacts).

This index is focused on job quality from the perspective of workers; it captures most of the areas of job quality from the social sciences literature.

Although the variables refer to information measured at the level of individual workers, job quality is computed and reported only at national level based on averages (¹¹⁴). Apart from a readily available gender breakdown, it is not possible to break it down in order to analyse specific groups of workers (e.g. by occupation, type of contract).

The results are consistent with most of the results from other indices: best performers are the Northern countries, and lowest values are found for Eastern and Southern European countries. The JQI also allows comparisons over time for the EU-15 countries. However, caution needs to be taken as the index is based on various data sources, not all of which are updated with the same periodicity.

European Job Quality Indicator: European Parliament

The European Parliament (2009) came up with an outline for the development of a European Job Quality Index. The authors suggest that the new indicator should be based only on variables that directly affect the quality of work and employment. Ideally, it should be constructed from individual data. They suggest as a leading source the EWCS. The future indicator should include the following dimensions: work, employment, and a joint dimension for work and employment (see European Parliament 2009).

(114) It is constructed by first normalising the indicators for each dimension, then weighting the normalised indicators within each dimension, and then summing up the dimensions (i.e. each dimension is equally weighted or equally important for the overall result of the index). As for each composite index this method involves some discretion in choosing the weights. The sensitivity analysis shows however that the results are quite stable to changing the weights (Leschke, Watt and Finn, 2008). The indicators are normalised by rescaling each value to the proportion they represent with respect to the difference between the maximum and minimum values for the base year, which is set to 2000 for EU-15 and 2007 for EU-27. This system of normalization is a widely used method for comparing countries' performance (for example, in the construction of the Human Development Index) and has the advantage of putting each value in relation to the best and worst cases. For more methodological details see Leschke, Watt and Finn, and 2012.

EU Seventh Framework Programme

Working conditions and job quality have been a prominent feature under the socio-economic research programmes of the EU's Research Framework Programmes. Below, the two most recent and relevant European research projects on this theme are briefly presented. For a more exhaustive overview of recent comparative research in Europe, see Chapter 4 'Toward better job quality and working conditions: increasing productivity and workrelated well-being' in the European Commission Policy Review, 'New skills and jobs in Europe: Pathways towards full employment' (115).

NEUJOBS project

NEUJOBS is a research project financed by the European Commission under the Seventh Framework Programme (FP7-SSH). Its objective is to analyse likely future developments in the European labour market(s), in view of four major transitions that are expected to impact employment and European societies in general(¹¹⁶).

The WP 2 (work package) called 'Good jobs-bad jobs, cultural attributes of decent work in Europe' looks at issues of job quality. The package considers the conceptualisation of job quality from both the labour law perspective and the perspective of employees through case studies and in-depth face-to-face interviews (¹¹⁷). In contrast to previous approaches, the NEUJOBS project does not seek to measure job quality nor come up with particular

(¹¹⁷) The interviews are semi-structured qualitative interviews, with many open questions. However, they give additional valuable information and allow taking into account cultural aspects. There are five groups of actors interviewed: social partners, governments and parties, civil society organisations, research communities, and separately, employees.

^{(&}lt;sup>115</sup>) Publications Office of the European Union, Luxembourg, 2012. http://ec.europa.eu/research/social-sciences/ pdf/new-skils-and-jobs-in-europe_en.pdf

^{(&}lt;sup>116</sup>) These transitions are: 1. socio-ecological transition (a change in the patterns of social organisation and culture, production and consumption beyond the current industrial model towards a more sustainable future); 2. societal transition produced by a combination of factors like population ageing, low fertility rates, changing family structures, urbanisation and growing female employment; 3. new territorial dynamics and the balance between agglomeration and dispersion forces; and, 4. skills (upgrading) transition and its likely consequences for employment and (in)equality.

indicators/dimensions thereof. On one hand, it concentrates on what is found in the labour codes, employment laws/guidelines, government plans, trade union strategies, NGO agendas and academic works with regard to job quality. Additionally, it tries to understand the attitudes of employees towards work and explain cleavages between 'collective' views expressed in the employment programmes/ labour law and those expressed by the employees themselves.

The project has already published a state-of-the-art report on job quality (¹¹⁸). The countries covered are Spain, Hungary, Slovakia and the United Kingdom (two 'old' and two 'new' Member States). More recently, the project finished its comparative qualitative ('quasianthropological') research (119), in which it finds the mainstream 'postmaterialist' academic discourses on good jobs (mainly obtained from quantitative surveys) quite distant from the preoccupations of the workers interviewed in these four countries. Researchers appeared to observe a 'retraditionalisation' of employment preferences (security-oriented: full-time work with permanent contracts and appropriate wages) and found sectoral and company type features to be more defining for job quality than the national contexts.

WALQING project: Work and Life Quality in New and Growing Jobs(¹²⁰)

Funded by the European Union's Seventh Framework Programme (FP7-SSH) from 2009-2012 and involving 11 European partners, the Walqing project investigated the linkages between new and expanding jobs, the conditions of work and employment in these jobs, and the outcomes for employees' quality of work and life. It did so by integrating several analytical levels and research paradigms. In particular, research in Walqing is divided into three pillars: 1. Data analysis — Employment growth, quality of work and quality of life in Europe; 2. Stakeholder involvement Comparative institutional analysis and action research; and 3. Qualitative research — Organisational strategies, vulnerability and individual agency.

Under the first pillar, in-depth analyses of the most important European data sources, such as EU-LFS, EWCS, EU-SILC and ESQL were used to identify 'new and growing' jobs and to assess the quality of jobs and life in these growth areas, particularly with regard to jobs with problematic working conditions in the service and manufacturing industries (¹²¹). Pillar 2 performed institutional analysis and action research to

disseminate good-practice examples aimed at improving working conditions beyond their national, company-specific or sectoral contexts. In particular, the approach involves interviews with representatives of key stakeholders about the emergence of lowquality jobs and vulnerable groups in the selected sectors and policy documents reviewed. It developed and disseminated strategies for improving unhealthy or dysfunctional working conditions to foster mutual learning and dialogue among stakeholders (122). Pillar 3 explored the practices of work organisation, HRM strategies, contractual relations and working conditions, by means of 53 in-depth case studies in companies.

The research focused on five sectors with substantial growth potential in quantity & quality of jobs: Commercial Cleaning, Contract Catering, Green Construction, Mobile Elderly Care and Waste Management. Moreover, these sectors address basic human needs and are difficult to delocalise. The main findings and recommendations were summarised in five sectoral brochures on good working practices and social dialogue issues (¹²³). This included an analysis of particularly vulnerable groups, such as young workers, older workers, migrants and some groups of women (¹²⁴).

(¹¹⁹) 'Travelling back in time? Job Quality in Europe as seen from below', Kovacs with Hilbert, Veselkova and Virag (2014) http://www.neujobs.eu/

⁽¹²⁰⁾ http://www.walqing.eu/index.php?id=2

^{(&}lt;sup>118</sup>) Kovacs with Hilbert, Veselkova and Virag (2012)

^{(&}lt;sup>121</sup>) Key final reports include: 'Comparative analysis of employment expansion and of job characteristics in selected business functions', 'Comparative analyses of job quality in new growth jobs', and 'Secondary analysis on working conditions and quality of life' (all available at http://www.walqing.eu/ index.php?id=29).

⁽¹²²⁾ See for example 'Synthesis report on sector specifics in stakeholder policies and quality of work and life', available at http://www.walqing.eu/index.php?id=32

⁽¹²³⁾ Available at: http://ww.walqing.eu/webresource

^{(&}lt;sup>124</sup>) See for example 'Integrated report on organisational case studies', available at: http://www.walqing.eu/index.php?id=34

Annex 2: Organisation of work — Technical details

Criteria for classification

Classification of work organisation is established on the basis of 15 dimensions that describe relevant and discriminating aspects of work organisation:

- Two binary variables measuring autonomy in work:
 - Autonomy in choosing methods of work;
 - Autonomy in pace or rate at which work is carried out.
- Two binary variables measuring the way quality is controlled:
 - Use of precise quality standards;
 - Self-assessment of the quality of work.
- Three binary variables measuring the cognitive dimensions of work:
 - Complexity of tasks;
 - Learning new things in work;
 - Work requires problem-solving.
- Four binary variables measuring constraints of the pace or rate of work:
 - Constraints linked to the equipment speed or movement of a product in production flow;
 - Constraints relating to numerical production or performance targets;
 - Constraints due to direct control by worker's immediate supervisors;
 - Constraints resulting from dependency on the work done by worker's colleagues.
- Three binary variables measuring degree of novelty in job tasks:
 - Perceived monotony of tasks;
 - Repetitiveness of tasks of less than one minute.
 - Task rotation between colleagues
- A three-level variable measuring of the use of teamwork, with categories of autonomous teamwork (team members decide the division of tasks), non-autonomous teamwork (managers/supervisors decide the division of tasks) and no teamwork(¹²⁵).

Different models of work organisation

The typology initially developed by Lorenz and Valeyre builds on a review of the literature on work organisation covering High Performance Work systems (HPWS) (Appelbaum and Batt, 1993, 1994; Pfeffer, 1998; Osterman, 1994), the lean production model (MacDuffie and Pil, 1997), the socio-technical system (Emery and Trist, 1960), learning organisations (Zarifian, 2003), Tayloristic organisations and adhocracies (Mintzberg, 1979). This review led to the identification of 15 dimensions that describe relevant and discriminating aspects of work organisation covering autonomy in work, quality control, cognitive dimensions of work, constraints of the pace or rate of work, novelty in job tasks and teamwork, and can be measured by the EWCS. In the 2010 survey, the same 15 dimensions to determine the presence and size of the four types of work organisations is used.

Traditional forms of work organisation are based on the principles of labour division, hierarchical and centralised authority and control. They are designed as static structures, optimised for a fixed set of external economic, social and cultural conditions. However, the emergence of new and uncertain environments has put these traditional work structures under increasing pressure. As a response, new forms of work organisation have emerged that are more flexible and more responsive to changing internal or external circumstances. Many of these 'new' forms are often grouped together under the label 'High Performance workplaces' (HPWP), but this group is far from being homogeneous and covers some of the defining characteristics of different organisational forms such as the socio-technical systems (STS), the learning organisations, lean production, high performance work systems and the adhocracy (Mintzberg).

Combs et al (2006), in a meta-analysis of 92 recent studies on HPWS and performance, found evidence that HPWS enhance organisational performance. An increase in one standard deviation in the use of HPWS is associated with a 4.6% increase in gross return on assets and a 4.4 percentage-point decrease in turnover from 18.4 to 14.0%. The effect is stronger when bundles of measures are considered together rather than individual practices. Effects sizes are larger in manufacturing industries than in service industries. Common to these HPWP forms are attention to knowledge as a competitive factor, decentralisation of decision-making and self-managed teams, performancebased compensation structures and rather extensive training and strong problem-solving opportunities. However, lean and learning forms of work organisation differ on a number of points such as:

- A higher level of individual autonomy granted to workers (higher in learning organisations as well as in the sociotechnical models but lower in lean production models);
- A higher level of standardisation in lean forms of work organisation (tasks, quality standards, etc.);
- A higher interdependent work structure as well as higher dependence on technologies to set the pace of workers and reliance on team work in lean production forms;
- An emphasis on workers' autonomy in organising and controlling the products of their work, decreased interdependency of work process in learning organisations;
- Attention to quality of working life is a key driver for example in the case of STS;
- While learning in lean production forms of organisation is mostly used to improve the work processes and increase productivity, learning activities in the case of learning organisations are seen as a critical activity for responding to unforeseen events and for the introduction of important innovations.

In contrast to Tayloristic organisations, which have been abundantly criticised for their physically demanding work, repetitive tasks and low learning opportunities, because conception is distinct from execution, lean production is designed to improve the overall performance of the organisation by assigning more autonomy to workers and their immediate managers, but with continued emphasis on the strict quality standards, standardisation of work and procedures and with reliance on individual performance-based pay structures. Lean forms of work organisation differ in a number of ways from the STS in that they promote development of more specialised, contextualised skills, organise work into wider production systems (greater interdependency) and provide feedback and support that are based on the degree to which strict performance criteria are satisfied.

⁽¹²⁵⁾ In the analyses of trend data a binary variable measuring presence of teamwork was used, since a three-level variable was not available in the 2000 EWCS dataset.

ANNEX 3: ADDITIONAL INDICATORS RELATING TO JOB QUALITY

Socioeconomic security





Job insecurity

During the crisis involuntary temporary work increased in a number of Member States, more significantly in Ireland, the United Kingdom, Luxembourg, Denmark and some New Member States (Slovakia, the Czech Republic, Hungary) (Chart A3.3), while transitions to permanent contracts worsened (Chart A3.4), most significantly in Slovakia (29 pps) and Spain (15 pps)(¹²⁶).

⁽¹²⁶⁾ Transitions improved notably in Finland (21 pps), Germany (14 pps) and Portugal (11 pps).







Chart A3.6: Labour transitions temporary to permanent, males, 2007-11



Notes: 2011 observation not available for IE and DK. 2007 observation not available for EU-28, HR, RO, UK and DK.



Notes: 2011 observation not available for IE and DK. 2007 observation not available for EU-28, HR, RO, UK and DK.



Prospects for career advancement

In some Member States job insecurity goes together with lower perceptions for career advancement, e.g. Romania, Slovakia, Italy, but the relationship is far from clear (Chart A3.9). The perceptions are also low in Member States like Germany and Austria where involuntary

temporary work is the lowest(127). In fact, career advancement prospects seem to be higher in countries where jobs involve more training and learning new things (128).

- (127) In fact the correlation between iob security and perceptions about career advancement (as measured by EWCS question 77c) is negative but close to zero.
- (128) The correlation coefficient between the two Eurofound indicators ('Job offers good prospects for career advancement', on one hand, and 'Job involves learning new things' on the other) is 0.5 significant at the 1% level.

excel file

jj



Work-life balance



Gender balance



excel file

Sources: WiiW (2014, Tables A1 to A3) based on EU Structure of Earnings Survey data, release 2002, 2006 and 2010.

Notes: The coefficients are taken from the full Mincer regressions estimated separately for each country. Countries are ranked according to gender wage gap in 2010. No information on career breaks available in the dataset. Since women are more likely to take career breaks, which may negatively impact upon their wages, a failure to control for career breaks will bias the estimates of the wage gap slightly upwards.

Table A4.1: Trend developments across Member States — increased learning EWCS survey wave Total Country 2000 2005 2010 38.2% 48.6 %_{a b} 57.6% 511% Learning Lean 40.9% 35.6% 29.7% 33.6% Malta 8.2% 5.6% Taylorist 7.3%_{a.b} 3.6% 9.6% Simple 13.6% 7.5% 9.1% 29.8% 28.5% 44.0% 34.4% Learning Lean 33.2% 27.1% 38.3% 32.4%_{a b} l atvia Taylorist 14.9% 15.6% 10.0% 13.4% 19.0% Simple 28.2% 17.6 %_L 13.7% Learning 23.8% 26.7% 35.2% 27.7% 33.3% 22.9% 25.4% Lean 21.7% Portugal Taylorist 30.7% 26.0% 26.7% 28.3% Simple 23.8% 14.0 %_b 15.3 %_b 18.6% 22.5% Learning 17.3% 23.6% 25.2% Lean 39.1% 40.1% 38.8% 39.3% Romania Taylorist 30.2% 25.7%_{a b} 19.6% 24.6% 13.5% 10.5% 16.4% Simple 13.4% Learning 59.6 %_{a b} 54.3 %_b 63.5% 59.3% Lean 22.9% 19.4% 20.5% 13.8% Netherlands Taylorist 11.4% 9.6% 94% 8.3% 13.1% 11.9% Simple 11.6% 11.4% 64.7% 58.4% 61.1% 621% Learning Lean 18.9% 29.8% 23.2% 22.9% Denmark 10.7% 5.0% 6.1% 8.0% Taylorist Simple 5.7% 6.9%_{a b} 9.6% 7.1% Learning 40.5% 25.6% 33.3% 32.7% 235% Lean 20.6 %_{a b} 30.1 %_b 20.6% Cyprus Taylorist 15.9% 14.7% 22.4% 18.4% 25.3% Simple 23.0% 29.5% 23.7% Learning 40.5% 36.0% 38.4% 38.6% Lean 38.6% 40.9% 39.6% 39.6% Estonia Taylorist 9.8% 9.7% 11.0% 102% Simple 13.4% 11.0% 11.6% 11.0% 37.6% Learning 36.7% 36.2% 39.1% 25.4%_{a b} Lean 33.3 %_b 20.9% 25.8% Poland Taylorist 14.2% 15.0% 19.6% 16.7% 19.9% Simple 23.8% 15.4% 20.4%_{ab} 26.6% Learning 24.2% 27.0% 28.1% Lean 19.6% 29.6% 30.9% 27.1% Lithuania Taylorist 19.6% 19.4% 18.5% 19.2% 36.6% 24.0 %_b 22.5 %_b Simple 27.1%

ANNEX 4: TREND DEVELOPMENTS IN LEARNING ORGANISATION

Source: Eurofound estimates based on EWCS.

Notes: Subscripts denote whether difference between values across columns (from different EWCS waves) are statistically significant at the 0.05 level. In particular, values having the same subscripts do not differ significantly while values with different subscripts are significantly different.

185

gif

Table A4.2: Trend developments across Member States — decreasing learning, but increasing lean forms						
			EWCS survey wave			
Country		2000	2005	2010	Total	
	Learning	47.6 % _a	45.2% _{a, b}	41.6% _b	44.4%	
6	Lean	16.1 % _a	19.1 % _a	25.8% _b	21.2%	
Germany	Taylorist	16.6 % _a	16.2 % _a	15.5% _a	16.0%	
	Simple	19.7 % _a	19.4% _a	17.0% _a	18.4%	
	Learning	41.6% _{a, b}	49.3 % _b	34.7% _a	41.2%	
Lungershauma	Lean	22.9% _a	28.4% _{a, b}	33.6% _b	29.1%	
Luxernbourg	Taylorist	12.7% _{a, b}	10.9% _b	19.1% _a	14.8%	
	Simple	22.9% _a	11.4% _b	12.6% _b	14.9%	
	Learning	44.9% _a	48.4% _a	42.6% _a	44.0%	
	Lean	18.4 % _a	25.1% _b	29.2% _b	25.7%	
Belgium	Taylorist	17.0% _a	11.2% _b	13.3% _{a,b}	14.0%	
	Simple	19.7 % _a	15.2% _{a.b}	14.9% _b	16.3%	
	Learning	51.8% _a	44.7 % _a	44.5%	47.6%	
	Lean	24.7 % _a	25.3% _a	30.4%	26.7%	
Austria	Taylorist	13.5%	20.3 % _b	14.7%	15.6%	
	Simple	9.9%	9.7%	10.4%	10.0%	
	Learning	45.0%	42.7 % _a	42.2%	43.1%	
	Lean	21.2%	31.8% _b	30.5% _b	28.1%	
Slovenia	Taylorist	17.3%	12.8%	13.0%	14.2%	
	Simple	16.5%	12.8%	14.3%	14.6%	
	Learning	41.7%	42.5%	40.4%	41.4%	
	Lean	17.8%	20.4%	23.7%	20.5%	
Italy	Taylorist	20.1 % _a	21.0%	14.8%	18.4%	
	Simple	20.4% _a	16.1 % _a	21.1%	19.6%	
	Learning	44.1 % _a	40.9% _a	43.8% _a	43.0%	
	Lean	30.2 % _a	32.7% _{a, b}	38.6 % _b	33.3%	
Finland	Taylorist	15.5 % _a	13.9% _{a, b}	8.8 % _b	13.2%	
	Simple	10.1 % _a	12.5 % _a	8.8% _a	10.5%	
	Learning	22.7 % _a	41.2% _b	22.6%	27.7%	
Iroland	Lean	32.9% _{a, b}	27.7 % _b	37.4% _a	32.7%	
neidhu	Taylorist	23.0% _a	12.4% _b	24.4% _a	20.5%	
	Simple	21.4% _a	18.7 % _a	15.6% _a	19.1%	
	Learning	57.0% _a	73.2 % _b	66.5 % _b	64.1%	
Sweden	Lean	18.9% _a	14.5 % _a	19.7% _a	17.8%	
Sweden	Taylorist	9.3 % _a	6.2 % _a	7.3 % _a	7.9%	
	Simple	14.7 % _a	6.2 % _b	6.4 % _b	10.2 %	

Notes: Subscripts denote whether difference between values across columns (from different EWCS waves) are statistically significant at the 0.05 level. In particular, values having the same subscripts do not differ significantly while values with different subscripts are significantly different.

Table A4.3: Trend developments across Member States — decreasing learning, but increasing Taylorist organisation						
. .			EWCS survey wave			
Country		2000	2005	2010	Iotal	
	Learning	38.0% _a	43.4% _a	30.6 % _b	35.7%	
France	Lean	31.4% _a	26.0% _{a, b}	24.8 % _b	27.2%	
France	Taylorist	16.8% _a	20.9% _{a, b}	23.5 % _b	20.8%	
	Simple	13.8% _a	9.6 % _a	21.1% _b	16.3%	
	Learning	23.3% _a	25.7% _a	23.4% _a	24.1%	
C	Lean	20.7% _a	30.1 % _b	21.1% _{a,b}	23.7%	
Greece	Taylorist	20.7% _a	21.9% _a	28.6 % _a	23.4%	
	Simple	35.3% _a	22.4% _b	26.9% _{a, b}	28.8%	
	Learning	41.4% _a	44.2 % _a	32.8 % _b	39.4%	
Uummenni	Lean	13.3% _a	17.2 % _{a, b}	22.0% _b	17.5%	
Hungary	Taylorist	23.3% _a	18.5 % _a	31.8% _b	24.6%	
	Simple	22.0% _a	20.1% _a	13.4% _b	18.5%	
	Learning	23.2% _a	25.5% _a	11.9% _b	20.6%	
Dulassia	Lean	25.6% _a	28.7 % _a	31.0% _a	28.5%	
Bulgaria	Taylorist	22.3% _a	25.9% _a	27.4% _a	25.3%	
	Simple	28.9% _a	19.9% _b	29.6 % _a	25.6%	
	Learning	39.3% _a	30.4 % _b	28.6% _b	33.1%	
Carala Davidalia	Lean	26.2% _a	28.3 % _a	27.8% _a	27.4%	
Czech Republic	Taylorist	19.9% _a	23.5 % _a	23.1% _a	22.0%	
	Simple	14.6%	17.8%	20.5%	17.5%	

Notes: Subscripts denote whether difference between values across columns (from different EWCS waves) are statistically significant at the 0.05 level. In particular, values having the same subscripts do not differ significantly while values with different subscripts are significantly different.

Table A4.4: Trend developments across M	ember States — No substantial	changes between 2000 and 2010

Country			Tetel		
Country		2000	2005	2010	Iotai
	Learning	24.2% _a	32.8 % _b	28.7 % _{a, b}	28.8%
Clavakia	Lean	31.2% _a	25.4% _a	26.9% _a	27.7%
Slovakia	Taylorist	28.1% _a	25.8% _a	25.1% _a	26.3%
	Simple	16.5% _a	16.1% _a	19.3% _a	17.3%
	Learning	25.6% _a	26.9% _a	27.2% _a	26.4%
Capin	Lean	28.6% _a	24.6% _a	31.9% _a	28.6%
Spain	Taylorist	28.3% _a	22.9% _a	21.1% _a	24.7%
	Simple	17.5% _a	25.7% _b	19.7 % _{a, b}	20.3%
	Learning	25.9% _a	29.7% _a	27.3% _a	27.4%
United Kingdam	Lean	40.9% _a	34.1% _a	37.8 % _a	38.0%
onitea kingaom	Taylorist	19.0% _a	19.4% _a	20.5 % _a	19.6%
	Simple	14.2% _a	16.9% _a	14.5 % _a	15.0%

Source: Eurofound estimates based on EWCS.

Notes: Subscripts denote whether difference between values across columns (from different EWCS waves) are statistically significant at the 0.05 level. In particular, values having the same subscripts do not differ significantly while values with different subscripts are significantly different.

gif

gif

Table A4.5: Trend developments across sectors						
			EWCS survey wave			
		2000	2005	2010	Total	
	Learning	47.8% _a	62.4% _b	55.0% _{a, b}	55.9%	
Electricity, gas, and	Lean	32.7 % _a	23.5 % _b	28.7% _{a, b}	27.8%	
water supply	Taylorist	3.4% _a	5.7 % _a	5.9% _a	5.2%	
	Simple	16.1% _a	8.4 % _b	10.4% _{a, b}	11.1%	
	Learning	54.9% _a	66.4% _b	59.1 % _a	60.0%	
Financial	Lean	21.5%	18.8 % _a	28.0% _b	23.1%	
intermediation	Taylorist	6.6% _a	3.1 % _b	3.9% _b	4.5%	
	Simple	16.9% _a	11.8% _b	9.0% _b	12.3%	
	Learning	36.9% _a	42.1% _b	33.8 % _a	37.1%	
Transport, storage	Lean	21.7%	19.9 % _a	23.1% _a	21.8%	
and communication	Taylorist	13.7 % _a	17.1% _b	16.2 % _{a, b}	15.6%	
	Simple	27.7% _a	20.9% _b	26.8% _a	25.5%	
	Learning	34.4% _a	37.4% _a	30.9 % _a	33.7%	
Hotels and	Lean	25.6% _a	17.9% _b	21.9% _{a, b}	21.9%	
restaurants	Taylorist	12.8% _a	24.0% _b	22.1% _b	19.8%	
	Simple	27.3% _a	20.8% _a	25.0% _a	24.5%	
Wholesale and retail	Learning	47.9% _a	45.5% _a	37.8 % _b	43.3%	
trade; repair of	Lean	17.2% _a	21.3% _b	22.4% _b	20.4%	
motor vehicles and	Taylorist	10.1% _a	10.1% _a	14.7% _b	11.9%	
motorcycles	Simple	24.8% _a	23.0% _a	25.1% _a	24.4%	
	Learning	43.0% _a	32.6% _b	36.6% _b	37.7%	
Construction	Lean	31.3% _a	37.2% _b	33.3% _{a, b}	33.7%	
Construction	Taylorist	13.0% _a	19.9% _b	16.2% _c	16.1%	
	Simple	12.7% _{a,b}	10.4% _b	13.9% _a	12.5%	
	Learning	33.7% _a	32.8% _a	32.9% _a	33.2%	
Mining, quarrying,	Lean	28.6% _a	32.1% _b	33.5% _b	31.1%	
Manufacturing	Taylorist	26.8% _a	26.4% _{a, b}	24.7% _b	26.0%	
	Simple	10.9% _a	8.7 % _b	8.9 % _b	9.7%	

Source: Eurofound estimates based on EWCS, Nace rev1.

Notes: Subscripts denote whether difference between values across columns (from different EWCS waves) are statistically significant at the 0.05 level. In particular, values having the same subscripts do not differ significantly while values with different subscripts are significantly different.

gif

Table A4.6: Trend developments across occupations						
EWCS survey wave						
Occupation: 1 st -l	evel ISCO codes		2000	2005	2010	Total
		Learning	68.5 % _a	55.6% _b	54.3 % _b	59.1%
Legislators,	High-skilled	Lean	26.5 % _a	36.3% _b	36.0% _b	33.1%
and managers	clerical	Taylorist	2.3 % _a	3.4% _{a,b}	4.4 % _b	3.5%
and managers		Simple	2.7 % _a	4.7 % _{a, b}	5.4 % _b	4.3%
		Learning	72.5 % _a	61.8% _b	68.4% _a	66.7%
	High-skilled	Lean	17.0% _a	32.1% _b	26.0% _c	26.0%
Professionals	clerical	Taylorist	7.1 % _a	2.9% _b	2.2 % _b	3.9%
		Simple	3.4 % _a	3.2 % _a	3.4% _a	3.3%
		Learning	58.9% _a	57.1% _a	60.8% _a	59.3%
Technicians	Low-skilled	Lean	23.9% _a	26.5 % _a	26.1% _a	25.4%
and associate	clerical	Taylorist	8.0% _a	10.0% _a	3.9% _b	6.8%
professionals		Simple	9.2 % _a	6.4 % _b	9.2 % _a	8.5%
		Learning	46.5 % _a	50.3 % _a	41.9% _b	46.0%
Clasha	Low-skilled	Lean	19.5 % _a	18.3 % _a	22.8% _b	20.3%
Clerks	clerical	Taylorist	8.6 % _a	9.6 % _a	13.3% _b	10.6%
		Simple	25.4% _a	21.8% _b	22.0% _b	23.1%
Service workers		Learning	35.7% _a	43.5% _b	26.4% _c	34.6%
and shop and	Low-skilled	Lean	18.1% _a	14.7% _a	23.9% _b	19.2%
market sales	clerical	Taylorist	12.0% _a	10.4% _a	19.0% _b	14.1%
workers		Simple	34.3 % _a	31.4% _a	30.7 % _a	32.1%
		Learning	33 .1% _a	31.2% _{a, b}	30.2 % _b	31.6%
Craft and related	High-skilled	Lean	34.1 % _a	36.3 % _{a, b}	38.1% _b	36.0%
trades workers	manual	Taylorist	22.4% _a	25.6% _b	23.0% _{a, b}	23.5%
		Simple	10.4% _a	6.8% _b	8.7 % _a	8.9%
Plant and		Learning	21.3% _a	17.3% _b	18.9% _{a, b}	19.4%
machine	Low-skilled	Lean	29.2 % _a	25.8% _b	27.2% _{a, b}	27.7%
operators and	manual	Taylorist	32.7 % _a	39.8% _b	31.2% _a	33.9%
assemblers		Simple	16.8% _a	17.0% _a	22.7% _b	19.0%
		Learning	19.6 % _a	26.4% _b	18.2 % _a	21.4%
Elementary	Low-skilled	Lean	18.9% _a	23.3% _b	23.5% _b	21.9%
occupations	manual	Taylorist	31.7% _a	29.4% _a	33.9% _a	31.6%
		Simple	29.8%	21.0% _b	24.4% _b	25.1%

Notes: Subscripts denote whether difference between values across columns (from different EWCS waves) are statistically significant at the 0.05 level. In particular, values having the same subscripts do not differ significantly while values with different subscripts are significantly different.

Table A4.7: Trend developments across firm size						
Number of paid			EWCS survey wave			
establishment		2000	2005	2010	Iotal	
	Learning	41.5% _a	39.5% _{a, b}	38.3 % _b	39.8%	
10,40	Lean	23.0% _a	24.1% _{a,b}	25.8% _b	24.4%	
10-49	Taylorist	15.3% _a	17.7 % _b	16.1% _{a,b}	16.3%	
	Simple	20.2% _a	18.6 % _a	19.8% _a	19.6%	
	Learning	36.8% _a	40.1 % _b	34.6% _c	37.0%	
50,400	Lean	27.1% _a	28.7% _{a, b}	29.9% _b	28.6%	
50-499	Taylorist	21.0% _a	19.8% _a	20.4% _a	20.4%	
	Simple	15.1% _a	11.3 % _b	15.2% _a	14.0%	
	Learning	38.4% _a	41.3% _a	39.1% _a	39.4%	
500	Lean	28.6% _a	31.2% _{a,b}	34.7 % _b	31.1%	
500 01 0001	Taylorist	20.8% _a	19.0% _a	19.1% _a	19.8%	
	Simple	12.2% _a	8.5 % _b	7.1 % _b	9.7 %	

gif

Note: Subscripts denote whether difference between values across columns (from different EWCS waves) are statistically significant at the 0.05 level. In particular, values having the same subscripts do not differ significantly while values with different subscripts are significantly different.

Table A4.8: Trend developments across different levels of seniority						
Number of		EWCS survey wave				
years working at the company		2000	2005	2010	Total	
1	Learning	35.0% _a	35.3 % _a	27.8% _b	32.9%	
	Lean	23.0% _a	24.9% _{a,b}	26.5% _b	24.6%	
I year or less	Taylorist	21.7% _a	23.0% _a	24.4% _a	22.9%	
	Simple	20.3% _a	16.7% _b	21.3% _a	19.6%	
	Learning	38.6% _a	37.5% _a	36.6% _a	37.5%	
2–10 years	Lean	25.9% _a	28.4% _b	28.4% _b	27.6%	
	Taylorist	17.8% _a	18.9% _a	17.8% _a	18.1%	
	Simple	17.7% _a	15.2% _b	17.2% _a	16.8%	
More than 10 years	Learning	41.6% _a	46.3% _b	40.8% _a	42.6%	
	Lean	26.8% _a	26.9% _a	29.9% _b	27.9%	
	Taylorist	17.9% _a	16.4% _a	16.6% _a	17.1%	
	Simple	13.6% _a	10.4% _b	12.7% _a	12.4%	

Source: Eurofound estimates based on EWCS.

Notes: Subscripts denote whether difference between values across columns (from different EWCS waves) are statistically significant at the 0.05 level. In particular, values having the same subscripts do not differ significantly while values with different subscripts are significantly different.

gif

Table A4.9: Trend developments across type of training					
Training		EWCS survey wave			Tatal
rraining		2000	2005	2010	Iotai
Training paid for or provided by your employer	Learning	50.4% _a	49.7% _a	44.2% _b	47.7%
	Lean	30.9% _a	33.1% _{a, b}	34.6% _b	32.9%
	Taylorist	9.6 % _a	10.3% _{a, b}	11.6% _b	10.6%
	Simple	9.1 % _a	6.8% _b	9.6 % _a	8.8%
	Learning	33.0% _a	44.6 % _b	49.7% _b	45.7%
Training paid for by yourself	Lean	18.0% _a	30.8% _b	32.9% _b	30.2 %
	Taylorist	25.0% _a	14.2% _b	9.4% _b	13.2%
	Simple	24.0% _a	10.4% _b	7.9% _b	10.9%
On-the-job training	Learning	38.8% _a	45.2% _b	39.1% _a	41.3%
	Lean	26.7% _a	32.4% _b	35.8% _c	34.0%
	Taylorist	16.0% _a	14.9% _a	14.8% _a	14.9%
	Simple	18.5 % _a	7.5 % _b	10.3 % _c	9.8%

Notes: Subscripts denote whether difference between values across columns (from different EWCS waves) are statistically significant at the 0.05 level. In particular, values having the same subscripts do not differ significantly while values with different subscripts are significantly different.

Table A4.10: Second job profiles

	Second job		EWCS survey wave			Tatal
			2000	2005	2010	TOLAL
	Yes	Learning	43.2 % _a	33.7 % _b	36.5% _b	38.0%
		Lean	29.1% _a	28.1% _a	23.7% _a	26.7%
		Taylorist	15.4% _a	22.8% _b	22.6% _b	20.2%
		Simple	12.3 % _a	15.4% _{a,b}	17.3% _b	15.1%

Source: Eurofound estimates based on EWCS.

Notes: Subscripts denote whether difference between values across columns (from different EWCS waves) are statistically significant at the 0.05 level. In particular, values having the same subscripts do not differ significantly while values with different subscripts are significantly different

gif

gif

ANNEX 5: COMPANIES' WELL-BEING POLICIES — CASE STUDIES

Many companies have recognised the link between positive mental states of workers and productivity on the job and have implemented specific well-being policies. The motivation of the these policies assumes that the benefits of having happier workers range from fewer interpersonal conflicts and less sick leave to stronger identification with the organisation or team and therefore more prevalent ethical behaviours, a better use of workers' skills for creative ideas and efficient problem-solving. Moreover, employers who have a reputation for caring for their people manage to attract and keep talent. These companies also manage to protect their quality workers from unnecessary stress, fatigue and frustration, assuring greater loyalty of workers and greater internalisation of corporate norms of behaviours.

AstraZeneca

AstraZeneca plc is a British-Swedish multinational pharmaceutical and biologics company headquartered in London.

The company launched a whole package of health and well-being initiatives ranging from a counselling and life-management programme, health promotion activities and ergonomic workspace design to fitness opportunities, healthy eating options and flexibility arrangements for a better work-life balance. The company reported savings in the range of GBP 500000-700000 through improved productivity after counselling. GBP 80000 was saved on health insurance costs for psychological illness. Global accident and occupational illness rates went down by 61%. The programme has also served company's image among its staff well. 84% of employees are proud to work for AstraZeneca and 82% would recommend the company as a good place to work, 80% of employees said they had enough flexibility in their job to be able to balance work and personal life, and 88% said AstraZeneca demonstrated commitment to the health and well-being of its employees.

British Gas Services

British Gas Services, Britain's largest energy and home services provider, needed to reduce the level of musculoskeletal disorders (MSDs), which accounted for one third of staff absences, to improve attendance and performance capability at work. To that effect, backcare workshops were introduced in 2005. 120 workshops were delivered over a two-year period with over 1200 participants. Back-related absence was reduced by 43% in the 2005 cohort one year after the seminar participation. 73% of the staff in the intervention group had no absence up to one year after participation. The company reported a solid return on investment: GBP 1660 per participating employee and GBP 31 on every pound invested.

The British Library

The British Library, a renowned research library based in London, developed a corporate well-being vision including personal development, diversity and a platform for dialogue and opinion survey to promote holistic health of employees. The employer guarantees free access to an employee assistance programme. This confidential service, run by an external contractor, offers support and advice on financial, legal and psychological issues for staff and their spouses, live-in partners and dependent children aged 16 to 23. Further, employees benefit from subsidised membership in gyms and discounted Tai Chi and yoga classes, osteopathy treatments and Shiatsu massages. The employees benefit from healthy on-site catering and nutritional guidance. The employer organises annual health events where employees can receive on-site lifestyle and health guidance and assessments, such as blood pressure and cholesterol tests, bone density scans and liver function tests. The Library tries to help staff and their families with health care costs. It facilitates access to medical diagnostic, surgical and medical support services via cheap flat-rate membership in the Beneden Healthcare Society and offers discounts with the HealthShield healthcare scheme. Staff also receive a 45% discount for travel healthcare insurance from BUPA.

The Library reported numerous business benefits of the well-being scheme and reports that over a two-year period absence dropped from 10.2 to 7.5 days per year, cost of absence dropped 11% (GBP 160000 per year), staff turnover was halved from 12% to 6% and performance management results increased from 86% to 98%.

Digital Outlook Communications

Digital Outlook Communications is a London-based digital marketing and creative agency specialising in the entertainment and media sectors. The company sought to address the challenge of ensuring the intense, long hours culture of its industry did not become a barrier to building the business on a foundation of sound health and well-being principles.

The company conducted a Best Companies survey to obtain employees' feedback on their well-being and the perceived quality of leadership and management. A Well-being Team, supported by senior management, was established to gather suggestions for, and implement, initiatives which included:

Introduction of flexible working; Revamping the agency's charging system to ensure clients paid for work actually done, optimise profitability and enable employees to reduce working hours while still meeting financial targets; Improved promotion of the employee benefits system; Introduction of a mentoring and development scheme; Improving the ergonomic working environment; Establishing health and well-being as a KPI for all senior managers.

Health and well-being survey scores improved 11% to a score of 4.9, better than all other small media companies surveyed in 2008. Sickness absence rates improved 95% from 4 days per person in 2006 to 0.22 days per person in 2008. Staff turnover was reduced from 34% in 2007 to 9% in 2008, resulting in savings in recruitment, training and induction costs.

Google

Inspired by the Framingham Heart Study, Google developed a long-term study called gDNA. The aim of the study is to learn how to improve well-being, cultivate great leaders, better understand how happiness affects work, and how work affects happiness (Bock, 2014). One issue that became a matter of corporate policy at Google pertains to managing the work-life balance and protecting the privacy of employees after work. Google's Dublin office, for example, ran a programme called 'Dublin Goes Dark' which asked people to drop off their devices at the front desk before going home for the night. Googlers reported blissful, stressless evenings.

Box 7: Office design integrating work, team space, privacy, entertainment and relaxation

The following is an example of how a contemporary multinational may try to recreate the informal start-up working environment that, in the opinion of its proponents, unleashes creativity and, according to its adversaries, blurs the line between working and private lives. Google's Zurich office is a very special case that became famous for its innovative design (¹²⁹) taking a radical step away from the norm. The design combines teamwork, privacy and individual work, entertainment, meditation and relaxation.

The (partially) open-plan office space is dotted with egg-shaped wigwams or arctic domes that serve as small meeting rooms. Some meeting rooms feature reclining chairs and sofas. Some people work with laptops while sitting in hammock-like facilities in tropical island-themed rooms. Some offices have a beach theme with sand, pebbles and lifebuoys. Some conference call rooms have a thematic design, e.g. ottoman-style sofas with a baldachin and other accessories. Some are styled as ski lifts or taxis, feature alpine designs or urban graffiti. The library is styled as a Victorian English parlour.

There is a quiet room where people go to relax or take a nap that features reclining chairs and a bathtub filled with foam in front of a fish tank. There are massage rooms. Google offers free breakfast, lunch and dinner all cooked by an in-house chef. There is a slide that drops employees into the eating area (a fun way to get to lunch). There are also poles allowing workers to drop down a floor. There are work-out spaces in the offices, as well as games: billiards, table football, ping pong, a basketball corner and a music stage.

How far this is replicated across the company or encouraged among its associates and suppliers is less clear, however. Also, time will show if this concept, in its current innovative yet very unusual form, will set a new trend in office design or remain an amusing yet unsuccessful path of corporate culture evolution.

(¹²⁹) See http://www.businessinsider.com/googles-zurich-office-2013-2?op=1

REFERENCES

Acemoglu, D., 'Good jobs versus bad jobs', *Journal of Labor Economics*, Vol. 19, No 1, 2001.

Acemoglu, D. and Autor, D. H., 'Skills, tasks and technology: Implications for employment and earnings', *NBER Working Paper No 16082*, The National Bureau of Economic Research, Cambridge, MA, 2010.

Adler, D. A., McLaughlin, T. J., Rogers, W. H., Chang, H., Lapitsky, L., et al., 'Job performance deficits due to depression', *American Journal of Psychiatry*, Vol. 163, 2006, pp. 1569–76.

Aghion, P., Caroli, E. and García-Peñalosa, C., 'Inequality and Economic Growth: The Perspective of the New Growth Theories', *Journal of Economic Literature*, Vol. XXXVII, 1999, pp. 1615–60, available at http://www.aeaweb.org/articles. php?doi=10.1257/jel.37.4.1615

Akerlof, G. A. and Yellen, J. L., 'The Fair Wage-Effort Hypothesis and Unemployment', *The Quarterly Journal of Economics*, Vol. 105, No 2, 1990, pp. 255–83.

Alesina, A. and Perotti, R., 'The Political Economy of Growth: A Critical Survey of the Recent Literature', *World Bank Economic Review*, Vol. 8, No 3, 1994, pp. 351–71, available at http://wwwwds.worldbank.org/servlet/WDSContentServer/IW3P/IB/1999/09/11/00017 8830_98101911401317/Rendered/PDF/ multi_page.pdf

Alesina, A. and Rodrik, D., 'Distributive politics and economic growth', *The Quarterly Journal of Economics*, Vol. 109, No 2, 1994, pp. 465–90.

Antonsson, A.-B., 'OSH in the green economy — a victim or an integrated aim?', *HesaMag No 9 Waste and recycling: workers at risk*, 2014, available at http:// www.etui.org/Topics/Health-Safety/News/ HesaMag-09-Waste-and-recycling-workers-at-risk

Appelbaum, E., Bailey, T., Berg, P. and Kalleberg, A., *Manufacturing Advantage — Why High-Performance Work Systems Pay Off*, ILR Press, Ithaca, NY, 2000.

Appelbaum, E. and Batt, R., *High Performance Work Systems: American Models* *of Workplace Transformation*, Economic Policy Institute, Washington, DC, 1993.

Appelbaum, E. and Batt, R., *The New American Workplace*, ILR Press, Ithaca, NY, 1994.

Appelbaum, E., Hoffer Gittell, J. and Leana, C., 'High-Performance Work Practices and Sustainable Economic Growth', 2011, available at http://50.87.169.168/ Documents/EPRN/High-Performance-Work-Practices-and-Sustainable-Economic-Growth.pdf

Association of Colleges, Why Careers Guidance: Guaranteed is crucial, a report from the Association of Colleges membership survey of careers advice and guidance, 2014, available at http:// www.aoc.co.uk/system/files/Why%20 Careers%20Guidance%20Guaranteed%20is%20crucial.pdf

Auer, P., Berg, J., and Coulibaly, I., 'Is a stable workforce good for productivity?', *International Labour Review*, Vol. 144, No 3, 2005, available at http://onlinelibrary.wiley.com/doi/10.1111/j.1564-913X.2005.tb00571.x/abstract

Autor, D., 'Polanyi's Paradox and the Shape of Employment Growth', Federal Reserve Bank of Kansas City's economic policy symposium on Re-Evaluating Labor Market Dynamics, 2014, available at http://www.kc.frb.org/publicat/ sympos/2014/2014093014.pdf

Autor, D. and Dorn, D., 'The Growth of Low Skill Service Jobs and the Polarization of the U.S. Labor Market', *American Economic Review*, Vol. 103, No 5, 2013, pp. 1553–97, available at http://www. aeaweb.org/articles.php?doi=10.1257/ aer.103.5.1553

Autor, D., Levy, F. and Murnane, R., 'The Skill Content of Recent Technological Change: An Empirical Exploration', *The Quarterly Journal of Economics*, Vol. 118, No 4, 2003, pp. 1279–333.

Bachmann, R., König, M. and Schaffner, S., 'Lost in Transition? Minimum Wage Effects on German Construction Workers', *Ruhr Economic Papers* No 358, 2012, available at http://www.rwi-essen. de/media/content/pages/publikationen/ ruhr-economic-papers/REP_12_358.pdf

Baron, J., Hannan, M. and Burton, D., 'Labor pains: Change in organizational models and employee turnover in young, high-tech firms', *American Journal of Sociology*, Vol. 106, No 4, 2001, pp. 960–1012.

Beniger, J. R., *The control revolution: Technological and economic origins of the information society*, Harvard University Press, 1986.

Bengt-Åke Lundvall, 'Organising for learning at work - an important but neglected dimension of Innovations systems', Keynote speech at The International Helix Conference 12 June 2013 Linköping Aalborg University, available at http:// www.liu.se/helix/helix-conference-2013/ conference-information?l=en

Biagi, F., Cavapozzi, D. and Miniaci, R., 'Employment transitions and computer use of older workers', *Applied Economics*, Vol. 45, No 6, 2013, pp. 687–96, available at http://peer.ccsd.cnrs.fr/ peer-00741504

Black, D., 'Discrimination in an equilibrium search model', *Journal of Labor Economics*, Vol. 13, 1995, pp. 309–34, available at http://www.jstor.org/discover /10.2307/2535106?uid=3737592&uid= 2&uid=4&sid=21104204622977

Black, S. and Lynch, L., 'How to compete: the impact of workplace practices and information technology on productivity', *Review of Economics and Statistics*, Vol. 83, 2001, pp. 434–45.

Blanco, M. I. and Rodrigues, G., 'Direct Employment in the Wind Energy Sector: An EU Study', *Energy Policy*, Vol. 37, 2009, pp. 2847–57, available at http://ac.elscdn.com/S0301421509001359/1s2.0-S0301421509001359-main. pdf?_tid=38c7d83a-b3f9-11e3-9fc6-00000aab0f6c&acdnat=1395737053_ 3ad85263fef4584018f5ac57e1626c05

Bloss, R., 'Mobile hospital robots cure numerous logistic needs', *Industrial Robot: An International Journal*, Vol. 38, No 6, 2011, pp. 567–71.

Bock, L., 'Google's Scientific Approach to Work-Life Balance (and Much More)', *Harvard Business Review*, 27 March 2014.

Böckerman, P. and Ilmakunnas, P. (2012). The Job Satisfaction-productivity Nexus: A Study Using Matched Survey and Register Data. Industrial and Labor Relations Review, 65:2, 244–262. Bonsdorff, von, M., Huuhtanen, P., Tuomi, K. and Seitsamo, J., 'Predictors of employees' early retirement intentions: an 11-year longitudinal study', *Occupational Medicine*, Vol. 60, 2010, pp. 94–100, available at http://occmed. oxfordjournals.org/content/60/2/94.full. pdf+html

Brown, S., McHardy, J., McNabb, R. and Taylor, K., 'Workplace Performance, Worker Commitment and Loyalty', *IZA Discussion Paper* No 5447, 2011, available at http://ftp.iza.org/dp5447.pdf

Brynjolfsson, E. and McAfee, A., *Race* against the machine: How the digital revolution is accelerating innovation, driving productivity, and irreversibly transforming employment and the economy, Digital Frontier Press, Lexington, MA, 2011.

Brynjolfsson, E. and McAfee, A., *The Second Machine Age*, W. W. Norton & Company, 2014.

Bustillo, R., Fernandez-Macias, E., Anton, J. I. and Esteve, F., 'On Job Quality. The Economic point of view', Paper presented at the International Expert Conference on Job Quality, Copenhagen, 10–11 July, 2012.

Caponi, V., Kayahan, B. and Plesca, M., 'The impact of aggregate and sectoral fluctuations on training decisions', *The B.E. Journal of Macroeconomics*, Vol. 10, No 1, 2010, pp. 1–35.

Cappelli, P. and Neumark, D., 'Do 'High-Performance' work practices improve establishment-level outcomes?', *Industrial and Labor Relations Review*, Vol. 54, 2001, pp. 737–75.

Card, D., Mas, A., Moretti, E. and Saez, E., 'Inequality at Work: The Effect of Peer Salaries on Job Satisfaction', *American Economic Review*, Vol. 102, No 6, 2012, pp. 2981–3003.

CEDEFOP, 'Skills for green jobs. European synthesis report', 2010, available at http://www.cedefop.europa.eu/EN/publications/16439.aspx

Charness, G., Masclet, D. and Villeval, M. C., 'The dark side of competition for status', *Management Science*, Vol. 60, No 1, 2013, pp. 38–55.

Chirkov, V. I., Ryan, R. and Sheldon, K. M. (eds.), *Human Autonomy in Cross-Cultural*

Context. Perspectives on the Psychology of Agency, Freedom, and Well-Being, Springer Publishing, 2011.

Christen, R. N., Alder, J. and Bitzer, J., 'Gender differences in physicians' communicative skills and their influence on patient satisfaction in gynaecological outpatient consultations', *Social Science & Medicine*, Vol. 66, No 7, 2008, pp. 1474–83.

Clark, A. E. and Oswald, A. J., 'Satisfaction and comparison income', *Journal of Public Economics*, Vol. 61, No 3, 1996, pp. 359–81.

Combs, J., Liu, Y., Hall, A. and Ketchen, D., 'How Much Do High-Performance Work Practices Matter? a Meta-Analysis of Their Effects on Organizational Performance', *Personnel Psychology*, Vol. 59, No 3, 2006, pp. 501–28, available at http://doi.wiley.com/10.1111/j.1744-6570.2006.00045.x

Cooke, W., 'Employee participation programs, group-based incentives, and company performance. A union-nonunion comparison', *Industrial and Labor Relations Review*, Vol. 47, 1994, pp. 594–609.

Cortada, J. W., *Before the Computer: IBM, NCR, Burroughs, and Remington Rand and the Industry They Created, 1865– 1956*, Princeton University Press, 2000.

Cottini, E. and Lucifora, C., 'Mental health and working conditions in Europe' (conference presentation), *Trends in Job Quality*, Cornell University, November 2011.

Dearden, L., Reed, H. and Van Reenen, J., 'The impact of training on productivity and wages: Evidence from British panel data', *Oxford Bulletin of Economics and Statistics*, Vol. 68, No 4, 2006, pp. 397–421.

Deaton, A. and Kahneman, D., 'High income improves evaluation of life but not emotional well-being', *Proceedings of the National Academy of Sciences*, Vol. 107, No 38, 2010.

DeCarlo, T. and Agarwal, S., 'Influence of Managerial Behaviors and Job Autonomy on Job Satisfaction of Industrial Salespersons', *Industrial Marketing Management*. Vol. 28, 1999, pp. 51–62.

Dedrick, J., Kraemer, K. L. and Linden, G., 'Who Profits from Innovation in Global Value Chains? A Study of the iPod and notebook PCs', Alfred P. Sloan Foundation, *Industry Studies*, 2008, available at http://web.mit.edu/is08/pdf/Dedrick_ Kraemer_Linden.pdf

Demyttenaere, K., Bruffaerts, R., Posada-Villa, J., Gasquet, I., Kovess, V. et al.,(WHO World Mental Health Survey Consortium), 'Prevalence, severity, and unmet need for treatment of mental disorders in the World Health Organization World Mental Health Surveys', *JAMA* Vol. 291, 2004, pp. 2581–90.

Dennison R., *Northern Ireland Childcare Cost survey*, 2013.

Dhondt, S., Kraan, K. and van Sloten, G., 'Work organisation, technology and working conditions', 2002, available at www. eurofound.europa.eu/pubdocs/2002/05/ en/1/ef0205en.pdf

Dieckhoff, M. (2013), 'Continuing Training in Times of Economic Crisis', Chapter 3 in Gallie (2013).

Driscoll, T., Takala, J., Steenland, K., Corvalan, C. and Fingerhut, M., 'Review of estimates of the global burden of injury and illness due to occupational exposures', *American Journal of Industrial Medicine*, Vol. 48, No 6, 2005, pp. 491–502.

Easterlin, R. A., 'Does economic growth improve the human lot? Some empirical evidence', in David, P. A. and Reder, M. W. (eds.), *Nations and households in economic growth: Essays in honour of Moses Abramovitz*, Academic Press, New York, NY, 1974.

Eaton, W. W., Martins, S. S., Nestadt, G., Bienvenu, O. J., Clarke, D., et al., 'The burden of mental disorders', *Epidemiological Review*, Vol. 30, 2008, pp. 1–14.

Edwards, R., *Contested terrain. The transformation of the workplace in the 20*th *century*, Basic Books, New York, NY, 1979.

Emery, F., and Trist, E. L., 'Socio-technical systems', in Churchman, C. W. and Verhulst, M. (Eds.), *Management Science. Models and Techniques*, Vol. 2, Pergamon Press, London, 1960.

Erez, A. and Isen, A. M., 'The influence of positive affect on the components of expectancy motivation', *Journal of Applied Psychology*, Vol. 89, 2012, pp. 1055–67. Etzioni, A., *Modern Organizations*, Prentice Hall, Englewood Cliffs, NJ, 1964.

Eurofound, 'Quality of work and employment in Europe: Issues and challenges', Foundation Paper No 1, Office for Official Publications of the European Communities, Luxembourg, 2002.

Eurofound, 'Work organisation and innovation', 2010a, available at http://www. eurofound.europa.eu/pubdocs/2012/72/ en/1/EF1272EN.pdf

Eurofound, 'Posted workers in the European Union', 2010b, available at http:// www.eurofound.europa.eu/docs/eiro/ tn0908038s/tn0908038s.pdf

Eurofound, 'Work organisation and innovation', 2012a, available at http://www. eurofound.europa.eu/publications/htmlfiles/ef1272.htm

Eurofound, 'Trends in Job Quality in Europe', 2012b, available at http://www. eurofound.europa.eu/publications/htmlfiles/ef1228.htm

Eurofound, 'After restructuring: Labour markets, working conditions and life satisfaction', ERM report 2012, 2012c, available at http://www.eurofound.europa.eu/ pubdocs/2012/61/en/1/EF1261EN.pdf

Eurofound, 'Fifth European working conditions survey : overview report', 2012d, available at http://www.eurofound.europa.eu/publications/htmlfiles/ ef1182.htm

Eurofound, 'Employment polarisation and job quality in the crisis: European Jobs Monitor 2013', 2013a, available at http://www.eurofound.europa.eu/pubdocs/2013/04/en/1/EF1304EN.pdf

Eurofound, 'Work organisation and employee involvement in Europe', 2013b, available at http://www.eurofound.europa. eu/pubdocs/2013/30/en/1/EF1330EN.pdf

Eurofound, 'Role of social dialogue in industrial policies', 2014, available at http://www.eurofound.europa.eu/eiro/ studies/tn1311011s/tn1311011s.htm

Eurofound, 'Changes to wage-setting mechanisms in the context of the crisis and the EU's new economic governance regime', 2014, available at http:// www.eurofound.europa.eu/eiro/studies/ tn1402049s/tn1402049s.htm Eurofound, 'Involving employees at the workplace pays off in higher levels of work performance', (s.a.), available at http://www.eurofound.europa.eu/areas/ workorganisation/innovation.htm

Eurofound, 'Workplace innovation', (s.a.), available at https://eurofound.europa.eu/ areas/industrialrelations/dictionary/definitions/workplaceinnovation.htm

European Agency for Safety and Health at Work (EU-OSHA), 'Foresight of new and emerging risks to occupational safety and health associated with new technologies in green jobs by 2020', 2011, available at https://osha.europa.eu/en/ publications/reports/foresight-greenjobs-drivers-change-TERO11001ENN

European Agency for Safety and Health at Work (EU-OSHA), 'Priorities for occupational safety and health research in Europe: 2013–20', 2013, available at https://osha.europa.eu/en/publications/ reports/priorities-for-occupationalsafety-and-health-research-in-europe-2013–2020

European Commission, 'Partnership for a new organisation of work', Green Paper, Bulletin of the European Union, Supplement 4/97, 1997, available at http://europa.eu/documents/comm/ green_papers/pdf/com97_128_en.pdf

European Commission, 'Modernising the organisation of work — A positive approach to change', COM (98) 592 final, 1998, available at http://ec.europa.eu/ social/BlobServlet?docId=2934&langId =en

European Commission, 'Ageing well in the Information Society. An i2010 Initiative Action Plan on Information and Communication Technologies and Ageing', COM(2007) 332, 2007a, available at http://ec.europa.eu/prelex/detail_dossier real.cfm?CL=en&DosId=195839

European Commission, 'Ageing well in the Information Society. An i2010 Initiative Action Plan on Information and Communication Technologies and Ageing', Commission Staff Working Document, SEC(2007) 811, 2007b, available at http://ec.europa.eu/transparency/regdoc/ rep/2/2007/EN/2-2007-811-EN-1-0.Pdf

European Commission, 'Raising productivity growth: key messages from the European Competitiveness Report 2007', 2007c, available at http:// eur-lex.europa.eu/LexUriServ/LexUriServ. do?uri=COM:2007:0666:FIN:EN:PDF

European Commission, 'Measuring the quality of employment in the EU', Chapter 4 in *Employment in Europe 2008*, 2008, available at http://ec.europa.eu/ social/main.jsp?langId=en&catId=89&n ewsId=415

European Commission, 'Strategy for equality between women and men', Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions, COM(2010) 491 final, 2010, available at http://eurlex.europa.eu/LexUriServ/LexUriServ. do?uri=COM:2010:0491:FIN:en:PDF

European Commission, 'New Skills for new jobs', 2010a, available at http:// ec.europa.eu/social/main.jsp?catId=568

European Commission, 'Variations and trends in European industrial relations in the 21st century's first decade', *Industrial Relations in Europe 2010*, 2010b, pp. 17–53, available at http://ec.europa.eu/ social/BlobServlet?docId=6607&langId=en

European Commission, Conference on Fundamental Social Rights and the Posting of Workers in the Framework of the Single Market, 2011, available at http:// ec.europa.eu/social/main.jsp?langId=en &catId=471&eventsId=347&furtherEv ents=yes

European Commission, 'Active ageing', Chapter 5 in *Employment and Social Developments in Europe*, 2011a, available at http://ec.europa.eu/social/BlobSe rvlet?docId=7294&langId=en

European Commission, 'Employment and Social Developments in Europe', 2012, available at http://ec.europa.eu/ social/main.jsp?catId=738&langId=en& pubId=7315

European Commission, 'Proposal for a Directive of the European Parliament and of The Council on the enforcement of Directive 96/71/EC concerning the posting of workers in the framework of the provision of services', COM(2012) 131 final, 2012a, available at http://www.europarl.europa.eu/meetdocs/2009_2014/ documents/com/com_com(2012)0131_/ com_com(2012)0131_en.pdf European Commission, 'A European strategy for Key Enabling Technologies — A bridge to growth and jobs', Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions, COM(2012) 341 final, 2012b, available at http:// eur-lex.europa.eu/LexUriServ/LexUriServ. do?uri=COM:2012:0341:FIN:EN:PDF

European Commission, 'Employment and social developments in Europe', 2013.

European Commission, 'Posting of workers: EU safeguards against social dumping', MEMO/13/1138 from 12/12/2013, 2013a, available at: http://europa.eu/rapid/ press-release_MEMO-13-1138_en.htm

European Commission, 'Free movement of EU citizens and their families: Five actions to make a difference', COM(2013)837 final, 2013b.

European Commission, 'LIFE creating green jobs and skills', 2013c, available at http://ec.europa.eu/environment/life/ publications/lifepublications/lifefocus/ documents/jobs_skills.pdf

European Commission, 'Overview of European, national and public sector industrial relations', *Industrial Relations in Europe 2012*, 2013d, pp. 21–52, available at http://ec.europa.eu/social/BlobSer vlet?docId=9994&langId=en

European Commission, 'Innovation Union Scoreboard 2014', 2014, available at http://ec.europa.eu/enterprise/policies/ innovation/files/ius/ius-2014_en.pdf

European Commission, 'State of the Industry, Sectoral overview and Implementation of the EU Industrial Policy', Commission Staff Working Document, SWD(2014) 14 final, 2014a, available at http://eur-lex.europa.eu/legal-content/ EN/TXT/PDF/?uri=CELEX:52014SC0014& from=EN

European Commission, 'For a European Industrial Renaissance', Communication from The Commission to the European Parliament, the Council, the Economic and Social Committee and the Committee of the Regions, 2014b, available at http://eur-lex.europa.eu/legal-content/ EN/TXT/PDF/?uri=CELEX:52014SC0014 &from=EN European Commission, 'Tackling the gender pay gap in the European Union', DG Justice, 2014c, available at http:// ec.europa.eu/justice/gender-equality/ files/gender_pay_gap/140227_gpg_brochure_web_en.pdf

European Commission, 'Social agenda', No 36, 2014d, available at: http:// ec.europa.eu/social/main.jsp?catId=737 &langId=en&pubId=7691

European Commission, 'Green Employment Initiative Communication', Communication from The Commission to the European Parliament, the Council, the Economic and Social Committee and the Committee of the Regions, 2014e, available http://eur-lex.europa.eu/legalcontent/EN/TXT/?qid=1400658727626 &uri=COM:2014:446:FIN

European Commission (2000), 'First Stage Consultation of social partners on modernising and improving employment relations', available at http://ec.europa. eu/social/BlobServlet?docId=2485&lan gld=en

European Commission (2001), 'Second stage consultation of social partners on modernising and improving employment relations', available at http://ec.europa. eu/social/BlobServlet?docId=2486&lan gld=en

European Commission (s.a.), 'Policies for Ageing Well with ICT', available at http://ec.europa.eu/digital-agenda/en/ policies-ageing-well-ict

European Commission, 'Dimensions of Quality in Work', s.a.1, available at http:// ec.europa.eu/social/BlobServlet?docId=2 134&langId=en

European Commission, 'Workplace innovation', s.a.2, http://ec.europa.eu/ enterprise/policies/innovation/policy/ workplace-innovation/index_en.htm

European Commission, 'Facing the Future — Modernisation of Work Organisation', s.a.3, available at http:// ec.europa.eu/social/main.jsp?catld=709 &langId=en&intPageId=216

European Institute for Construction Labour Research, 'Undeclared labour in the construction industry', Study commissioned by the European Social Partners in the construction sector, FIEC and EFBWW, financed by the European Commission, 2006.

European Parliament, 'Indicators of job quality in the European Union', DG for Internal Policies, Policy Department A: Economic and Scientific Policy, 2009, available at http://www.europarl.europa. eu/document/activities/cont/201107/20 110718ATT24284/20110718ATT2428 4EN.pdf

European Parliament, 'Role of women in the green economy', European Parliament resolution of 11 September 2012 on the role of women in the green economy (2012/2035(INI)), 2012, available at http://www.europarl.europa.eu/sides/get-Doc.do?pubRef=-//EP//NONSGML+TA+P7-TA-2012-0321+0+D0C+PDF+V0// EN

European Parliament, 'Indicators of Job Quality', Directorate-General for Internal Affairs, 2012a, available at http://www. europarl.europa.eu/document/activities/ cont/201107/20110718ATT24284/201 10718ATT24284EN.pdf

European Parliament, 'Report on the impact of the economic crisis on gender equality and women's rights', Committee on Women's Rights and Gender Equality, 2012/2301(INI), 2013, available at http://www.europarl.europa.eu/sides/ getDoc.do?type=REPORT&reference=A7-2013-0048&language=EN

European Workplace Innovation Network (EUWIN), at http://ec.europa.eu/ enterprise/policies/innovation/policy/ workplace-innovation/euwin/index_en.htm

Eurostat, 'Reconciliation between work, private and family life in the European Union', 2009, available at http:// epp.eurostat.ec.europa.eu/cache/ITY_ OFFPUB/KS-78-09-908/EN/KS-78-09-908-EN.PDF

Eurostat, 'Active ageing and solidarity between generations. A statistical portrait of the European Union', 2012, available at http://epp.eurostat.ec.europa. eu/portal/page/portal/product_details/ publication?p_product_code=KS-EP-11-001

Eurostat, 'Active Ageing', Special Eurobarometer 378, 2012, available at http:// ec.europa.eu/public_opinion/archives/ ebs/ebs_378_en.pdf Feldstead, A. and N. Jewson, N., 'The impact of the 2008–9 recession on the extent, form and patterns of training at work', *LLAKES research paper 22*, LLA-KES Centre, Institute of Education, London, 2011.

Fernández-Macías, E., 'Job Polarization in Europe? Changes in the Employment Structure and Job Quality, 1995–2007', *Work and Occupations*, Vol. 39, 2012, pp. 157–82, available at http://wox.sagepub.com/content/39/2/157.full.pdf+html

Fernie, S. and Metcalf, D., 'Participation, contingent pay, representation and workplace performance: evidence from Great Britain', *British Journal of Industrial Relations*, Vol. 33, 1995, pp. 379–415.

Finn, C. P., 'Autonomy: an important component for nurses' job satisfaction', *International Journal of Nursing Studies*, Vol. 38, 2001, pp. 349–57.

Frey, B. S. and Stutzer, A., *Happiness and economics: How the economy and institutions affect human well-being*, Princeton University Press, 2010.

Frey, C. B. and Osborne, M. A., *The Future* of *Employment: How Susceptible Are Jobs to Computerisation?*, 2013.

Gallie, D. (Ed.), *Economic Crisis, Quality of Work and Social Integration, The European Experience,* Oxford University Press, 2013.

Gallie, D. and Zhou, Y. (2013), 'Job control, work intensity, and work stress.' In Gallie (2013).

Gallup, 'State of the American Workplace — Employee Engagement Insights for U.S. Business Leaders', 2013, available at http://www.gallup.com/strategicconsulting/163007/state-american-workplace.aspx

Galor, O. and Zeira, J., 'Income distribution and macroeconomics', *Review of Economic Studies*, Vol. 60, No 1, 1993, pp. 35–52.

Gaušas, S., 'Greening of industries in the EU: Anticipating and managing the effects on quantity and quality of jobs', Eurofound, 2013, available at http:// www.eurofound.europa.eu/publications/ htmlfiles/ef1248.htm Gellatly, I. R. and Irving, G., 'Personality, Autonomy and Contextual Performance for Managers', *Human Performance*, Vol. 43, No 3, 2001, pp. 231–45.

Georgantzis, N. and Vasileiou, E., 'Are Dangerous Jobs Paid Better? European Evidence', *Research in Labor Economics*, Vol. 38, 2013, pp. 163–92.

Gill, D., Prowse, V. and Vlassopoulos, M., 'Cheating in the workplace: An experimental study of the impact of bonuses and productivity', *Journal of Economic Behavior & Organization*, Vol. 96, 2013, pp. 120–34.

Godard, J. and Delaney, J., 'Reflections on the "High Performance" paradigm's implications for industrial relations as a field', *Industrial and Labor Relations Review*, Vol. 53, 2000, pp. 482–502.

Godart, O., Görg, H. and Hanley, A., 'Trust-Based Work-Time and Product Improvements: Evidence from Firm Level Data', *IZA Discussion Paper 80/97*, Bonn, April 2014.

Goldthorpe, J. H. and Hope, K., *The Social Grading of Occupations*, Oxford University Press, Oxford, 1974.

Goos, M., Manning, A. and Salomons, A., 'Job polarization in Europe', American Economic Review: Papers and Proceedings, 2009, Vol. 99, No. 2, pp. 58–63, available at https://www.aeaweb.org/ articles.php?doi=10.1257/aer.99.2.58

Gordon, R. J., 'The Demise of U.S. Economic Growth: Restatement, Rebuttal, and Reflections', *NBER Working Paper No 19895*, Cambridge, MA, 2014, available at http://www.nber.org/papers/w19895

Green, A. E., de Hoyos, M., Barnes, S.-A., Owen, D., Baldauf, B. and Behle, H., 'Literature Review on Employability, Inclusion and ICT, Report 2: ICT and Employability', JRC Technical Reports Series, 2013, available at http://ftp.jrc.es/EURdoc/JRC78601.pdf

Green, F., *Demanding work. The paradox of job quality in the affluent economy*, Princeton University Press, Princeton, NJ, 2006.

Green, F., 'Unpacking the misery multiplier: how employability modifies the impacts of unemployment and job insecurity on life satisfaction and mental health', *Journal of Health Economics*, Vol. 30, No 2, 2011, pp. 265–76.

Greenan, N. and Mairesse, J., 'How do New Organizational Practices Change the Work of Blue Collars, Technicians and Supervisors? Results from Matched Employer–Employee Survey for French Manufacturing', Mimeo, INSEE-CREST, Paris, 2002.

Grimshaw, D., 'What do we know about lowwage work and low-wage workers? Analysing the definitions, patterns, causes and consequences in international perspective', *ILO Conditions of Work and Employment Series*, No 28, 2011, available at https:// research.mbs.ac.uk/european-employment/Portals/0/docs/gendersocial/ILO%20 Paper%202011%20-%20CWES28.pdf

Gringart, E., Helmes, E. and Speelman, C., *The Role of Stereotypes in Age Discrimination in Hiring: Evaluation and Intervention*, Lambert Academic Publishing, Saarbrucken, 2010, available at http:// eprints.jcu.edu.au/12160/8/12160_Gringart_et_al_2010_Front_pages.pdf

Grossman, G. and Rossi-Hansberg, E., 'Trading Tasks: A Simple Theory of Offshoring', *NBER Working Paper 12721*, 2012, available at http://www.nber.org/ papers/w12721

Guillen, A. and Dahl, S., *Quality of work in the European Union. Concept, data and debates from a transnational perspective*, P.I.E. Peter Land S.A., 2009.

Halko, M.-L., Sääksvuori, L. and Timko, K., *Stress, Competitiveness and Gender*, In preparation, 2014.

Hansen, J. A. and Andersen, S. K., 'Employment Relations Research Centre (FAOS), East European workers in the building and construction Industry — Recruitment strategies and effects on wages, employment terms and agreements', 2008, available at http://faos.ku.dk/english/pdf/ publications/2008/East_European_workers_in_the_building_and_construction_ industry_0108.pdf

Healthy People = Healthy Profits, 20 case studies from UK companies: A Business in the Community report, 2009, available at http://www.work4wellbeing. org/uploads/2/3/4/1/23413848/bitc_-_4647_healthy_profits.pdf Hempell, T. and Zwick, T., 'New Technology, Work Organisation, and Innovation', *Economics of Innovation and New Technology*, Vol. 17, No 4, 2008, available at http://www.tandfonline.com/doi/ abs/10.1080/10438590701279649#. U467qvl tbE

Heywood, J. S. and Siebert, W. S., 'Understanding the Labour Market for Older Workers: A Survey', Discussion Paper No 4033, 2009, available at http://ftp. iza.org/dp4033.pdf

High Level Group on Key Enabling Technologies (HLGKET), 'Thematic Report by the Working Team on Advanced Manufacturing Systems', 2010, available at http://ec.europa.eu/enterprise/sectors/ ict/files/kets/6_advanced_manufacturing_report_en.pdf

Hobson, A. and Pace-Schott, E. F., 'The cognitive neuroscience of sleep: neuronal systems, cosciouncesness and learning', *Nature Reviews Neuroscience 3*, (September) 2002, pp. 679–693.

Holmlund, B., 'What do labor market institutions do?', IZA DP No 7809, 2013, available at http://uu.diva-portal.org/smash/ get/diva2:668186/FULLTEXT01.pdf

Hübler, O. and Jirjahn, U., 'Arbeitsproduktivität, Reorganisationsmaßnahmen und Betriebsräte', in Bellmann, L. and Kölling, A. (eds.), *Betrieblicher Wandel und Fachkräftebedarf*, Beiträge zur Arbeitsmarkt- und Berufsforschung, Vol. 257, 2002, pp. 1–45.

Huselid, M. A., 'The impact of human resource management practices on turnover, productivity, and corporate financial performance', *Academy of Management Journal*, Vol. 38, 1995, pp. 635–72.

Huwart, J.-Y. and Verdier, L., 'Does Globalization promote employment?', in *Economic Globalization: Origins and consequences*, OECD Publishing, 2013, available at http://www.oecd-ilibrary. org/economics/economic-globalisation/does-globalisation-promoteemployment_9789264111905-7-en

Ichino, A. and Riphahn, R., 'The effect of employment protection on worker effort: Absenteeism during and after probation', *Journal of the European Economic Association*, Vol. 3, No 1, 2005, pp. 120–43, available at http://onlinelibrary.wiley.com/ doi/10.1162/1542476053295296/abstract Ichniowski, C., Kochan, T. A., Levine, D., Olson, C. and Strauss, G., 'What works at work: overview and assessment', *Industrial Relations*, Vol. 35, 1996, pp. 356–74.

Ichniowski, C., Shaw, K. and Prennushi, G., 'The effects of human resource management practices on productivity: a study of steel finishing lines', *American Economic Review*, Vol. 87, 1997, pp. 293–313.

Idea Consult and ECORYS, 'Study on the economic and social effects associated with the phenomenon of posting of workers in the EU', Report on behalf of DG EMPL, 2011, available at http://ec.europa.eu/social/BlobServ let?docId=6678&langId=en

International Labour Foundation for Sustainable Development, 'Green Jobs and Women Workers Employment, Equity, Equality', 2009, available at http://www.sustainlabour.org/IMG/pdf/ women.en.pdf

International Labour Organisation (ILO) , 'Green jobs: Improving the climate for gender equality too!', 2008, available at http://www.preventionweb.net/ files/8307_wcms1015051.pdf

International Labour Organisation (ILO), 'Sustainable development, decent work and green jobs', 2013a, available at http://www.ilo.org/ wcmsp5/groups/public/---ed_norm/--relconf/documents/meetingdocument/ wcms_207370.pdf

International Labour Organisation (ILO), 'Social Dimensions of Free Trade Agreements', 2013b, available at http://www. ilo.org/global/research/publications/ WCMS_228965/lang--en/index.htm

International Labour Organisation (ILO), 'Decent work agenda', s.a., available at http://www.ilo.org/global/aboutthe-ilo/decent-work-agenda/lang--en/ index.htm

International Labour Organisation (ILO) and Organisation for Economic Co-operation and Development (OECD), 'Sustainable development, green growth and quality of employment', Background paper for the meeting of the G2O Labour and Employment Ministers, Guadalajara, 17–18 May, 2012, available at http://www.oecd.org/els/ emp/50318559.pdf Inanc, H., *Labour Market Insecurity and Family Relationships*, 2010.

Jackall, R., *Moral Mazes*, Oxford University Press, New York, 1988.

Jackson, T., 'Let's Be Less Productive', *The New York Times*, 26 May 2012.

Jencks, C., Perman, L. and Rainwater, L., 'What Is a Good Job? A New Measure of Labour-Market Success', *American Journal of Sociology*, Vol. 93, No 6, 1988, pp. 1322–57.

Kahneman, D., *Thinking, fast and slow,* Macmillan, 2011.

Kalleberg, A. L. and Vaisey, S., 'Pathways to a good job: perceived work quality among machinists in North America', *British Journal of Industrial Relations*, Vol. 43, 2005, pp. 431–54.

Karasek, R. A. and Theorell, T., *Healthy work, stress, productivity and the reconstruction of work life,* Basic Books, New York, 1990.

Kärreman, D. and Alvesson, M., 'Cages in Tandem: Management Control, Social Identity, and Identification in a Knowledge-Intensive Firm', *Organization*, Vol. 11, 2004, p. 149.

Kelley, R., *The Gold Collar Worker — Har*nessing the Brainpower of the New Workforce, Addison-Wesley, Reading, MA, 1990.

Kim, G. and Lee, H.-R., 'Mothers' working hours and children's obesity: Data from the Korean national health and nutrition examination survey 2008–10', Annual Occupational & Environmental Medicine Symposium, Vol. 25, No 28, 2013.

Klaver, de P., van der Graaf, A., Grijpstra, D. and Snijders, J., 'More and better jobs in home-care services', 2013, available at http://www.eurofound.europa.eu/publications/htmlfiles/ef1353.htm

Knabe, A. and Plum, A., 'Low-wage Jobs — Springboard to High-paid Ones?', *Labour*, Vol. 27, No 3, 2013, pp. 310–30, available at http://onlinelibrary.wiley. com/doi/10.1111/labr.12015/full

Knights, D. and Willmott, H., 'Power and Subjectivity at Work', *Sociology*, Vol. 23, No 4, 1989, pp. 535–58. Kovacs, J. with Hilbert, C., Veselkova, M. and Virag, T., 'Jobs first? In search of quality', NEUJOBS State of the art Report N.D 2.1, 2012, available at http://www.neujobs.eu/sites/default/ files/publication/2012/08/NEUJOBS%20 STATE%200F%20THE%20ART%20 REPORT%20N0.%20D%202.1.pdf

Kunda, G., *Engineering Culture. Control and Commitment in a High-Tech Corporation*, Temple University Press, Philadelphia, PA, 1992.

Langfred, C. W. and Moye, A., 'Effects of Task Autonomy on Performance: An Extended Model Considering Motivational, Informational and Structural Mechanisms', *Journal of Applied Psychology*, Vol. 89, No 6, 2004, pp. 934–45.

Layard, R., *Happiness. Lessons from a New Science*, Allen Lane.

Lazear, E. P. and Rosen, S., 'Rank-order Tournaments as Optimum Labor Contracts', *Journal of Political Economy*, Vol. 89, 1981, pp. 841–64.

Lehndorff, S. and Voss-Dahm, D., 'The delegation of uncertainty: flexibility and the role of the market in service work', in Bosch, G. and Lehndorff, S. (eds.), *Working in the service sector* — *a tale from different worlds*, Routledge, London and New York, 2005, pp. 289–315.

Leschke, J. and Watt, A., 'Job quality in Europe', ETUI Working Paper WP 2008.07, 2008, available at http://www. etui.org/Publications2/Working-Papers/ Job-quality-in-Europe

Leschke, J. and Watt, A. with Finn, M., 'Putting a number on job quality?', ETUI Working Paper WP 2008.03, 2008, available at http://www. etui.org/Publications2/Working-Papers/ Putting-a-number-on-job-quality

Leschke, J. and Watt, A. with Finn, M., 'Job quality in the crisis — an update of the Job Quality Index (JQI)', ETUI Working Paper 2012.07, 2012, available at http://www.etui.org/Publications2/Working-Papers/Job-quality-in-the-crisis-anupdate-of-the-Job-Quality-Index-JQI

Levine, L. and Tyson, L. D., 'Participation, productivity, and the firm's environment', in Blinder, A. (ed.), *Paying for Productivity*, Brookings Institution, Washington, 1990, pp. 183–236.

Lewis, S. and Malecha, A., 'The impact of workplace incivility on the work environment, manager skill, and productivity', 2011, available at http://www.ncbi.nlm. nih.gov/pubmed/21799351

Lillie, N., 'Subcontracting, Posted Migrants and Labour Market Segmentation in Finland', *British Journal of Industrial Relations*, Vol. 50, No 1, 2011, pp. 148–67, available at http://www.researchgate.net/ publication/227371573_Subcontracting_Posted_Migrants_and_Labour_Market_Segmentation_in_Finland

Lindbeck, A. and Snower, D., 'The Insider-Outsider Theory: A Survey', IZA DP No 534, 2002, available at http://ftp.iza. org/dp534.pdf

Lindström, M., 'Psychosocial work conditions, social participation and social capital: A causal pathway investigated in a longitudinal study', *Social Science and Medicine*, Vol. 62, pp. 280–91, available at http://www.ncbi.nlm.nih.gov/ pubmed/16098650

Liu, C., Spector, P. E. and Jex, S., 'The relation of job control with job strains: A comparison of multiple data sources', *Journal of Occupational and Organizational Psychology*, Vol. 78, 2005, pp. 325–36.

Lloyd-Ellis, H., 'On the impact of inequality on productivity growth in the short and long run: a synthesis', *Canadian Public Policy* — *Analyse de Politiques*, Vol. XXIX, Supplement/Numéro Spécial, 2003.

Lorenz, E. and Valeyre, A., 'Les formes d'organisation du travail dans les pays de l'Union européenne', Document de Travail du Centre d'études de l'emploi, No 32, Noisy-le-Grand, France, Centre d'études de l'emploi, 2004.

Lorenz, E. and Valeyre, A., 'Organisational Innovation, Human Resource Management and Labour Market Structure', *The Journal of Industrial Relations*, Vol. 47, No 4, 2005, pp. 424–42.

Lucas, R., 'On the mechanics of economic development', *Journal of Monetary Economics*, Vol. 22, 1988, pp. 3–42.

Luttmer, E. F. P., 'Neighbors as Negatives: Relative Earnings and Well-Being', *Quarterly Journal of Economics*, Vol. 120, No 3, 2005, pp. 963–1002. MacDuffie, J. P. and Pil, F. K., 'Changes in Auto Industry Employment Practices: An International Overview', in Thomas, K., Lansbury, R. and MacDuffie, J. P. (eds.), *After Lean Production*, Cornell University Press, Ithaca, 1997, pp. 9–42.

Machin, S., 'Wage inequalities since 1975', in Dickens, R., Gregg, P. and Wadsworth, J. (Eds.), *The Labour Market Under New Labour*, Palgrave MacMillan, London, 2003.

MacIntyre, A., *After Virtue*, University of Notre Dame Press, Notre Dame, IN, 1984.

Mahdi, A. F., Zin, M. Z. M., Nor, M. R. M, Sakat, A. A. and Naim, A. S. A., 'The relationship between job satisfaction and turnover intention', *American Journal of Applied Science*, 2012, Vol. 9, pp. 1518–26.

Majundar, S., 'Market conditions and worker training: how does it affect and whom?' *Labour Economics*, Vol. 14, No 1, 2007, pp. 1–23.

Mankins, M., Bahm, C. and Caimi, G., 'Your Scarcest Resource, a Bain & Company report', *Harvard Business Review*, 2014.

Mark, G., Gudith, D. and Klocke, U., 'The Cost of Interrupted Work: More Speed and Stress', CHI '08 Proceedings of the SIGCHI Conference on Human Factors in Computing Systems, ACM, New York, USA, 2008.

Marmot, M., *Status syndrome: How your social standing directly affects your health and life expectancy*, Bloomsbury, London, 2004.

Martinson, K., Stanczyk, A. and Eyster, L., 'Low-Skill Workers' Access to Quality Green Jobs', The Urban Institute, Brief 13, 2010, available at http://www. urban.org/uploadedpdf/412096-lowskilled-worker.pdf

Mather, M. and Lighthall, N. R., 'Risk and reward are processed differently in decisions made under stress', *Current Directions in Psychological Science*, Vol. 21, No 1, 2012, pp. 36–41.

Mathers, C. D. and Loncar, D., 'Projections of global mortality and burden of disease from 2002 to 2030', *PLoS Med*, Vol. 3, 2006.

Mattera, P., Dubro, A., Gradel, T., Thompson, R., Gordon, K. and Foshay, E., 'High Road or Low Road? Job quality in the new green economy', Sierra Club, 2009, available at http://www.sierraclub.org/ greenjobs/downloads/high-low.pdf

McGinnity, F. and Russel, H. (2013), 'Work-Family Conflict and Economic Change', Chapter 7 in Gallie (2013).

Mintzberg, H., *The Structuring of Organisations*, Prentice Hall, Englewood Cliffs, NJ, 1979.

Mintzberg, H., *Structure in Fives. Designing effective organizations*, Prentice Hall, Englewood-Cliffs, NJ, 1983.

Monti, M., 'New strategy for the single market — at the service of Europe's Economy and Society', Report to the President of the European Commission, 2010, available at http://ec.europa.eu/bepa/pdf/ monti_report_final_10_05_2010_en.pdf

Morrissey, T., Dunifon, R. and Kalil, A., 'Maternal employment, work schedules, and children's body mass index', *Child Development*, Vol. 82, No 1, pp. 66–81.

Mosthaf, A., 'Low-wage jobs — stepping stones or just bad signals?', IAB Discussion Paper 11/201, 2011, available at http://doku.iab.de/discussionpapers/2011/dp1111.pdf

Muñoz de Bustillo, R. and Fernández Macías, E., 'Job satisfaction as an indicator of the quality of work', *Journal of Socio-Economics*, Vol. 34, No 5, 2005, pp. 656–73.

Muñoz de Bustillo, R., Fernández-Macías, E., Antón, J. I. and Esteve, F., *Measuring More than Money* — *The Social Economics of Job Quality*, 2011.

Mustilli, F. and Pelkmans, J., 'Access Barriers to Services Markets. Mapping, tracing, understanding and measuring', CEPS Special Report, No 77, 2013, available at http://www.ceps.eu/book/ access-barriers-services-marketsmapping-tracing-understanding-andmeasuring

NEUJOBS (s.a.), 'Job Quality in EU Research: Building Up the Knowledge Base', available at http://www.neujobs. eu/events/job-quality-eu-researchbuilding-knowledge-base Newhouse, J., 'Boeing Versus Airbus: The Inside Story of the Greatest International Competition in Business', 2007, available at http://www.youtube.com/ watch?v=dnDoJ-KFJa4

Nguyen, A. N., Taylor, J. and Bradley, S., 'Job Autonomy and Job Satisfaction: A New Evidence', Working Paper Lancaster University: The Department of Economics, 2003, available at http://www.research. lancs.ac.uk/portal/en/publications/ job-autonomy-and-job-satisfactionnew-evidence(c62c01f2-bac6-4f84b740-9a0ecdd92e4a)/export.html

Nonaka, I., 'A dynamic theory of organizational knowledge creation', *Organization Science*, Vol. 5, No 1, 1994, pp. 14–37.

Oorschot, van, W. and Jensen, P. H., 'Early retirement differences between Denmark and The Netherlands. A crossnational comparison of push and pull factors in two small European welfare states', *Journal of Aging Studies*, Vol. 23, No 4, 2009, available at http:// www.sciencedirect.com/science/article/ pii/S089040650800145X

Ophir, E., Nass, C. and Wagner, A. D., 'Cognitive control in media multitaskers', *Proceedings of the National Academy of Science*, 2009, Vol. 106, No 37.

Organisation for Economic Co-operation and Development (OECD), 'Chapter 3. How good is your job? Measuring and assessing job quality', OECD Employment Outlook (2014), 2014, available at http:// www.oecd.org/employment/oecdemploymentoutlook.htm

Organisation for Economic Co-operation and Development (OECD), 'Building quality jobs in the recovery — Issues paper', 2011, available at http://www.oecd.org/ cfe/leed/48859871.pdf

Organisation for Economic Co-operation and Development (OECD), 'OECD Guidelines for Multinational Enterprises — 2011 Update', 2011a, available at http:// www.oecd.org/daf/inv/mne/oecdguidelinesformultinationalenterprises.htm

Organisation for Economic Co-operation and Development (OECD), 'Moving Up the Value Chain: Staying Competitive in the Global Economy', 2007, available at http://www.oecd.org/sti/ ind/38558080.pdf Organisation for Economic Co-operation and Development (OECD), 'New enterprise work practices and their labour market implications', OECD Employment Outlook (1999), 1999, pp. 177–221, available at http://www.oecd.org/els/employmentoutlook-previouseditions.htm

Organisation for Economic Co-operation and Development (OECD), Conference on Job Quality, s.a., available at http:// www.oecd.org/cfe/leed/internationalconferencebuildingqualityjobsintherecoverydublinireland.htm

Osterman, P., 'How Common is Workplace Transformation and Who Adopts It?', *Industrial and Labor Relations Review*, Vol. 47, 1994, pp. 173–89.

Ostry, J., Berg, A. and Tsangarides, C. G., 'Redistribution, Inequality, and Growth', IMF Staff Discussion Note 14/02, 2014, available at http://www.imf.org/external/ pubs/ft/sdn/2014/sdn1402.pdf

Oswald, A. J., Proto, E. and Sgroi, D., 'Happiness and Productivity', University of Warwick, 2014, available at http://www2. warwick.ac.uk/fac/soc/economics/staff/ academic/proto/workingpapers/happinessproductivity.pdf

Park, J., 'Health factors and early retirement among older workers', Statistics Canada, 2010, available at http://www. statcan.gc.ca/pub/75-001-x/2010106/ article/11275-eng.htm

Patterson, M., Warr, P. and West, M., 'Organizational climate and company productivity: the role of employee affect and employee level', CEP Discussion Paper No 626, 2004, available at http://eprints.lse.ac.uk/19977/1/ Organizational_Climate_and_Company_ Productivity_the_Role_of_Employee_ Affect_and_Employee_Level.pdf

Pedersen, K. and Andersen, S. K., 'The enlarged EU and the free movement of East European service providers — Extent and effects on the Danish labour market', 2008, available at http:// faos.ku.dk/english/publications/2008/ the_enlarged_eu

Pedersini, R. and Pallini, M., 'Posted workers in the European Union', Eurofound Study, 2010, available at http://www.eurofound. europa.eu/eiro/studies/tn0908038s/ tn0908038s_5.htm Pellegrina, Dalla, L. and Saraceno, M., 'Posted Workers: Complements or Substitutes for Local Employment? Empirical Evidence from the EU', "Paolo Baffi" Centre Research Paper Series No 2013–136, 2013, available at http://papers.ssrn.com/sol3/papers. cfm?abstract_id=2287841

Pfeffer, J., *The Human Equation: Building Profits by Putting People First*, Harvard Business School Press, Boston, 1998.

Piketty, T., *Capital in the Twenty-First Century*, Harvard University Press, 2014.

Pink, D. P., Drive — The Surprising Truth About What Motivates Us, 2010.

Piore, M., 'The dual labor market', in Gordon, D. M. (ed.), *Problems in Political Economy: An Urban Perspective*, D.C. Heath, Lexington, MA, 1971.

Pissarides, C., 'Equilibrium unemployment theory', Basil Blackwell, Cambridge, MA, 1990.

Pollak, C., 'Employed and Happy Despite Weak Health? Labour Market Participation and Job Quality of Older Workers with Disabilities', IRDES Working Paper, No 45, 2012, update version available at http://www.touteconomie.org/ conference/index.php/afse/aim/paper/ viewFile/342/131

Ponce Del Castilllo, A., 'EU waste legislation: current situation and future development', *HesaMag No 9 Waste and recycling: workers at risk*, 2014, available at http:// www.etui.org/Topics/Health-Safety/News/ HesaMag-09-Waste-and-recycling-workers-at-risk

Prandy, K., 'The revised Cambridge scale of occupations', *Sociology*, Vol. 24, 1990, pp. 629–55.

Rath, T. and Harter, J., *Well-being — The Five Essential Elements*, Gallup Press, New York, 2010.

Ravn, M. O. and Sterk, V., 'Job Uncertainty and Deep Recessions', University College London, October 2013 version, available at http://www.homepages.ucl. ac.uk/~uctpvst/Ravn-Sterk-Uncertainty-Oct31.pdf Rebelo, S., 'Long-Run Policy Analysis and Long-Run Growth', *Journal of Political Economy*, Vol. 99, No 3, 1991.

Robertson, M. and Swan, J., 'Going public: The emergence and effects of soft bureaucracy in a knowledge intensive firm', *Organization*, Vol. 11, No 1, 2004, pp. 123–48.

Robotics-VO, 'A Roadmap for US Robotics. From Internet to Robotics', 2013 Edition, available at https://robotics-vo.us/ sites/default/files/2013%20Robotics%20 Roadmap-rs.pdf

Rosen, S., 'The Theory of Equalizing Difference', in Ashenfelter, O. and Card, D., *Handbook of Labor Economics, Vol. 1,* Elsevier, Amsterdam, 1986, pp. 641–92.

Rubery, J. and Grimshaw, D. (2001), 'ICTs and employment: The problem of job quality', *International Labour Review*, Vol. 140, No 2, pp. 165–92, available at http://onlinelibrary.wiley.com/doi/10.1111/ j.1564-913X.2001.tb00219.x/abstract

Rustico, L. and Sperotti, F., 'A Green Economy that Works for Social Progress. Working Conditions in "Green Jobs" — Results from WiRES', WiRES (Women in Renewable Energy Sector), 2011, available at http:// www.ituc-csi.org/IMG/pdf/Paper_Sperotti_ Rustico_2011.pdf

Rustico, L. and Sperotti, F., 'Working conditions in "green jobs": Women in the renewable energy sector', *International Journal of Labour Research*, Vol. 4, No 2, 2012, available at http://www.skillsforemployment.org/wcmstest4/groups/skills/documents/skpcontent/ddrf/mdq1/~edisp/ wcmstest4_045236.pdf

Sacks, D., Stevenson, B. and Wolfers, J., 'The New Stylized Facts About Income and Subjective Well-Being', *Emotion*, Vol. 12, No 6, 2012, pp. 1181–7.

Schwieren, C. and Weichselbaumer, D., 'Does competition enhance performance or cheating? A laboratory experiment', *Journal of Economic Psychology*, Vol. 31, No 3, 2010, pp. 241–53.

Shapiro, C. and Stiglitz, J. E., 'Equilibrium Unemployment as a Worker Discipline Device,' *American Economic Review*, Vol. 74, 1984, pp. 433–44. Shleifer, A., 'Does Competition Destroy Ethical Behavior?', *American Economic Review*, Vol. 94, No 2, 2004, pp. 414–18.

Siegrist, J. and Wahrendorf, M., 'Quality of Work, Health and Early Retirement: European Comparisons', in Börsch-Supan, A. et al. (eds.), *The Individual and the Welfare State*, Springer-Verlag, Berlin Heidelberg, 2011, available at http://link.springer.com/ chapter/10.1007%2F978-3-642-17472-8_15#page-1

Siegrist, J., Wahrendorf, M., Knesebeck, von dem, O., Jürges, H. and Börsch-Supan, A., 'Quality of work, well-being, and intended early retirement of older employees baseline results from the SHARE Study', *European Journal of Public Health*, Vol. 17, No 1, 2007, pp. 62–8, available at http:// eurpub.oxfordjournals.org/content/17/1/62. short

Spanish Wind Energy Association (AEE), 'Accident rates report within the wind energy sector, period 2007–11', Report No 2, 2012, available at http://www.aeeolica.org/ uploads/documents/4063-accident-ratesreport-within-the-wind-energy-sectorreport-2-period-2007-2011.pdf

Standing, G., *The Precariat, The New Dangerous Class*, Bloomsbury, London, 2011.

Stanoevska-Slabeva, K., Blijsma, M., Gareis, K., Vartiainen, M. and Verburg, R., 'Collaborative work: Globalisation and New Collaborative Working Environments — New Global', Final Report, 2009, available at http:// www.empirica.com/publikationen/documents/2009/FinalReport_FinalVersion.pdf

Starbuck, W., 'Learning by knowledgeintensive firms', *Journal of Management Studies*, Vol. 29, No 6, 1992, pp. 713–40.

Stevenson, B. and Wolfers, J. (2013), 'Subjective Well-Being and Income: Is There Any Evidence of Satiation?', *American Economic Review, Papers and Proceedings*, Vol. 101, No 3, 2013, pp. 598–604.

Stevenson, B. and Wolfers, J., 'Economic Growth and Subjective Well-Being: Reassessing the Easterlin Paradox', *Brookings Papers on Economic Activity*, Spring 2008.

Stewart, A., Prandy, K. and Blackburn, R. M., *Social Stratification and Occupations*, Macmillan, London, 1980. Stiglitz, J., Sen, A. and Fitoussi, J.-P., 'Report by the Commission on the Measurement of Economic performance and Social Progress', 2009, available at http://www. stiglitz-sen-fitoussi.fr/documents/rapport anglais.pdf

Sumner, S. A. and Layde, P. M., 'Expansion of renewable energy industries and implications for occupational health', *Journal of the American Medical Association*, Vol. 302, No 7, 2009.

Sverke, M., Hellgren, J. and Näswall, K., 'Job insecurity. A literature review', 2006, available at http://nile.lub.lu.se/arbarch/ saltsa/2006/wlr2006_01.pdf

Taylor, P. and Walker, A., 'Age discrimination in the Labour Market and Policy responses: the Situation in the United Kingdom', *The Geneva Papers on Risk and Insurance*, Vol. 28, No 4, 2003, pp. 612–24, available at http:// www.genevaassociation.org/PDF/ Geneva_papers_on_Risk_and_Insurance/ GA2003_GP28(4)_Taylor&Walker.pdf

Tahlin, M. (2013), 'Distribution in the Downturn', Chapter 3 in Gallie (2013).

Theorell, T., 'Psychosocial factors in research on work conditions and health in Sweden', *Scandinavian Journal of Work and Environmental Health*, Vol. 33, No 1, 2007, pp. 20-6.

Theorell, T. and Karasek, R., 'Current issues relating to psychosocial job strain and cardiovascular disease research', *Journal of Occupational Health and Psychology*, Vol. 1, No 1, pp. 9–26.

Thompson, C.A. and Prottas, D., 'Relationship Among Organizational Family Support, Job Autonomy, Perceived Control, and Employee Well-Being', *Journal of Occupational Health Psychology*, Vol. 10, No 4, 2005, pp. 100–18.

Totterdill, P., 'The Future We Want? Work and Organisations in 2020', A Report by the Advisory Board of the UK Work Organisation Network, 2014, available at http:// www.uk.ukwon.eu/_literature_1812/ UKWON_2020_Report

United Nations Development Programme (UNDP), 'Powerful synergies. Gender Equality, Economic Development and Environmental Sustainability', 2012, available at http://www.undp.org/ content/dam/undp/library/gender/Gender%20and%20Environment/Powerful-Synergies.pdf

United Nations Economic Commission for Europe (UNECE), 'Introduction of the Conceptual Framework for Measuring the Quality of Employment', 2009, available at http://www.unece.org/fileadmin/DAM/stats/ documents/ece/ces/ge.12/2009/zip.4.e.pdf

United Nations Economic Commission for Europe (UNECE), 'Measuring quality of employment. Country pilot reports', 2010, available at http://www.unece.org/ fileadmin/DAM/publications/oes/STATS_ MeasuringQualityEmploment.E.pdf

United Nations Environment Programme (UNEP), 'Green Jobs — Towards Decent Work in a Sustainable, Low 'Carbon World', 2008, available at http://www.unep.org/ PDF/UNEPGreenjobs_report08.pdf

United States Department of Labor, 'Why Green Is Your Color: A Woman's Guide to a Sustainable Career', available at http:// www.dol.gov/wb/Green_Jobs_Guide/Green-Jobs%20Ch%202.pdf

Valeyre, A., Lorenz, E., Cartron, D., Csizmadia, P., Gollac, M., Illéssy, M. and Makó, C., Eurofound, Work organisation in Europe, Technical report, 2008.

Valeyre, A., Lorenz, E., Cartron, D., Csizmadia, P., Gollac, M., Illéssy, M., and Makó, C., 'Working conditions in the European Union: Work organisation', Eurofound report, 2009, available at http://www.eurofound.europa.eu/publications/htmlfiles/ef0862.htm

Vieira, J. A. C., 'Low pay, higher pay and job quality: empirical evidence for Portugal', *Applied Economics Letters*, Vol. 12, No 8, 2005, pp. 505–11, available at http://www.tandfonline.com/doi/ abs/10.1080/13504850500109907#. UzghEfl_vl8

Vienna Institute for International Economic Studies, WiiW, 'Study on various aspects of earnings distribution using micro-data from the European Structure of Earnings Survey', Final report submitted to DG EMPL (contract VC/2013/0019), 2014, available at http://ec.europa.eu/ social/keyDocuments.jsp?advSearchKey =analysislabourmarket&mode=advanc edSubmit&langId=en&policyArea=&typ e=0&country=0&year=0 Virtanen, M., Stansfeld, S. A., Fuhrer, R., Ferrie, J. E. and Kivimaäki, M., 'Overtime Work as a Predictor of Major Depressive Episode: A 5-Year Follow-Up of the Whitehall II Study', PLoS ONE, Vol. 7, No 1, 2012.

Visser, J., 'The Institutional Characteristics of Trade Unions, Wage Setting, State Intervention and Social Pacts', ICTWSS Database, version 4.0, 2013, available at http://www.uva-aias.net/208

Wahrendorf, M., Sembajwe, G., Zins, M., Berkman, L., Goldberg, M. and Siegrist, J., 'Long-term Effects of Psychosocial Work Stress in Midlife on Health Functioning After Labor Market Exit — Results from the GAZEL Study', *The Journals of Gerontology: Series B*, Vol. 67, No 4, 2012, pp. 471–80, available at http://psychsocgerontology.oxfordjournals.org/content/67/4/471.short

Wang, P. S., Beck, A. L., Berglund, P., McKenas, D. K., Pronk, N. P. et al., 'Effects of major depression on moment-in-time work performance', *American Journal of Psychiatry*, Vol. 161, No 10, 2004, pp. 1885–91.

Weintraub, E. R., 'Neoclassical Economics', in *The Concise Encyclopedia Of Economics*, 2007.

Westgaard, R. H. and Winkel, J., 'Occupational musculoskeletal and mental health: Significance of rationalization and opportunities to create sustainable production systems — A systematic review', *Applied Ergonomics*, Vol. 42, 2011, pp. 261–96.

Wills, J., 'Subcontracted Employment and its Challenge to Labor', *Labor Studies Journal*, Vol. 443, No 4, p. 441.

Wolf, E. and Zwick, T., 'Welche Personalmaßnahmen entfalten eine Produktivitätswirkung?', *Zeitschrift für Betriebswirtschaft*, Vol. 73 (EH4/2003), 2003, pp. 43–62.

Work Foundation, 'Is Knowledge Work Better For Us? Knowledge workers, good work and wellbeing', 2010, available at http://www.theworkfoundation.com/ downloadpublication/report/238_238_ ke_wellbeing_final_final.pdf

Wright, P., *Managerial Leadership*, Routledge, London, 1996. Yellen, J. L., 'Efficiency Wage Models of Unemployment', *American Economic Review*, Vol. 74, 1984, pp. 200–5.

Yeung, N., 'Between-task competition and cognitive control in task switching', *Journal of Neuroscience*, Vol. 26, 2006, pp. 1429–38. Youndt, M. A., Snell, S. A., Dean, J. W. and Lepak, D. P., 'Human resource management, manufacturing strategy, and firm performance', *Academy of Management Journal*, Vol. 39, 1996, pp. 836–66.

Zarifian, P., '*A quoi sert le travail ?*', La Dispute, Paris, 2003.

Zwick, T., 'Employee participation and productivity', *Labour Economics*, Vol. 11, No 6, 2004, pp. 715–40.